



Sorbonne Université LPSM

Doctoral School École Doctorale Sciences Mathématiques de Paris Centre

University Department Laboratoire de Probabilités, Statistique et Modélisation

Thesis defended by Miguel MARTÍNEZ HERRERA

Defended on November 12, 2024

In order to become Doctor from Sorbonne Université

Academic Field Applied Mathematics

Speciality **Statistics**

Inference of non-linear or imperfectly observed Hawkes processes

Thesis supervised by	Arnaud Guyader	Supervisor
	Anna Bonnet	Co-Supervisor
	Maxime Sangnier	Co-Supervisor

Committee members

Referees	François Roueff Gordon Ross	Telecom Paris, Institut Polytechnique de Paris University of Edinburgh	
Examiners	Sophie Donnet Céline Duval	INRAE Sorbonne Université	Committee President
Guest	Felix Cheysson	Université Gustave Eiffel	
Supervisors	Arnaud Guyader Anna Bonnet Maxime Sangnier	Sorbonne Université Sorbonne Université Sorbonne Université	





COLOPHON

Doctoral dissertation entitled "Inference of non-linear or imperfectly observed Hawkes processes", written by Miguel MARTÍNEZ HERRERA, completed on December 3, 2024, typeset with the document preparation system IAT_{EX} and the yathesis class dedicated to these prepared in France.

This thesis has been prepared at

Laboratoire de Probabilités, Statistique et Modélisation Sorbonne Université Campus Pierre et Marie Curie 4 place Jussieu 75005 Paris France



a +33 1 57 27 93 16

 Web Site
 https://www.lpsm.paris/

A ese niño que pasaba las noches haciendo sumas.

La vida sería mucho más agradable si uno pudiera llevarse a dónde quiera que fuera los sabores y los olores de la casa materna.

Laura Esquivel

What I need is the dandelion in the spring. The bright yellow that means rebirth instead of destruction. The promise that life can go on, no matter how bad our losses. That it can be good again.

Suzanne Collins

Ce que je ferai ici aura au moins le mérite de ne ressembler à personne, parce que ce sera l'impression de ce que j'aurai ressenti, moi tout seul.

Claude Monet

Agradecimientos

Como lo dicta la tradición tras escribir un tan largo y arduo trabajo, esta sección es una pequeña impresión del agradecimiento que tengo por mucha gente. Esta fue la gente que me apoyó durante estos tres años de tesis (y más allá) y sin quienes no habría podido llevar a cabo una tal labor.

Como muchos saben y otros se estarán dando cuenta, soy una persona que no sabe comenzar las cosas, tampoco terminarlas y mucho menos resumirlas. Sin embargo intentaré con todo mi ser hacer de esta lectura una experiencia amena. Por ejemplo, pueden divertirse contando el número de veces que le agradezco a alguien por los chismes (una de mis actividades preferidas).

Las personas que me conocen sabe que soy una persona que se revitaliza con las relaciones humanas. Muchos me describen como sonriente, lleno de color, de luz, de energía e intensidad. La verdad es que soy lo que soy gracias a la bondad y al amor que todas las personas que menciono aquí tienen por mí.

Quiero empezar agradeciendo a mis supervisores, quienes tuvieron la difícil tarea de soportarme durante 3 años. Arnaud, gracias por tu carácter determinante, decisivo y tajante, ya sea para arreglar problemas o resolver crucigramas. Maxime y Anna, esta aventura empezó con ustedes cuando decidimos, de una manera vagamente espontánea, trabajar juntos. Anna, gracias por no haber respondido a aquella candidatura que te envié ya que de otra manera nunca me habría aventurado en el mundo de los procesos de Hawkes. Gracias por todos los momentos conviviales, las largas discusiones de matemáticas, ecología y los chismes. Me dio mucho gusto compartir tantas conferencias juntos, en especial las escapadas a Toulouse con los Hawkes. Muchas gracias también por permitirme dar clases con los RESPE y mostrarme que siempre podemos usar nuestro tiempo para ayudar a los demás. Mis años de doctorado habrían sido mucho menos divertidos sin tu buen humor, que me harán tanta falta en lo que viene. Maxime, gracias por hacer de tu bureau nuestro cuartel general de reuniones, de siempre sacar las galletas ocultas cuando nos veíamos a las tempranas 9 de la madrugada y sobre todo por tu rigor científico. Aunque no tengamos las mismas maneras de trabajar (un día te convenceré de venir a trabajar a un café conmigo) me da gusto saber que contigo siempre es posible reír y bromear. Les agradezco porque siento que hacer un doctorado es un reto enorme y estoy contento de lo que logramos hacer juntos.

Le agradezco a mi jurado por haber aceptado evaluar mi trabajo. Gracias François Roueff y Gordon J. Ross por haber tomado el tiempo de leer mi manuscrito. Gracias Céline Duval y Sophie Donnet por haber aceptado ser mis examinadoras.

Muchas gracias tambien a Eva Löcherbach, Viet Chi Tran y Etienne Roquain por haber formado parte de mi comité de seguimiento de tesis.

Felix, te agradezco por haber sido un guía de los procesos de Hawkes y sobre todo de la

teoría espectral. Recuerdo haberte conocido en tu posdoctorado en el laboratorio y como toda buena historia ahora continuamos a trabajar juntos esta vez para el mio. Gracias por siempre dar de tu tiempo para explicarme cosas, para compartir con tanta pasión tus hobbies, opiniones y conocimientos, o simplemente para charlar. Aprovecho para agradecerle a la comunidad de Hawkes con los que compartí conferencias, seminarios, pláticas y comidas, ya sea en Toulouse, Lille, Paris o donde sea (porque al parecer estamos en todos lados).

Le agradezco a todos los miembros del LPSM que animan la vida científica de esta universidad con sus personalidades tan variadas, tan únicas pero sobre todo siempre bondadosas. Gracias Claire por haberme integrado en este laboratorio desde mi época de pasante y por hacernos sentir a mi y a los otros doctorantes como verdaderos colegas. Suerte en tu nueva aventura. Gracias Stéphane por siempre tener una sonrisa, por siempre echar la plática y compartir tu vasto saber. Espero que estos años como vecinos de oficina donde te toco escuchar música, risas y cantos hayan sido no muy desagradables. Gracias Antoine por siempre recordarnos a los doctorantes que hay que disfrutar de la vida y saber cuando dejar de trabajar. Gracias Charlotte por tu gentileza, por organizar el grupo de Hawkes. Fue muy agradable formar parte del grupo de profesores de L3 Estadística contigo. Gracias a los demás miembros Anna BH, Jean-Patrick, Maud, Erwan, Sylvain, Lorenzo (y todos y todas que habré olvidado poner aquí). Le agradezco de la misma manera a los miembros de la administración Valerie, Nathalie, Marouane. En particular, gracias Louise que me acompañaste de pasante y en mis primeros años de tesis. Adoré cada momento que pude platicar contigo, mil gracias por constantemente preguntar como podías ayudar, hiciste que mi vida laboral fuera más amena. Gracias Hugues, el Atlas que carga el mundo informático de este laboratorio a base de mails con un humor singular e hilarante. No hay nada más divertido que leer que no puedes hacer algo "inmediatamente" pero que en 1 hora estará listo. Si hubiera sido tan eficaz, habría probablemente publicado 77 papeles durante mi tesis. Gracias a todos los miembros de limpieza porque gracias a ustedes podemos trabajar en condiciones óptimas.

En toda aventura se conoce gente que nos acompaña y con quienes compartimos logros, felicidad, desesperación y sobre todo cafés... muchos cafés. Camila, Cami, no podría estar más feliz de haber estado en la misma oficina contigo, nada mejor que tener un pedacito de Latinoamérica tan cerca. Aprecio cada momento de canto y música, de enviarnos memes, de pegar el rostro del otro en las sillas y más que nada la felicidad de gritar cuando nos volvemos a ver, no importa que hayan pasado unos días o varias semanas sin vernos. Aún recuerdo perfectamente que tan cercanos nos volvimos en Porquerolles y nuestro subsecuente verano bailando cada viernes en la noche en los quais. Que sepas que admiro muchísimo la intensidad y la determinación que te lleva a hacer todo lo que te gusta con tanto fervor. Espero seguir con nuestras sesiones de shopping, karaoke, chisme y comida por años y años, porque contigo la vida es tan divertida y amena. Te quiero más que a la diosa Shaki. Iqraa, gracias por esas pláticas, a veces tan filosóficas, en el quinto piso o en un café perdido de Paris. Eres una persona con mucho corazón y que sabe ser fiel a sus principios y creencias: que sepas que es algo que admiro muchísimo de ti. Gracias porque contigo siempre puedo contar para bailar y cantar a todo pulmón. Gracias por haber compartido un Airbnb conmigo en nuestra conferencia en Lyon, donde nos quedabamos hasta tarde platicando, fue muy agradable vivir contigo esos días. Sobre todo, gracias por todas esas veces que me escuchaste y me tendiste los brazos para abrazarme. Ariane, aprecio mucho el habernos acercado tanto con los años y formar parte de ese grupo exclusivo de gente que puede abrazarte (con moderación). En particular, recuerdo mucho aquella vez que terminamos comiendo por la noche en Burger King, o aquella plática llena de emociones en Bruselas sentados en la barra de un restaurante japonés. Entre mis mejores recuerdos con todas ustedes están esas pláticas de chisme, amores y desamores, risas y llantos. Ahora que oficialmente nos vamos todos del Labo, habrá que encontrar donde hacer nuestras sesiones de películas en verano ! Quiero que sepan que su amistad es un regalo que atesoro, y que su amor y cariño llena de color y vida mi día a día.

Ahora toca agradecerle a dos personas que conocí en la Maestría. Claire, Clairounette, Clairicot, ma chère Claire, soy tan pero tan feliz de haberte conocido, de haberte propuesto de pasar tu pasantía en mi oficina porque fue el inicio de una amistad que hasta hoy es de las más preciadas para mí. Tienes más energía que el Sol, siempre te veo corriendo de derecha a izquierda (a veces literalmente), ya sea para compartir unos juegos de mesa en invierno o la playa en verano (mil mil mil gracias por hacerme descubrir Royan). Llevo siempre conmigo todas esas largas y largas caminatas por todo París, compartiendo ideas, opiniones, chismes y comida, mucha mucha comida. Gracias también por todos los consejos y por siempre ser tan directa conmigo. No importa donde estés, siempre estarás cerca mio. Te mando mis mejores deseos para este nuevo episodio que estás comenzando. Ludovic, espero que andes leyendo la versión en español de los agradecimientos. Ludo, amo verte sobrellevar la vida con una sonrisa, con calma, con una aura tan chill. Siempre has sabido bromear conmigo y por siempre compartir tu pasión por la cultura, la historia, y los bebés feos del arte sacro. En especial, gracias por bailar conmigo en tantas noches al borde del río (no te preocupes ya no me duele el codo).

Gracias Alexis. Gracias por todas las noches comiendo tu deliciosa comida. Gracias por enseñarme el mundo de las fiestas techno, por haberme acompañado a tantas fiestas pop (por más que decías que las odiabas). Te quiero agradecer por todo el cuidado que me diste, por todo ese amor y amistad, por todos esos tragos que compartimos, esas pláticas, los tantos chismes, los grafos, los consejos, los abrazos. Acercarme a ti fue de las cosas mas bonitas que me sucedieron en el doctorado. Sigue viviendo con la pasión e intensidad con la que la techno te hace bailar. Antonin, mi ciela, que gusto me da tenerte en mi vida, poder reírnos tanto con esos edits de las estrellas pop, el poder cantar a todo pulmón Beyoncé y que siempre estés ahí para escucharme. Sin importar lo que pase, ustedes siempre ocupan un lugar especial en mi corazón y espero poder tenerlos cerca por años.

Le agradezco a una doctorante que es tan dedicada como feminista e hiperactiva: Adeline. No pensé encontrar alguien que piensa y habla tan rápido como yo, nuestras platicas en mi oficina (o la del jefe) eran caóticas, divertidas y llenas de consejos amorosos. Muchas gracias porque supiste venir y darme apoyo cuando lo necesité tanto. Aún recuerdo ver Batman juntos aquella noche.

Hubo un verano en el que conocí a dos pasantes con los que compartí y comparto mucho. Me da risa saber que no importa lo que escriba aquí los dos vendrán a molestarme, y por eso los adoro. Ángel, nuestras pláticas sin fin son algo que siempre he atesorado. Me gusta saber que puedo platicar de todo contigo, y más tarde ir a bailar pop o techno (aunque haya que cuidarte después). Tu eres mi niña fresa. Eyal, tal vez te forzamos mucho con la música pop, pero veo que tuvo sus frutos así que no me arrepiento de nada. Gracias por todas las veces que salimos a comer, a compartir un trago y a bailar. Yo se que el bullying es tu lenguaje de amistad, eso aprendí en aquella noche en tu terraza. Yo se que puedo contar contigo y que sepas que yo también estoy aquí para ti. De cierta manera ustedes toman el relevo, y aunque se sientan grandes y poderosos siempre estaré aquí para recordarles que los conocí en pañales.

Pierre, muchas gracias por divertirme tanto durante los exámenes (tú sabes de lo que hablo). Gracias Ferdinand por esas noches de diversion entre música en ruso y música en español. Nina, que llegaras al labo fue como ver el sol salir después de los largos inviernos en Francia. Eres una persona con tanta vibra positiva y sassiness que no puedo más que adorar tu compañía.

Gracias a los otros doctorantes pasados y presentes del LPSM con los que compartimos pausas, historias de felicidad, de tristeza y de (des)esperanza. Gracias Francesco por siempre traer una sonrisa, Grace por siempre pasar a saludar a mi oficina, Paul por los bares y fiestas sobre todo en Bordeaux, Romain por las pláticas de Hawkes. Gracias al trio de grandes Nicklas, Gloria y Thibault con los que pase muchos ratos divertidos alrededor de un café o un vinito, como en Lyon. Gracias Antonio, Yazid, Gabriel, Lucas, Barbara, Mathis, Alessandro, Anna, Arthur, Stan.

Haydee, beibi, muchas gracias por ser quien eres, por reír a todo pulmón conmigo. Atesoro tanto los recuerdos tan largos, numerosos y diversos que tenemos. Después de 11 años de amistad, me parece una locura ver que cada que nos vemos seguimos tan conectados. No puedo creer que seguimos a veces terminando la frase o las bromas del otro. Tu amistad vale oro puro para mí y te agradezco tanto pero tanto por permitirme tenerte tan cerca. Te adoro.

Tim, amo nuestra amistad que perdura sin importar que pasen años sin vernos, y meses sin hablarnos, porque se que siempre estarás ahí para platicar, reír y ayudarme. Que locura que ya llevemos 8 años conociéndonos. No puedo evitar reír con todos los recuerdos que tenemos en Toulouse

Romain, te agradezco por los años que compartimos, por todo lo que vivimos y todo lo que maduramos juntos. Nunca olvidaré todos esos bancos que fueron ni todas esas películas y series que devoramos. Llevo conmigo siempre todo lo que fuimos, gracias por acompañarme al inicio de este viaje. Espero sigas riendo fuerte y siendo feliz.

El trabajo que concluye hoy empezó de alguna manera en 2015 cuando llegue a Francia, y tengo que hablar de esa persona con la que empezó todo: Fernanda. Llegamos juntos, y fuimos el soporte del otro durante tanto tiempo, entre llamadas de 5 horas y salidas de fiesta hasta las 6 (con su correspondiente comida antes de dormir). Yo se que no habría logrado quedarme y lograr todo lo que logré sin tu apoyo ya que, pasara lo que pasara, yo sabía que estarías ahí para escucharme. Extraño esos cafés de trabajo con nuestro "amigo" que siempre andaba escribiendo y leyendo. Me da gusto ver lo que te has vuelto, y aunque ahora estemos tan lejos físicamente, siempre te tengo conmigo.

Alfredo, me da tanto gusto el rencontrarnos y darnos cuenta que desde el liceo pensábamos en acercarnos. Estoy feliz por todo lo que has logrado, y como sigues evolucionando y avanzando. Sigamos cantando, bailando y criticando mi cochina.

Es momento de agradecerle a la persona más importante en este mundo para mí. Mi señora madre, mamá, hermosa. Muchas gracias por haberme traído, por haberme criado, por haberme hecho la persona que soy hoy, fuerte, dedicada y sobre todo feliz. Es gracias a ti que puedo ver mis manos llenas de frutos de todo lo que he sembrado. Espero poder darte todo lo que te mereces por todo el amor que me has dado y los sacrificios que has hecho. Todo lo que soy, lo que tengo, lo que deseo es por ti, vive por ti, respira por ti. Tenerte tan lejos es extrañarte a un punto que parece a veces insoportable pero yo se que siempre estarás aquí conmigo, y que sepas que siempre estoy ahí para ti. Te amo con todo lo que soy.

Lidi, muchas gracias por haber sido como mi segunda madre, gracias porque tu me hiciste siempre sentir parte de tu familia, me diste tanto amor que a veces no se como retribuirlo. Fuiste la que primero me mostró el mundo de este lado del charco. Siempre estuviste para apapacharme, consentirme y apoyarme en tantas pero tantas cosas. Te amo idi.

Fa, papá, muchas gracias por todo lo que me has dado, por la paciencia y el amor que me has tenido. Algo que siempre me dio y me da seguridad es saber que siempre estás ahí para mí. Que siempre estarás cuando lo necesito y que por más que nuestros humores choquen (lo heredé un poquito de ti hahaha) siempre me amas con todo tu ser. Te amo.

Muchas gracias Mar porque crecer contigo fue como tener una hermana que me protegía y cuidaba. No sabes como te agradezco porque siempre te sentí tan gentil conmigo. Gracias a ti entendí muchas cosas y mi vida no habría sido para nada lo que es hoy sin ti. Marco, German, gracias por haberme tratado como a un hermano (con todo lo bueno y malo que viene en ese paquete) porque gracias a ustedes me sentí mucho menos solo.

Agradecimientos

Mari, prima, Mariana, tu alma, tu presencia, tu ser me hacen sentir acompañado en este océano gigantesco que es la realidad. Pensamos, sentimos, vivimos la vida de una manera tan similar, que seguramente fue lo que nos acerco desde chicos y nos unió tan rápido y tan fuerte. Gracias por cada segundo donde me apoyaste, donde me escuchaste y me diste tus consejos. Te quiero tanto pero tanto que quisiera tener tu pluma para poder expresarlo con palabras. Sigue tu fuego interior siempre, nunca olvides quien eres, lo que has logrado, y el poder que tienes, porque al menos mi vida la cambiaste radicalmente.

Nay, Ive, agradezco tanto a la vida por habernos acercado tanto cuando lo hizo. Soy tan feliz que sean mis hermanas y que me muestren su amor en cada oportunidad que tienen. Bailar con ustedes, cantar con ustedes, festejar con ustedes siempre ha tenido un calor singular que me llena de felicidad. Las adoro.

Palola, Pola, Dude, mi infancia no habría sido infancia sin ti, y mis pulmones no serían tan fuertes sin las horas y horas de risas que compartimos. No creo conocer a nadie que me haya dado tantos dolores de estomago con tanta tontería. Tengo tantos recuerdos de pelearnos, encontentarnos, hacer travesuras, hacer enojar a nuestras madres y reír a nuestros abuelos. Te extraño mucho y te quiero todavía más.

Quiero agradecerle a mis tías porque, como muchos notaran, es gracias a mujeres muy fuertes que soy lo que soy. Muchas gracias Lale por tus cuidados y por todas las tardes de películas, telenovelas y sobre todo risas con Pao y mi mamá. Maruka, extraño tanto las cositas de la mujer en las noches tu y yo. Tengo mis recuerdos llenos de pequeños momentos contigo que tal vez no he sabido decirte a que punto llenan mi alma de dulzura, calidez y tranquilidad. Gracias. Gracias Kenuki y Elsuki porque me hicieron siempre sentir como un niño, algo que adoro. Extraño tanto las navidades con ustedes en las que nos la pasábamos en la cocina entre locuras, gritos y, mis favoritos, chismes. Gracias Ceci por siempre haber estado ahí y por compartir tu amor (y tu amor por el pastel de fresa hahah).

Quiero agradecerle a mis abuelitos y abuelitas que siempre me llenaron de amor y abrazos, cada uno y una a su manera. Espero que estén donde estén se sientan bien y orgullosos.

Para terminar con estos agradecimientos, quiero hablar de dos personas que son mis mas grandes tesoros.

Cosette, Cosa, la reina, o como otros te conocen, Carolina. Eres la prueba viviente que la familia se construye, la familia se elige, la familia son esas personas que están ahí siempre. Eres mi familia, eres esa persona que me sostiene, que me da mis regañones, que me escucha, que me mira a los ojos y me dice "Se va a estabilizar", porque sabes que así será. Tiene tantos años que te conozco que siento que siempre has estado ahí, que siempre has compartido conmigo tu risa, tus chismes, tus consejos de belleza y, sobre todo, tu ser. La vida te trajo tan cerca de mí, y soy tan feliz de poder contar contigo. No estaría donde estoy hoy sin ti. Podría escribir mil párrafos agradeciéndote por cada bella cosa que me has dado, por cada acto de afecto y amor, y aunque yo se que amarías, me faltaría papel en Paris para imprimir los manuscritos. Así que resumiré esto en una frase: you are my person.

Gwen. Gracias Gwen. Gracias por existir. Gracias por haber aceptado aquella primera cita. Gracias por las risas, los llantos, las palomas, las series, las noches de películas de terror y las mañanas de café viendo Bob's. Eres de lo más grande que tengo. Eres una persona llena de gentileza, de amor, de humanidad y no dejas de mostrarme tu amabilidad en cada día que llevo conociéndote. Desde el primer día has sido un apoyo incondicional, estuviste ahí en especial durante la escritura de este manuscrito, y vaya que fue difícil, pero gracias a ti pude lograrlo. Gracias por cocinarme, por escucharme, por abrazarme. Gracias por quedarte cuando lo necesito, por darme la libertad que tanto amo. No hay un lugar en donde me sienta más seguro que entre tus brazos. Gracias por ser quien eres. This nest is beautiful because it's ours.

Remerciements

Comme le veut la tradition après avoir réalisé un aussi long et éprouvant travail, cette section est une petite impression de la gratitude que j'ai pour beaucoup de gens. Ici se trouvent les gens qui m'ont soutenu pendant ces trois dernières années (et bien au-delà) et sans qui je n'aurai pu mener à bout un tel travail.

Comme beaucoup le savent, et d'autres le découvrent, je suis une personne qui ne sait pas commencer les choses, ni les conclure, et encore moins les résumer. Toutefois, je vais essayer avec tout mon être de faire de cette lecture une expérience chaleureuse, ou du moins amusante. Par exemple, vous pouvez vous amuser à compter le nombre de fois que je remercie quelqu'un pour les ragots (une de mes activités préférées).

Les personnes qui me connaissent savent que je suis une personne qui se revitalise à travers les relations humaines. Beaucoup me décrivent comme souriant, plein de couleurs, de lumière, d'énergie et d'intensité. Il s'avère que je suis tout cela grâce à la bienveillance et l'amour que toutes les personnes mentionnées ici éprouvent pour moi.

Je veux commencer en remerciant mes encadrants qui ont eu la lourde tâche de me supporter pendant 3 ans. Arnaud, merci pour ton caractère déterminé, décisif et tranchant, que ce soit pour résoudre des problèmes ou des mots-croisés. Maxime et Anna, cette aventure a commencé avec vous quand nous avons décidé, d'une manière vaguement spontanée, de travailler ensemble. Anna, merci de ne pas avoir répondu à cette candidature que je t'avais envoyé car autrement je ne me serais pas lancé dans le monde des processus de Hawkes. Merci pour tous ces moments de convivialité marqués par tant de discussions de maths, d'écologie et de ragots. Je suis tellement ravi d'avoir partagé autant de conférences ensemble, en particulier nos escapades à Toulouse avec les Hawkes. Merci également de m'avoir permis d'enseigner chez les RESPE et de m'avoir montré que nous pouvons toujours utiliser notre temps pour aider les autres. Mes années de doctorat auraient été beaucoup moins amusantes sans ta bonne humeur qui me manquera tant à l'avenir. Maxime, merci d'avoir fait de ton bureau notre quartier général, d'avoir toujours sorti les cookies secrets quand on se voyait à 9 heures de l'aube, et surtout de ta rigueur scientifique. Même si nos manières de travailler ne sont pas toujours alignées (un jour je te convaincrai de venir travailler dans un café avec moi), je suis content de savoir que je pouvais toujours rire et blaguer avec toi. Je vous remercie car faire un doctorat est un défi énorme et je suis content de ce que l'on a accompli ensemble.

Merci à mon jury d'avoir accepté d'évaluer mon travail. Merci François Roueff d'avoir pris le temps de lire mon manuscrit. Thank you Gordon J. Ross for having taken the time to read my manuscript. Merci Céline Duval et Sophie Donnet d'avoir accepté d'être mes examinatrices.

Merci beaucoup aussi à Eva Löcherbach, Viet Chi Tran et Etienne Roquain d'avoir accepté

d'être dans mon comité de suivi de thèse.

Felix, je te remercie car tu as été un guide dans l'univers des processus de Hawkes et surtout de la théorie spectrale. Je me souviens de t'avoir rencontré dans tes années de postdoc au laboratoire, et comme toutes les bonnes histoires, on continuera de travailler ensemble cette fois pour le mien. Merci de toujours avoir donné ton temps pour m'expliquer des choses, pour discuter et pour partager avec autant de passion tes hobbies, tes opinions et connaissances. Je profite ici pour remercier aussi à la communauté Hawkes avec qui j'ai pu partager de nombreuses conférences, séminaires, discussions et repas, que ce soit à Toulouse, Lille, Paris ou ailleurs (car au final on est un peu partout).

Je remercie tous les membres du LPSM qui animent la vie scientifique de cette université avec vos personnalités aussi variées qu'uniques et bienveillantes. Merci Claire de m'avoir intégré dans la vie de ce labo depuis mon époque en tant que stagiaire, et de nous avoir fait sentir, moi et les autres doctorants, comme de vrais collègues. Bon courage dans ta nouvelle aventure. Merci Stéphane de toujours avoir un sourire, de toujours partager une bonne discussion ou tes connaissances. J'espère que ces années où nous avons été voisins, où tu as eu affaire à de la musique, aux rires et aux chants, n'auront pas été trop désagréables. Merci Antoine de toujours être là pour rappeler aux doctorants que parfois il faut arrêter de bosser et profiter de la vie dehors. Merci Charlotte de ta gentillesse et d'avoir organisé le séminaire de Hawkes. J'ai beaucoup aimé faire partie du groupe de TD de la L3 Statistiques avec toi. Merci aux autres membres: Anna BH, Jean-Patrick, Maud, Erwan, Sylvain, Lorenzo (et celles et ceux que j'aurai oublié ici).

Merci aux membres de l'administration du labo, Valérie, Nathalie, Marouane. En particulier, merci Louise de m'avoir accompagné dans mon stage et dans mes premières années de thèse. J'ai adoré chaque moment où nous avons papoté, merci infiniment de toujours me demander comment tu pouvais aider car tu rendait ma vie au travail tellement plus agréable. Merci Hugues, Atlas portant le monde informatique de ce labo à base de mails avec ton humour hilarant et singulier. Il n'y a rien de plus drôle que de lire que tu ne pouvais pas faire un truc "dans l'immédiat" mais que dans une heure tout serait prêt. Si j'avais été aussi efficace que toi j'aurais sûrement publié 77 papiers de plus. Merci aussi aux membres du ménage qui nous permettent de travailler dans les meilleures conditions chaque jour.

Dans toute aventure on rencontre des gens qui nous accompagnent et avec qui nous partageons nos réussites, notre bonheur, notre désespoir et, surtout, du café... beaucoup de café. Camila, Cami, je ne peux pas expliquer à quel point je suis le plus heureux d'avoir partagé le bureau avec toi; rien de mieux que d'avoir un petit morceau de l'Amérique Latine aussi proche. Je chéris chaque moment de chant et de musique, de s'envoyer des memes, de coller le visage de l'autre dans nos chaises et, en particulier, de crier de joie à chaque rencontre, peu importe si c'était après quelques heures ou plusieurs semaines sans se voir. Je me souviens parfaitement de à quel point on est devenus proches à Porquerolles et l'été d'après où l'on a passé à danser ensemble chaque vendredi soir aux quais. J'admire tellement ton intensité et ta détermination à faire tout ce qui te plaît en t'investissant à fond. J'espère que l'on continuera nos sessions de shopping, de karaoke, de ragot et de nourriture pendant longtemps car la vie avec toi est tellement amusante et agréable. Je t'aime plus qu'à notre déesse Shaki. Iqraa, merci pour toutes ces discussions, parfois très philosophiques, au cinquième étage ou dans un café perdu dans Paris. Tu es une personne avec un coeur énorme et qui sait être fidèle à ses principes et à ses croyances: que tu saches que c'est une chose que j'apprécie énormément chez toi. Merci, car je peux compter sur toi pour danser et chanter jusqu'à l'épuisement. Merci d'avoir partagé un Airbnb avec moi pendant la conférence à Lyon où l'on restait jusqu'à tard le soir à discuter, j'ai adoré vivre avec toi ce petit bout de temps. En particulier, merci pour toutes ces fois où tu m'as écouté et tu m'as

tendu les bras pour me câliner. Ariane, je suis content de nous être approchés autant avec les années et de savoir que je fais partie des chanceux qui peuvent te câliner (avec modération). Je pense en particulier à cette nuit qu'on a fini par manger Burger King, et à cette discussion plein d'émotions à Bruxelles assis au bar d'un restaurant japonais. Parmi mes meilleurs souvenirs avec vous toutes il y a tous ces ragots, toutes ces discussions d'amour, de désamour, tous ces rires et pleurs.. Maintenant que nous sommes tous et toutes partis du labo il faudra bien que nous retrouvions où continuer les films de l'été ! Je veux que vous sachiez que votre amitié est un cadeau que je chéris, et que votre amour et affect remplissent de couleur et de vie mon quotidien.

Maintenant le spotlight se tourne vers deux personnes que j'ai rencontrées dans le Master. Claire, Clairounette, Claricot, ma chère Claire, je suis tellement mais tellement heureux de t'avoir rencontré, de t'avoir proposé de faire ton stage dans mon bureau car ça a marqué le début d'une amitié qui, aujourd'hui, est l'une de plus chères pour moi. Tu as plus d'énergie que le Soleil même, je te vois courir de droite à gauche (parfois littéralement), que ce soit pour partager un jeu de société en hiver ou la plage en été (merci beaucoup beaucoup beaucoup de m'avoir fait découvrir Royan). Je porte dans moi toutes ces longues marches ensemble dans Paris à partager des idées, des opinions, des ragots et de la nourriture... beaucoup de nourriture. Je te remercie aussi pour tous tes conseils et de toujours être aussi directe avec moi. Peu importe où tu seras, tu seras toujours à mes côtes. Je t'envoie mes meilleurs souhaits dans ce nouvel épisode qui démarre pour toi. Ludovic, j'espère que tu es en train de lire ceci mes remerciements en espagnol. Ludo, j'adore voir ta manière de vivre la vie avec un sourire, avec un calme propre à toi, avec un aura tellement chill. Tu as toujours su rigoler avec moi et partager ta passion pour la culture, l'histoire et les bébés moches de l'art sacré. Surtout, merci d'avoir dansé avec moi autant de nuits dans les quais (et je te rassure, je n'ai plus mal au coude).

Merci Alexis. Merci pour ta nourriture aussi délicieuse, Merci pour toutes les soirées partagées chez toi, en soirée techno, ou même dans les soirées pop (auxquelles tu m'accompagnais même si tu disais les détester). Je veux te remercier d'avoir pris soin de moi, pour tout l'amour et l'amitié que tu m'as donné, pour tous les verres, les discussion, les ragots, les graphes, les conseils, les câlins. Te rencontrer est l'une des meilleures choses qui me sont arrivées pendant le doctorat. Continue de vivre avec la même intensité et passion avec laquelle la techno te fait danser. Antonin, mi ciela, je suis heureux de t'avoir dans ma vie, de rire autour de ces edits des stars pop, de pouvoir chanter Beyoncé de toutes nos forces. Merci d'être toujours là pour moi. Peu importe ce qui se passe, vous occupez un endroit spéciale dans mon coeur et j'espère pouvoir vous garder proche de moi pour des années à venir.

Je remercie une doctorante aussi dédiée que féministe et hyperactive: Adeline. Je ne pensais pas rencontrer quelqu'un qui pense et qui parle aussi vite que moi, nos discussions dans mon bureau (ou celui du chef) étaient chaotiques, drôles et remplies de conseils d'amour. Merci beaucoup car tu as su venir et me donner ton support quand j'en avais beaucoup besoin. Je me souviens très bien de notre soirée à regarder Batman.

Il y a eu un été où j'ai rencontré deux stagiaires avec qui j'ai beaucoup partagé, et partage encore aujourd'hui. Ça me fait rire de savoir que, peu importe ce que j'écrirai ici, ils viendront tous les deux m'embêter, et pour cela je vous adore. Ángel, nos discussions sans fin sont un trésor pour moi. J'aime savoir que je peux discuter de tout et de rien avec toi pour plus tard aller danser autour de la pop ou de la techno (même s'il faut s'occuper de toi après). Tu es ma niña fresa. Eyal, on t'a peut-être beaucoup imposé la musique pop mais je vois que cela a fait ses fruits donc je ne regrette rien. Merci pour toutes les fois où nous sommes allés manger, prendre un verre et danser. Je sais que le bullying est ta façon d'éprouver ton amitié, je l'ai bien appris à cette soirée sur ta terrasse. Je sais que je peux compter sur toi et je veux que tu saches que c'est réciproque. D'une certaine façon vous prenez le relais, et bien que vous vous sentiez être grands et puissant je serai toujours là pour vous rappeler que vous portiez des couches quand je vous ai rencontrés.

Pierre, merci beaucoup de m'avoir fait autant rire pendant les surveillances (tu sais de quoi je parle). Merci Ferdinand pour les nuits entre musique en russe et musique en espagnol. Nina, ton arrivée au labo a été comme voir le soleil sortir après les longs hivers en France. Tu es une personne avec une énergie tellement positive et sassiness que je ne peux pas m'empêcher d'adorer ta compagnie.

Merci à tous les autres doctorants passés et présents du LPSM avec qui on a passé tellement de pauses ensemble, tellement d'histoires de bonheur, de tristesse et de (des)espoir. Merci Francesco de toujours avoir un sourire, Grace de toujours venir venir faire coucou dans mon bureau, Paul pour les bars et les soirées surtout à Bordeaux, Romain pour les discussions de Hawkes. Merci au trio mythique de Nicklas, Gloria et Thibault avec qui je me suis amusé à de nombreuses reprises autour de cafés, bières et vins fancy, comme à Lyon. Merci Antonio, Yazid, Gabriel, Lucas, Barbara, Mathis, Alessandro, Anna, Arthur, Stan.

Haydee, beibi, merci beaucoup d'être la personne que tu es et de toujours rire de toutes tes forces avec moi. Je chéris énormément ces nombreux souvenirs aussi divers que l'on partage. Après 11 ans d'amitié, cela me paraît fou de voir à quel point on est toujours aussi connectés, à finir les phrases et les blagues de l'autre. Ton amitié c'est de l'or pur pour moi et je te remercie tellement de me laisser être aussi proche de toi. Je t'adore.

Tim, j'adore notre amitié qui persiste au fil des années, peu importe le temps que nous passons sans se voir. Je sais que tu seras toujours là pour discuter, rire et m'aider. C'est déjà 8 ans de se connaître et je rigole encore en pensant à tout ce que l'on a partagé à Toulouse.

Romain, je te remercie pour toutes ces années ensemble, tout ce que nous avons vécu et pour toute la maturité que l'on a acquise ensemble. Je n'oublierai jamais tous ces bancs qui ont été ni tous ces films et séries que nous avons dévorés. Je porte avec moi tout ce que nous avons été, et merci de m'avoir accompagné dans le début de ce voyage. J'espère que tu continues de rire fort et d'être heureux.

Le travail qui conclut aujourd'hui a commencé en quelque sorte en 2015 quand je suis arrivé en France et je dois parler d'une personne avec qui tout a commencé : Fernanda. Nous sommes arrivés ensemble et on a été le pilier l'un pour l'autre pendant tellement longtemps, entre appels de 5 heures et sorties jusqu'à 6h du matin (avec son respectif repas avant de dormir). Je sais que sans toi je ne serais pas resté ici et je n'aurai pas accompli autant, car tu étais toujours là pour m'écouter. Ça me manque tous ces cafés à travailler à côté de notre "pote" de café qui passait à lire et à écrire. Je suis content de voir ce que tu es devenue, et que tu saches que même si tu es loin physiquement, je te sens toujours proche de moi.

Alfredo, je suis vraiment content de s'être retrouvés et de s'être rendu compte que nous nous cherchions autant depuis le lycée. Je suis heureux de tes réussites et surtout de voir comment tu continues d'évoluer et d'aller vers l'avant. Continuons de danser, chanter et critiquer mi cochina.

C'est le moment de remercier la personne la plus importante au monde pour moi. Mi señora madre, maman, hermosa. Merci beaucoup de m'avoir donné la vie, de m'avoir élevé, d'avoir fait de moi la personne forte, dédiée et surtout heureuse que je suis. C'est grâce à toi que mes mains sont remplies de tous les fruits que j'ai pu semer pendant ma vie. J'espère pouvoir te donner tout ce que tu mérites pour tout l'amour que tu m'as donné et pour tous les sacrifices que tu as faits. Tout ce que je suis, ce que j'ai, que je souhaite c'est pour toi, vis pour toi, respire pour toi. T'avoir aussi loin de moi implique que tu me manques à un point qui parfois m'est insupportable mais je sais que tu seras toujours ici avec moi, et que tu saches que je serai toujours là pour toi. Je t'aime avec tout mon être.

Lidi, merci énormément d'avoir été comme une deuxième mère pour moi, merci de toujours m'avoir fait sentir comme un de plus dans ta famille, tu m'as donné tellement d'amour que parfois je ne sais pas comment te le donner. Tu as été la première à me montrer ce monde de l'autre côté du "charco". Tu as toujours été là pour me câliner, me gâter et m'encourager dans tellement de choses. Je t'aime idi.

Fa, papa, merci beaucoup pour tout ce que tu m'as donné, pour la patience et l'amour que tu as toujours eu pour moi. Quelque chose qui m'a toujours donné, et qui me donne encore aujourd'hui tellement de sécurité est de savoir que je peux toujours compter sur toi, que tu seras toujours là quand j'en ai besoin, et que, même si nos caractères choquent constamment (je l'ai hérité de toi hahaha), tu m'aimes toujours de toutes tes forces. Je t'aime.

Merci beaucoup Mar car grandir à tes côtés a été comme avoir une soeur qui me protégeait et qui prenait soin de moi. Tu ne sais pas à quel point tu m'as rendu heureux avec toute la gentillesse que tu éprouvais pour moi. Grâce à toi j'ai compris tellement de choses et ma vie ne serait pas la même aujourd'hui sans cela. Marco, German, merci de m'avoir traité comme un frère (avec tout le bon et mauvais) parce que grâce à ça je me sentais moins seul.

Mari, prima, Mariana, ton âme, ta présence, ton être me font sentir tellement accompagné dans cet océan gigantesque que nous appelons la réalité. Nous pensons, ressentons et vivons la vie d'une façon tellement similaire que je suis sûr que c'est ce qui nous a rapproché autant depuis que nous étions enfants, et c'est ce qui a créé un lien aussi fort. Merci pour chaque seconde que tu as su me donner pour m'écouter, me soutenir et me donner tes conseils. Je t'aime tellement mais tellement que j'aimerai avoir ta plume pour pouvoir le décrire avec des mots. Continue de suivre ton feu intérieur et n'oublie jamais qui tu es, ce que tu as accompli et tout le pouvoir que tu as, parce que tu as changé ma vie radicalement.

Nay, Ive, je remercie la vie de nous avoir rapproché autant quand elle l'a fait. Je suis heureux de vous avoir comme mes soeurs et de l'amour que vous éprouvez envers moi à chaque occasion que vous avez. Danser avec vous, chanter avec vous, fêter avec vous a toujours eu une chaleur singulière qui me remplit de joie. Je vous adore.

Palola, Pola, Dude, mon enfance n'auraient pas été une vraie enfance sans toi, et mes poumons ne seraient pas aussi coriaces sans toutes les heures que l'on a passé à rire. Je ne crois pas connaître quelqu'un qui m'ait infligé autant de maux de ventre avec toutes ses bêtises. Ma tête est remplie de souvenirs à se disputer, à se pardonner, à faire des bêtises qui énervaient nos mères et faisaient rire nos grand-parents. Tu me manques beaucoup et je t'aime encore plus.

Je remercie mes tantes car, comme beaucoup le savent, c'est grâce à toutes ces femmes fortes que je suis qui je suis aujourd'hui. Merci beaucoup Lale de t'occuper de moi autant, et merci aussi pour tous ces films, ces telenovelas et ces rires avec Pao et ma mère. Maruka, nos nuits avec les "cositas de la mujer" me manquent tellement. Mes souvenirs sont remplis de petits moments avec toi que peut-être je n'ai pas su exprimer à quel point ils remplissent mon âme de douceur, de chaleur et de calme. Merci. Merci Kenuki y Elsuki car vous avez toujours su me faire sentir comme un enfant, et j'adore ça. Ça me manque beaucoup tous ces noëls dans la cuisine entre rires, cris et, mes préférés, ragots. Merci Ceci de toujours avoir été là et d'avoir partagé ton amour avec moi (et ton amour pour le gâteau à la fraise).

Je veux remercier mes grands-pères et grandes-mères qui ne sont plus avec nous mais qui m'ont toujours donné tout leur amour et câlins, chacun et chacune à sa manière. J'espère que, où vous serez, vous vous sentez bien et fiers.

Je vais finir ces remerciements avec deux personnes qui sont pour moi mes plus grands trésors.

Cosette, Cosa, la reine, ou comme les autres te connaissent, Carolina. Tu es la preuve que la famille nous la construisons, que la famille nous la choisissons, que la famille est faite des personnes qui sont toujours là pour nous. Tu es ma famille, tu es la personne qui me soutient, qui me dispute, qui m'écoute, qui me regarde dans les yeux pour me dire que "tout va se stabiliser", car tu sais que ce sera toujours le cas. Ça fait tellement des années que je te connais que maintenant j'ai l'impression que tu as toujours été avec moi, que tu as toujours partagé ton rire, tes ragots, tes conseils de beauté mais, avant tout, ton être. La vie t'a amené aussi proche de moi et je suis tellement content de cela. Je ne serais pas là aujourd'hui sans toi. Je pourrais écrire mille paragraphes pour te remercier pour chaque belle chose que tu m'as donnée, pour chaque acte d'affection et d'amour, et même si je sais que tu aimerais ça, il n'y a pas assez de papier à Paris pour imprimer un tel manuscrit. Donc je vais tout résumer en une phrase: you are my person.

Gwen. Merci Gwen. Merci d'exister. Merci d'avoir accepté ce tout premier date. Merci pour les rire, les pleurs, les pigeons, les séries, les nuits de films d'horreur et les matins de café à regarder Bob's. Tu es la chose la plus grande que j'ai. Tu es une personne remplie de gentillesse, d'amour, d'humanité et tu ne cesses pas de me montrer ta bienveillance chaque jour depuis que je te connais. Depuis ce premier jour tu as su être un soutien inconditionnel, tu as été là pendant l'écriture de ce manuscrit, et tu as pu voir à quel point ça a été difficile, mais c'est grâce à toi que j'ai réussi. Merci de me cuisiner, de m'écouter, de me câliner. Merci de rester avec moi quand j'en ai besoin, et de me donner la liberté que j'aime tant. Il n'y a pas un seul endroit sur terre où je me sens le plus en sécurité qu'entre tes bras. Merci d'être qui tu es. This nest is beautiful because it's ours.

INFERENCE OF NON-LINEAR OR IMPERFECTLY OBSERVED HAWKES PROCESSES Abstract

The Hawkes point process is a popular statistical tool to analyse temporal patterns. Modern applications propose extensions of this model to account for specificities in each field of study, which in turn complexifies the task of inference. In this thesis, we advance different approaches for the parametric estimation of two submodels of the Hawkes process in univariate and multivariate settings. Motivated by the modelling of complex neuronal interactions observed from spike train data, our first study focuses on accounting for both inhibition and excitation effects between neurons, modelled by the non-linear Hawkes process. We derive a closed-form expression of the log-likelihood in order to implement a maximum likelihood procedure. As a consequence of our approach, we gain access to a goodness-of-fit scheme allowing us to establish ad hoc model selection methods to estimate the interaction network in the multivariate setting. The second part of this thesis focuses on studying Hawkes process data noised by two different alterations: adding or removing points. The absence of knowledge on the noise dynamics makes classical inference procedures intractable or computationally expensive. Our solution is to leverage the spectral analysis of point processes to establish an estimator obtained by maximising the spectral log-likelihood. By deriving the spectral densities of the noised processes and by establishing identifiability conditions on our model, we show that the spectral inference method does not necessitate any information on the structure of the noise, effectively circumventing this issue. An additional result of the study of Hawkes processes with missing points is that it gives access to a subsampling paradigm to enhance the estimation methods by introducing a penalisation parameter. We illustrate the efficiency of all of our methods through reproducible numerical implementations.

Keywords: hawkes processes, parametric inference, identifiability, inhibition, spectral theory, neuronal data

Résumé

Le processus ponctuel de Hawkes est un outil statistique très répandu pour analyser des dynamiques temporelles. Les applications modernes des processus de Hawkes proposent des extensions du modèle initial pour prendre en compte certaines caractéristiques spécifiques à chaque domaine d'étude, ce qui complexifie les tâches d'inférence. Dans cette thèse, nous proposons différentes contributions à l'estimation paramétrique de deux variantes du processus de Hawkes dans les cadres univarié et multivarié. Motivée par la modélisation d'interactions complexes au sein d'une population de neurones, notre première étude porte sur la prise en compte conjointe d'effets excitateurs et inhibiteurs entre les signaux émis par les neurones au cours du temps, modélisés par un processus de Hawkes non-linéaire. Dans ce modèle, nous obtenons une expression explicite de la log-vraisemblance qui nous permet d'implémenter une procédure de maximum de vraisemblance. Nous établissons également une méthode de sélection de modèle qui fournit notamment une estimation du réseau d'interactions dans le cadre multivarié. La deuxième partie de cette thèse est consacrée à l'étude des processus de Hawkes bruités par deux types d'altérations : l'ajout ou la suppression de certains points. Le manque d'information lié à ces mécanismes de bruit rend les méthodes classiques d'inférence non-applicables ou numériquement coûteuses. Notre solution consiste à s'appuyer sur l'analyse spectrale des processus ponctuels afin d'établir un estimateur obtenu en maximisant la log-vraisemblance spectrale. Nous obtenons l'expression des densités spectrales des processus bruités et, après avoir établi des conditions d'identifiabilité pour nos différents modèles, nous montrons que cette méthode d'inférence ne nécessite pas de connaître la structure du bruit, contournant ainsi le problème d'estimation. Notre étude sur les processus bruités donne accès à une méthode de sous-échantillonnage qui nous permet d'améliorer les approches d'estimation en introduisant un paramètre de pénalisation. Nous illustrons la performance des différentes méthodes proposées à travers des implémentations numériques reproductibles.

Mots clés : processus de hawkes, inférence paramétrique, identifiabilité, inhibition, théorie spectrale, données neuronales

Sorbonne Université – Campus Pierre et Marie Curie – 4 place Jussieu – 75005 Paris – France

Contents

A	grad	lecimientos	vi
R	leme	rciements	xi
A	$\Delta bstra$	act	xvii
C	Conte	nts	xviii
1	Intr	oduction	1
	1.1	Statistics for Hawkes processes	2
		1.1.1 The self-exciting point process	2
		1.1.2 Inference and applications	4
		1.1.3 Challenges and contributions	7
	1.2	Factoring in inhibition for Hawkes processes	8
		1.2.1 Context and related works	8
		1.2.2 Maximum Likelihood Estimation for Hawkes Processes with self-excitation	
		or inhibition	9
		1.2.3 Inference of multivariate exponential Hawkes processes with inhibition and	
		application to neuronal activity	10
	1.3	Something is amiss: spectral methods for imperfect data	12
		1.3.1 Context and related works	12
		1.3.2 Spectral analysis for the inference of noisy Hawkes processes	13
		1.3.3 A numerical exploration of thinned Hawkes processes through spectral theory	14
	1.4	Outline of the manuscript	15
2	An	introduction to Hawkes processes study	16
	2.1	Introduction	16
	2.2	Random measures and point processes	17
		2.2.1 Random measures	17
		2.2.2 Point processes	18
		2.2.3 Integrals and moment measures	19
		2.2.4 The conditional intensity function	20
	2.3	Hawkes process	22
		2.3.1 The self-exciting Hawkes process	22
		2.3.2 The multivariate Hawkes process	23

	 2.4 2.5 	2.3.3 Non-linear Hawkes processes23Inference for Hawkes processes242.4.1 MLE for Hawkes process with exponential kernel242.4.2 Least-squares minimisation262.4.3 A Bayesian estimation approach26Simulation272.5.1 Ogata's thinning simulation272.5.2 Simulation through branching theory28
3	Max	cimum Likelihood Estimation for Hawkes Processes with self-excitation
	or i	nhibition 31
	3.1	Introduction
	3.2	The Hawkes process
	3.3	Maximum likelihood estimation and the exponential model
	3.4	Goodness-of-fit
	3.5	Numerical Results 36
	3.6	Discussion
	3.A	Proof of Proposition 3.3.2
	3.B	Simulation algorithm
4	Infe	rence of multivariate exponential Hawkes processes with inhibition and
1	ann	lication to neuronal activity
	4.1	Introduction 41
	4.2	The multivariate Hawkes process
		4.2.1 Definition
		4.2.2 Related work
	4.3	Estimation and goodness-of-fit
		4.3.1 Introductive example
		4.3.2 Underlying intensity and restart times in the multivariate setting 46
		4.3.3 Identifiability and likelihood computation
		4.3.4 On identifiability conditions
		4.3.5 Recovering the graph of interactions
		$4.3.6 \text{ Goodness-of-fit} \qquad 53$
	4.4	Illustration on synthetic datasets
		$4.4.1 Simulation procedure \dots 54$
		4.4.2 Proposed methods and comparison to existing procedures
		4.4.3 Robustness on misspecified models
	4.5	Application on neuronal data
		4.5.1 Preprocessing and data description
		$4.5.2 \text{ Resampling } \dots $
		4.5.3 Goodness-of-fit and multiple testing procedure
	16	4.0.4 Estimation results
	4.0	Discussion
	4.A 1 R	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
	4.D	Proof of Theorem $4.3.1$ 70
	4 D	Proof of Corollary 4.3.1
	4.E	Algorithm for computing the log-likelihood
	4.F	Reconstructed interaction functions for synthetic data

5	Spee	ctral analysis for the inference of noisy Hawkes processes	77
	5.1	Introduction	78
		5.1.1 Mathematical setting	79
		5.1.2 Related works	80
	5.2	Spectral analysis	81
		5.2.1 The Bartlett spectrum	81
		5.2.2 Superposition of processes and noisy Hawkes process	82
	5.3	The univariate noisy Hawkes process	84
		5.3.1 General setting	84
		5.3.2 Exponential model	84
		5.3.3 Beyond the exponential model	86
	5.4	The bivariate noisy Hawkes process	86
	5.5	Numerical results	89
	0.0	5.5.1 Universate setting	89
		5.5.2 Bivariate setting	01
	56	Discussion	96
	5 A	Proof of Proposition 5.2.1	90 96
	5 R	Proof of Proposition 5.3.1	08
	5.C	Proof of Proposition 5.3.9	100
	5.0	Proof of Proposition 5.3.2	100
	5.D 5.F	Proof of Proposition 5.4.1	102
	5.E	Proof of Proposition 5.4.2 Situations 1 and 2	104
	5.C	Proof of Proposition 5.4.2, Situations 1 and 2	107
	J.G	1 Tool of 1 Toposition 5.4.2, Situations 5 and 4	109
6	A n	umerical exploration of thinned Hawkes processes through spectral the-	
	\mathbf{ory}		111
	6.1	Introduction	112
	6.2	Mathematical setting	113
		6.2.1 The <i>p</i> -thinned point processes \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	113
		6.2.2 Spectral theory for point processes	114
	6.3	Parametric estimation of a thinned process	115
		6.3.1 The spectrum of a thinned point process	115
		6.3.2 The <i>p</i> -thinned Hawkes process estimation and the exponential kernel	116
	6.4	Numerical illustrations	118
		6.4.1 Spectral estimator for missing data	118
		6.4.2 <i>p</i> -thinning as a subsampling method	119
	6.5	Discussion	122
	6.A	Proof of Proposition 6.3.1	122
	$6.\mathrm{B}$	Proof of Proposition 6.3.2	124
7	Disc	cussion	127

Bibliography

129

Introduction

This chapter is a general introduction to Hawkes processes and the challenges explored in this manuscript. After a succinct presentation of Hawkes processes with excitation, we contextualise the state-of-the-art literature concerning estimation methods in Section 1.1 and present the main questions that guide our research. This allows us to exhibit the two paradigms that are studied in this work: inhibition and imperfect data. Section 1.2 pertains to the parametric estimation of both univariate and multivariate Hawkes processes with potential inhibiting interactions (detailed in Chapters 3 and 4). Section 1.3 presents our contributions to the study of imperfectly observed exciting Hawkes processes (detailed in Chapters 5 and 6). A general outline of this manuscript is described in Section 1.4.

Outline of the current chapter

1.1	Statistics for Hawkes processes	2
	1.1.1 The self-exciting point process	2
	1.1.2 Inference and applications	4
	1.1.3 Challenges and contributions	7
1.2	Factoring in inhibition for Hawkes processes	8
	1.2.1 Context and related works	8
	1.2.2 Maximum Likelihood Estimation for Hawkes Processes with self-excitation	L
	or inhibition \ldots	9
	1.2.3 Inference of multivariate exponential Hawkes processes with inhibition	
	and application to neuronal activity	10
1.3	Something is amiss: spectral methods for imperfect data	12
	1.3.1 Context and related works	12
	1.3.2 Spectral analysis for the inference of noisy Hawkes processes	13
	1.3.3 A numerical exploration of thinned Hawkes processes through spectral	
	theory	14
1.4	Outline of the manuscript	15

1.1 Statistics for Hawkes processes

1.1.1 The self-exciting point process

In probability and statistics, modelling random collections of points in a certain space is commonly done through a point process. When studying point processes on the real line \mathbb{R} (or the half line $\mathbb{R}_{\geq 0}$), the natural order of this space often incurs an ordering of any countable sequence of points $(T_k)_{k \in \mathbb{Z}}$: we talk then of *temporal* point processes.

In statistics, it is a common question to analyse the dynamics describing the occurrences of a certain phenomenon: the time of arrival of buses at a bus stop, the apparition of symptoms in a population or earthquake incidents in a region of the world. The simplest model for point processes is the homogeneous Poisson process, where the waiting times between any two event times are i.i.d distributed as an exponential random variables with parameter $\lambda > 0$, known as the intensity of the process. A natural extension of this model is obtained by allowing the intensity to be a deterministic non-negative function $\lambda : \mathbb{R} \to \mathbb{R}_{\geq 0}$, adding a temporal dependence on point arrivals.

The Hawkes process

In 1971, Alan G. Hawkes introduces a past-dependent model, initially called self-exciting point processes (Hawkes 1971), which will later be known as the Hawkes point process. Let $\mathcal{H}_t = \sigma(\{T_k \mid k \in \mathbb{Z}, T_k \leq t\})$ denote the past history of a process N for any $t \in \mathbb{R}$, we define the conditional intensity function $\lambda : \mathbb{R} \to \mathbb{R}_{>0}$ of this process as:

$$\lambda(t \mid \mathcal{H}_t) = \lim_{h \to 0} \frac{\mathbb{E}[N([t, t+h]) \mid \mathcal{H}_t]}{h}.$$

The Hawkes process is then defined by the following expression of the conditional intensity function:

$$\lambda(t \mid \mathcal{H}_t) = \mu + \sum_{T_k \le t} h(t - T_k) \,. \tag{1.1}$$

Remark. In the literature, it is common to omit writing the history \mathcal{H}_t as it is directly implied that λ has access to the entire past of the process. We follow this convention throughout this work, unless marked otherwise.

The term μ is often referred to as the baseline intensity which dictates a constant rate of occurrences, similar to a homogeneous Poisson process. The dependence on the past is represented by the second term in Equation (1.1), where each event time that precedes t contributes to the intensity function through the interaction function $h : \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$, also calles the kernel.

Positivity of h represents an excitation effect between points: each point T_k in the past increases the value of the intensity function, which in turn increases the rate at which points in the future appear. For stability reasons, h is assumed to converge to 0 as $t \to +\infty$, representing the rate at which the effects from the past are "forgotten".

Different shapes of h allow to account for different effects. On the one hand, strictly decreasing functions such as exponential (Ozaki 1979; Ogata 1988) or power-law kernels (Zhang 2016) represent instantaneous effects, which appear as spikes in the conditional intensity function (Equation (1.1)). On the other hand, Rayleigh or gamma kernels (Lesage et al. 2022), where the maximum of the function is not at the origin t = 0, can be chosen to represent delayed effects. Figure 1.1 illustrates two Hawkes processes with two different kernels. The properties of the conditional intensity function are intrinsically connected to those of h, and so when working





Figure 1.1: Kernel functions (top) and respective conditional intensity functions (bottom) of a Hawkes process started at t = 0 with baseline intensity $\mu = 1$.

Clustering and branching structures

A direct consequence of the expression of the intensity function (Equation (1.1)) is that the Hawkes process model can be interpreted as a Poisson cluster process (Hawkes et al. 1974). The most practical way of defining a cluster process (Bartlett 1963) is from a generative point of view. We begin by simulating a homogeneous Poisson process on the real line \mathbb{R} with parameter μ : these points T_k^c are often called ancestors, immigrants or cluster centers. Each ancestor generates an inhomogeneous Poisson process with intensity function $h(\cdot - T_k^c)$, forming a family of children points, also called descendants. The iteration is repeated with each new point generating its own subprocess until no descendants are generated. In the end, the cluster process is formed by the union of both ancestors and descendants. The particularity of a Hawkes process is that the support of the function h is a subset of $\mathbb{R}_{\geq 0}$, meaning that each occurrence influences solely the future of the process.

This kind of process is also known as the Poisson branching process (Lewis 1964) as it describes a generation dynamic similar to the Galton-Watson branching process (see Watson et al. (1875) for the discrete time version and Harris (2002, Chapter III) for the generalised version). More precisely, the Hawkes process can be seen as a time-continuous branching process with immigration: let us assume we observe the arrival of an immigrant at a time T_k^c . This immigrant will produce a first generation of children, forming the first set of points, or branches. Then, each child becomes a parent and generates a new set of branches. A tree is then formed by the union of each immigrant and all of its branches, and then the Hawkes process is once again formed by the union of all trees.

A visualisation of both structures is illustrated in Figure 1.2 for a Hawkes process with

exponential kernel. The clustering (equivalently branching) structure is one of the properties that made Hawkes processes so attractive in the literature. From a theoretical point of view, the many existing developments of branching theory allowed to quickly obtain results concerning existence, stability and stationarity. A main example concerns the existence of a self-exciting Hawkes process: in order for the point process to have a finite number of points inside any bounded set, a necessary and sufficient condition (Hawkes et al. 1974, Lemma 1) is:

$$\int_0^{+\infty} |h(t)| \, \mathrm{d}t < 1$$

which is derived from the subcriticality condition of Galton-Watson processes.

From a numerical point of view, the definition of a Poisson cluster process, as given above, provides a simulation algorithm. An inference method inspired from branching process theory was adapted to the study of Hawkes processes (Veen et al. 2008).

The aforementioned properties, and the quick developments that followed in the literature, created a growing interest in other scientific fields. From this, different versions of the original self-exciting Hawkes process appeared in the literature, each one trying to accommodate to more complex phenomena.

The multivariate Hawkes process

Probably the most natural and useful extension is the definition of the *multivariate* Hawkes process. Instead of studying a single phenomenon (a univariate process), we define a *d*-variate Hawkes process as *d* individual point processes $(N_i)_{i=1:d}$ interacting together, where each process is defined by a conditional intensity function λ^i :

$$\lambda^{i}(t) = \mu_{i} + \sum_{j=1}^{d} \sum_{T_{k}^{j} \leq t} h_{ij}(t - T_{k}^{j}), \quad \text{for any } t \in \mathbb{R}$$

In this formulation, each process N_j has its own event times $(T_k^j)_{k\in\mathbb{Z}}$ and baseline intensity $\mu_j > 0$. This model introduces interactions between processes, which appear in the second term of the intensity. The kernel functions h_{ij} represent the effect of points from N_j on process N_i . This model allows to study a group of individuals linked by a network of interactions, adding a new dimension of study for such processes.

From modelling infection spread all the way to analysing clusters of earthquakes, the study of Hawkes processes from a statistical point of view quickly became a central topic of interest.

1.1.2 Inference and applications

Statistical estimation

Statistical challenges for Hawkes processes with excitation focus around estimating both the baseline intensity μ and the interaction function h.

To our knowledge, the very first paper implementing an estimation procedure in a parametric framework is Adamopoulos (1976). In his work, the author proposes a study of exponential kernels for both univariate and bivariate Hawkes processes through the spectral log-likelihood, closely related to time series theory. The exponential kernel is often chosen as a parametric interaction function because in this case the univariate Hawkes process becomes a Markov process which presents the advantage of simplifying the expression of the intensity function. This is



Figure 1.2: Illustration of both branching (top) and clustering (bottom) structures of an exponential Hawkes process (middle). Each color represents a single cluster/tree. Simulation is done thanks to the cluster process algorithm.

exemplified in the implementation of the maximum likelihood estimator (MLE) in Ozaki (1979). The expression of the log-likelihood of a general point process, for an observation in [0, T], is:

$$\ell_T(\theta) = -\int_0^T \lambda(u) \,\mathrm{d}u + \sum_{k=1}^{N(T)} \log(\lambda(T_k^-)), \qquad (1.2)$$

where N(T) is the number of points in the observation window. Other works address estimation in the parametric setting, such as Ogata (1988) for other kernel functions using the MLE, Veen et al. (2008) leveraging the branching structure, Bacry et al. (2020) minimising a least-squares contrast and Da Fonseca et al. (2013) via the method of moments.

The literature regarding non-parametric inference is also vast with implementations of the MLE in Guo et al. (2018) and a penalised variant optimised by an Expectation-Maximisation algorithm. The least-squares minimisation is often used as shown in Reynaud-Bouret et al. (2009), Eichler et al. (2016), and Kirchner (2017). Other methods include solving Wiener-Hopf equations (Bacry et al. 2016) or approximating the interaction functions with autoregressive models (Kirchner 2017).

All of these references concern frequentist approaches of statistics, and so it is essential to mention that an equal effort has been made from a Bayesian point of view. Rasmussen (2013) proposes two procedures through log-likelihood estimation: one through the classical intensity function and another similar to Veen et al. (2008) with the branching structure. In Lemonnier et al. (2014), the authors propose an approximation through exponential kernels and taking advantage of the Markovian properties. The multivariate case is deeply studied in Donnet et al. (2020) with illustrations on estimating the underlying interaction graph.

This is jus a small sample of the plethora of approaches that have been developed in order to study these kind of processes.

Applications and generalisations

The temporal-dependency structure of Hawkes process has motivated its use accross a variety of application fields with many generalisations to obtain more explicative models.

The historical example is the study of seismic activity where the occurrence of an earthquake is usually followed by a number of smaller tremors known as aftershocks. This behaviour is similar to the self-exciting effect modelled by the Hawkes process and associated with the clustering structure, as shown in Adamopoulos (1976). As in seismology the spatial placement and magnitude of each quake are important factors to take into account, applications in this field tend to include this information via the concept of marked point processes. Each event time has an associated mark representing the detected magnitude (Ogata 1988) and the location of each epicenter (Ogata 1998; Kwon et al. 2023).

A similar approach is proposed in the study of social media interactions for the analysis of subject trends. The excitation effect appears in the form of reposting where users have the option to share a news post among their followers which in turn may continue to spread the information in their social circles. In this context, the impact of each repost is dependent on the influence of the account and the impact of a topic on a certain community. This is represented for example by including information in the form of number of followers (Mishra et al. 2016) again through marks, or by adding an additional dimension to the process (Pinto et al. 2015) to represent the impact of a topic over another.

In criminology, it is commonly assumed that delinquent acts may incite other crimes in a population, which can be modelled by a self-exciting Hawkes process. As a branch of social sciences, it is important to account for different factors that affect human behaviour such as time of the day, demographics and social interactions. A way of accounting for these effects is to adapt the Hawkes process to time and spatial dependent baseline intensities. A time-varying baseline intensity can represent periodic criminal behaviour (Lewis et al. 2011) with higher values for nightime. A space-dependent baseline is used to represent the link between residential density and burglary (Mohler et al. 2011) or the spatial distributions of gangs in a city (Linderman et al. 2014).

Many other fields include finance (Embrechts et al. 2011; Bacry et al. 2013; Roueff et al. 2019; Lotz 2024), genomics (Reynaud-Bouret et al. 2009; Carstensen et al. 2010), ecology (Denis et al. 2024; Nicvert et al. 2024), epidemiology (Rizoiu et al. 2018; Chiang et al. 2022), TV browsing behaviour (Xu et al. 2016), event data streams in football (Baouan et al. 2023; Narayanan et al. 2023), each one often accompanied with new formulations of the Hawkes process.

1.1.3 Challenges and contributions

Our work is motivated by the applications of Hawkes processes in neurobiology for the study of neuronal activity data (Reynaud-Bouret et al. 2013; Lambert et al. 2018; Duval et al. 2022). Neurons in the brain are connected through a deep network of synapses allowing them to communicate through electrical impulses usually studied by measuring the action potential or spikes emitted as the membrane potential of the cells is depolarised. These measurements generate a spike train of instants when a neuron activates and influences the membrane potential of connecting neighbours.

These exchanges can appear either as excitatory or inhibitory to respectively incite or stop other neurons from activating. The first effect can be clearly modelled by a multivariate Hawkes process but the original modelling does not account for inhibiting effects. In practice, this requires to allow for the interaction functions h_{ij} to take negative values, introducing the concept of Hawkes processes with inhibition.

In order to guarantee the non-negativity of the intensity function, a common practice in the literature is to consider the concept of *non-linear Hawkes processes*. The main difficulty is that the branching structure of Hawkes processes is not valid any longer and the intensity function presents more complex behaviour, so pre-established inference methods are not accessible anymore. The first part of this manuscript pertains to the study of inhibiting Hawkes processes in order to propose estimation methods in the parametric setting. We present an overview of the studied model and our contributions in Section 1.2.

The second part of this manuscript is focused around studying imperfect data. Collecting spike trains data is a procedure that can present measurement errors, which can appear as missing neuronal activation instants or by attributing spikes to the wrong neuron.

This setting represents a common statistical framework of accounting for missing data in an observed process. In our context, not having a full knowledge on the past history of the process makes the conditional intensity function intractable. Our contributions, as summarised in Section 1.3, are focused on exhibiting an inference paradigm through the spectral analysis of point processes for observations noised either by adding points from an external process or by deleting event times.

Although the models inbetween chapters may differ, our contributions follow a common thread by addressing the following four questions.

- What approaches can we take to establish parametric estimation procedures for more complex Hawkes processes dynamics?
- Under which conditions are our statistical models identifiable?
- How to evaluate and select the best estimators through data-driven methods?
- Which modern statistical tools can we leverage to improve our inference procedures?

1.2 Factoring in inhibition for Hawkes processes

1.2.1 Context and related works

The original Hawkes process was proposed as a way of modelling the effect of excitation between points, which is characterised by the positivity of the interaction function h. Our first goal in this work is to study the opposite effect which is commonly known in the literature as *inhibition*.

An inhibiting Hawkes process consists in modelling repulsion between points: each event will reduce the chances of others occurring during a certain period of time. Mathematically, a way of translating this effect is by allowing h to take negative values. However, the non-negativity condition on λ prevents us from adopting this approach without adding some constraints.

One of the most common solutions is the *non-linear* Hawkes process model. In the univariate setting, let $h : \mathbb{R}_{\geq 0} \to \mathbb{R}$ and $\Phi : \mathbb{R} \to \mathbb{R}_{\geq 0}$ be two measurable functions. We define a univariate non-linear Hawkes process N on the real half-line $\mathbb{R}_{\geq 0}$ with event times $(T_k)_{k \in \mathbb{N}}$ by the conditional intensity function, for any $t \in \mathbb{R}_{\geq 0}$:

$$\lambda(t) = \Phi\left(\mu + \sum_{T_k \le t} h(t - T_k)\right).$$
(1.3)

By allowing h to be negative, the function Φ has to be a non-linear function. Existence of such processes is ensured as long as Φ is an *L*-Lipschitz function (Brémaud et al. 1996, Theorem 1) such that:

$$L\int_0^{+\infty} |h(t)| \,\mathrm{d}t < 1\,.$$

Multiple choices exist in the literature like a clipped exponential function (Chornoboy et al. 1988; Carstensen et al. 2010; Gerhard et al. 2017), a softplus function (Mei et al. 2017), a sigmoid function (Menon et al. 2018), among others.

In our work, we choose the positive part (ReLU) $\Phi(\cdot) = (\cdot)^+ = \max(0, \cdot)$ like in Lemonnier et al. (2014), Hansen et al. (2015), Lu et al. (2018), and Costa et al. (2020). The intensity function (Equation (1.3)) becomes:

$$\lambda(t) = \left(\mu + \sum_{T_k \le t} h(t - T_k)\right)^+, \qquad (1.4)$$

and the extension to a *d*-variate Hawkes process is:

$$\lambda^{i}(t) = \left(\mu_{i} + \sum_{j=1}^{d} \sum_{T_{k}^{j} \le t} h_{ij}(t - T_{k}^{j})\right)^{+}, \qquad (1.5)$$

for each process N_i .

Our general contribution in the context of Hawkes processes with inhibition is to provide parametric estimation procedures in a frequentist framework through maximum likelihood estimation. To our knowledge, and by the time of publication of both corresponding papers, other frequentist approaches (Reynaud-Bouret et al. 2014; Bacry et al. 2016) do not model inhibition but can provide negative estimations. In Bayesian contexts, works modelling non-linear Hawkes processes focus around the log-likelihood. In Deutsch et al. (2024), the compensator is computed exactly in the case where the interaction functions h_{ij} are exponential with shared decay rate (a similar result to the one presented in Chapter 4, obtained independently during the same period). In practice, an approximation of the compensator through a Simpson's rule is proposed for more general scenarios. This gives access to posterior density estimates in a parametric setting. Works in non-parametric settings include Sulem et al. (2024) that derives posterior concentration rates and Sulem et al. (2023) proposing a variational Bayes method and a sparsity-inducing version based on a model selection paradigm.

1.2.2 Maximum Likelihood Estimation for Hawkes Processes with selfexcitation or inhibition

In Chapter 3, we focus on the study of the univariate Hawkes process N with intensity function described by Equation (1.4). In order to establish the maximum likelihood estimator for a parametrised model of the intensity with parameter $\theta \in \Theta$, it is necessary to obtain a closed-form expression of the log-likelihood:

$$\ell_T(\theta) = -\int_0^T \lambda_\theta(u) \,\mathrm{d}u + \sum_{k=1}^{N(T)} \log(\lambda_\theta(T_k^-)) \,.$$

For Hawkes processes with excitation, the linearity of the intensity function results in an explicit expression of ℓ_T without much trouble, as shown in Ozaki (1979). The main difficulty when including inhibition is that the cumulated effects from the past events may saturate process N, meaning that its intensity is null for a certain period of time. In the work of Lemonnier et al. (2014), the authors propose to approximate the computation in this context by ignoring the non-linear function. This allows them to circumvent this problem at the cost of assuming that the inhibition effects are small enough to be negligible.

In order for us to obtain an exact computation of the log-likelihood that accounts for inhibition, we introduce two novel concepts: the *underlying* intensity function λ^* and the *restart times* $(T_k^*)_{k\geq 1}$. We define $\lambda^* \colon \mathbb{R}_{\geq 0} \to \mathbb{R}$, for any $t \geq 0$, as:

$$\lambda^{\star}(t) = \mu + \sum_{T_k \le t} h(t - T_k),$$

and the restart times T_k^{\star} for any integer k > 0 as:

$$T_k^{\star} = \inf\{t \ge T_k \mid \lambda(t) > 0\}$$

The advantage of working with λ^* instead of λ is that it inherits properties such as smoothness and strict monotonicity from the interaction function h between any two consecutive event times T_k and T_{k+1} . If assumed so, it follows that T_k^* is the moment from which both functions coincide. Our main contribution is to provide a closed-form expression of the compensator $\int_0^T \lambda_{\theta}(u) du$, which we establish in Proposition 1.2.1.

Proposition 1.2.1. For any t > 0:

$$\int_{0}^{t} \lambda(u) \, \mathrm{d}u = \begin{cases} \mu t & \text{if } t < T_{1} \\ \mu T_{1} + \sum_{k=1}^{N(t)-1} \int_{T_{k}^{\star}}^{T_{k+1}} \lambda^{\star}(u) \, \mathrm{d}u + \int_{T_{N(t)}^{\star}}^{t} \lambda^{\star}(u) \, \mathrm{d}u & \text{if } t \ge T_{1} \,, \end{cases}$$
(1.6)

with the conventions that the sum is equal to 0 if N(t) = 1 and the last integral is equal to 0 if $t < T^*_{N(t)}$.

If we consider the exponential kernel $h: t \mapsto \alpha e^{-\beta t}$ for parameters $\alpha > 0$, $\beta > 0$, then the restart times read:

$$T_k^{\star} = T_k + \beta^{-1} \log \left(\frac{\lambda_0 - \lambda^{\star}(T_k)}{\lambda_0} \right) \mathbb{1}_{\lambda^{\star}(T_k) < 0},$$

and, for any integer $k \geq 1$ and any $\tau \in [T_k^*, T_{k+1}]$, each integral simplifies to :

$$\int_{T_k^{\star}}^{\tau} \lambda^{\star}(u) \, \mathrm{d}u = \lambda_0 (\tau - T_k^{\star}) + \beta^{-1} (\lambda^{\star}(T_k) - \lambda_0) (\mathrm{e}^{-\beta(T_k^{\star} - T_k)} - \mathrm{e}^{-\beta(\tau - T_k)})$$

This result allows to establish the maximum likelihood estimation procedure that accounts for both self-exciting and self-inhibiting effects, while keeping the same computational complexity as the method implemented in previous works. In particular, for the exponential kernel function, the computational complexity of ℓ_T is O(N(T)) thanks to the Markovian property and, as shown in our numerical study, the derived estimations are better than by using an approximation framework (as presented in Lemonnier et al. (2014)) especially for high levels of inhibition effects.

A second consequence of Proposition 1.2.1 is that it gives access to a goodness-of-fit measure for our estimations by leveraging the Time Change theorem of point processes (Daley et al. 2003, Theorem 7.4.IV). In particular, this allows us to introduce a hypothesis testing procedure that evaluates the quality of our estimations on an independent set of observations.

1.2.3 Inference of multivariate exponential Hawkes processes with inhibition and application to neuronal activity

Chapter 4 builds upon the bases established in Chapter 3 to derive an estimation procedure for multivariate Hawkes processes $N = (N_1, \ldots, N_d)$. In this setting the log-likelihood of N reads:

$$\ell_T(\theta) = \sum_{i=1}^d \left(-\int_0^T \lambda_\theta^i(u) \,\mathrm{d}u + \sum_{k=1}^{N_i(T)} \log(\lambda_\theta^i(T_k^{i-1})) \right),$$

where λ_{θ}^{i} and $(T_{k}^{i})_{k}$ are respectively the candidate intensity and event times of N_{i} . Process N can be seen as a univariate point process with its event times $(T_{(k)})_{k}$ being the ordered union of $(T_{k}^{i})_{k}$ for all integers $i = \{1, \ldots, d\}$. To alleviate the notations, we will omit the subscript θ unless said otherwise. Similarly to the univariate case, we propose to study the underlying

intensity functions $\lambda^{i\star}$:

$$\lambda^{i\star}(t) = \mu_i + \sum_{j=1}^d \sum_{T_k^j \le t} h_{ij}(t - T_k^j)$$

but this time the monotonicity condition of h_{ij} is not enough to retrieve useful properties for this functions.

This is a consequence of the increased complexity of the dynamics of each subprocess that allows both exciting and inhibiting effects to exist simultaneously. In the multivariate setting, it is possible for a subprocess N_i to be saturated ($\lambda^i(t) = 0$) at an instant t and to receive an external excitation of another subprocess. So, in order to simplify our problem, we restrict our study to the exponential kernel case $h_{ij}(t) = \alpha_{ij} e^{-\beta_{ij}t}$ with the following mild assumption.

Assumption 1. For each $i \in \{1, \ldots, d\}$, there exists $\beta_i \in \mathbb{R}_{>0}$ such that $\beta_{ij} = \beta_i$ for all $j \in \{1, \ldots, d\}$.

This allows us to recover the strict monotony of $\lambda^{i\star}$ between any two event times $T_{(k)}$ and $T_{(k+1)}$ of the general process N. With this, we adapt the expression of the restart times $(T_{(k)}^{i\star})_{k\geq 1}$ and the compensator $\int_0^t \lambda^i(u) \, du$ for each process N_i and every event time $T_{(k)}$, allowing us to establish our first contribution in the form of Proposition 1.2.2.

Proposition 1.2.2. Let us suppose that Assumption 1 is granted. Then, for each $i \in \{1, ..., d\}$ and any $k \ge 1$, we have:

$$T_{(k)}^{i\star} = \min\left(t_k^{\star}, T_{(k+1)}\right).$$
(1.7)

where

$$t_k^{\star} = \left(T_{(k)} + \beta_i^{-1} \log \left(\frac{\mu_i - \lambda^{i \star}(T_{(k)})}{\mu_i} \right) \mathbb{1}_{\{\lambda^{i \star}(T_{(k)}) < 0\}} \right) \,.$$

And, for each $i \in \{1, ..., d\}$, the compensator of the process N_i reads, for any $t \ge 0$:

$$\int_{0}^{t} \lambda^{i}(u) \,\mathrm{d}u = \begin{cases} \mu_{i}t & \text{if } t < T_{(1)} \\ \mu_{i}T_{(1)} + \sum_{k=1}^{N(t)} J_{k} & \text{if } t \ge T_{(1)} , \end{cases}$$
(1.8)

where for all integer $k \in \{1, \ldots, N(t)\}$:

$$J_{k} = \mu_{i} \left[\min(t, T_{(k+1)}) - T_{(k)}^{i\star} \right] + \beta_{i}^{-1} \left(\lambda^{i\star}(T_{(k)}) - \mu_{i} \right) \left[e^{-\beta_{i}(T_{(k)}^{i\star} - T_{(k)})} - e^{-\beta_{i}(\min(t, T_{(k+1)}) - T_{(k)})} \right].$$

With this result, we establish an exact procedure to compute the log-likelihood ℓ_T^i of each process N_i , giving us access to the complete log-likelihood ℓ_T and in turn, access to the maximum likelihood estimation method. Let us note that as in the univariate case, we also get access to a goodness-of-fit procedure via the Time Change theorem.

Another complication in the multivariate setting is that the identifiability of our statistical model for the exponential Hawkes process is not as straightforward as in the univariate case. To partially answer this question, we provide a sufficient condition in Theorem 1.2.1 to ensure the identifiability of the intensity functions.

Theorem 1.2.1 (Identifiability). For all $i \in \{1, ..., d\}$, let $\theta_i \in \Theta$ be a parameter for λ^i .

Let us assume that a.s. for every $(i, j) \in \{1, ..., d\}^2$, $i \neq j$, there exist an event time τ from process N_j , and an event time $\tau_+ > \tau$ from process N_i , such that:

1. $\lambda_{\theta_i}^i(\tau^-) > 0;$

2. there are only events of process N^j in the interval $[\tau, \tau_+)$.

Then,

 $\forall i \in \{1, \dots, d\}, \ \lambda^{i}_{\theta_{i}}(t) = \lambda^{i}_{\theta'_{i}}(t) \ a.e. \qquad \Longleftrightarrow \qquad \forall i \in \{1, \dots, d\}, \ \theta_{i} = \theta'_{i}.$

Overall, this condition allows to avoid pathological situations either when some process N_i is not observed (strong external inhibition effects), or when there is a cyclical observation of points (strong internal inhibition effects).

Finally, we present a solution for estimating the interaction network of a multivariate Hawkes process. As it is common in real-data contexts, it is often unrealistic to assume that all functions h_{ij} are non-null, meaning that all processes interact with each other. For this, we propose three different data-driven techniques based on a thresholding approach inspired by principal component analysis, and based on the construction of confidence intervals (Student and empirical). Our results on synthetic data show a significant improvement with respect to other state-of-the-art methods (Lemonnier et al. 2014; Bacry et al. 2020). In particular, when the interaction matrix $(\alpha_{ij})_{i,j=1:d}$ contains null entries, we implement our post-hoc procedures to obtain an estimation of the connectivity network before re-estimating all non-null entries on a reduced model. This provides the best estimations, in the sense of the ℓ_2 relative error, among the considered models, as confirmed by our results on real-data concerning the analysis of neuronal activity.

1.3 Something is amiss: spectral methods for imperfect data

1.3.1 Context and related works

In the second part of the thesis, we turn the spotlight to a common issue in statistics: establishing efficient estimation procedures when the observed data is noised in some sense.

In the context of point processes, a natural approach to define noised models is through the operations of thinning, superposition and jittering.

- 1. The superposition of two point processes X and Y, usually noted X + Y is the point process defined by the ordered union of event times $(T_k^X)_k$ and $(T_k^Y)_k$.
- 2. The **thinning** of a point process X corresponds to erasing some event times T_k^X through a random rule. When point are erased independently from each other with a common probability 1 p, we talk about a p-thinning of process X.
- 3. The **jittering** of a point process consists in the random displacement of points.

The main interest of studying such alterations is to account for various measurement errors in real-world data. The most commonly studied in the literature is jittering which tends to appear when spatial imprecisions add a layer of uncertainty. Literature tends to focus on Poisson processes (Antoniadis et al. 2006; Hohage et al. 2016; Bonnet et al. 2022b), with Hawkes processes being studied in Trouleau et al. (2019) and Deutsch et al. (2020).

Thinning is often used to represent missing data but has been scarcely studied in the literature. Mei et al. (2019) propose a method to complete a sequence of times that have missing points via a Long Short-Term Memory neural network model based on the Hawkes process.

The study of superposition allows to account for observations that are contaminated by points originated from an external process, but are indistinguishable from those of the original process. To our knowledge, the only work that studies this kind of noise is Lund et al. (2000), which in fact tackles a general point process that is potentially noised by the three aforementioned mechanisms. Their work establishes an approach by maximising the conditional log-likelihood dependent on the original process with an application to Hawkes processes. In practice and for numerical reasons, they propose a local maximiser of this quantity in a small set of parameters when having access to an estimation of the unnoised process.

Other imperfect data models have been presented in Linderman et al. (2014) where the authors favour a Bayesian approach through latent variables to study processes with missing marks, unobserved time intervals or unobserved subprocesses. The last work we mention here is Cheysson et al. (2022), where the exact locations of the points of a Hawkes process are not known. Instead, the only information available is the number of points that fall inside equally-sized intervals. They leverage then the spectral theory of point processes and time series in order to provide an estimation procedure that does not necessitate the precise location of event times.

Our contribution is to make use of the spectral theory of point processes, centered around the Bartlett spectrum (Bartlett 1964) and the Whittle estimator (Whittle 1952), in order to propose estimators for Hawkes processes altered by superposition (Chapter 5) and thinning (Chapter 6). Another contribution is to leverage the obtained inference method to incorporate a subsampling paradigm through thinning for small observation windows. The existing spectral theory for Hawkes processes is limited to exciting interactions, so we restrict our models to this framework.

1.3.2 Spectral analysis for the inference of noisy Hawkes processes

Chapter 5 is dedicated to the study of the superposition of a Hawkes process H with exciting kernel functions and an independent homogeneous Poisson process P with intensity $\lambda_0 > 0$. We consider their superposition, that we note N = H + P, and we assume that we have no knowledge on the origin of each event time T_k^N (either Hawkes or Poisson). The intensity function of this resulting process is intractable and any method based on it is therefore inaccessible.

As mentioned previously, we take inspiration from Cheysson et al. (2022) in order to leverage the spectral approach of point processes to circumvent this issue. The Bartlett spectrum Γ^N of N is a measure defined by the means of the Fourier transform of its reduced covariance measure. When this measure is absolutely continuous, its density is noted f^N and often called the spectral density of the process. We can establish an estimation procedure of a parametric model for f^N (with parameter $\theta \in \Theta$) through the periodogram function I^T . For an observation $(T_k^N)_{k\geq 1}$ in [0, T], the periodogram is defined, for any $\omega \in \mathbb{R}$, as:

$$I^{T}(\omega) = \frac{1}{T} \sum_{k=1}^{N(T)} \sum_{l=1}^{N(T)} e^{-2\pi i \omega (T_{k}^{N} - T_{l}^{N})} .$$
(1.9)

Asymptotically, this quantity follows an exponential distribution with mean $f^N(\omega)$ and so it is possible to define the maximum spectral log-likelihood estimator $\hat{\theta}$ as:

$$\hat{\theta} \in \arg\max_{\theta \in \Theta} - \frac{1}{T} \sum_{k=1}^{M} \left(\log\left(f_{\theta}^{N}(\omega_{k}) \right) + \frac{I^{T}(\omega_{k})}{f_{\theta}^{N}(\omega_{k})} \right),$$
(1.10)

where $\omega_k = k/T$ for $k \in \{1, ..., M\}$. What makes this quantity so useful in our context is that it can be exactly computed without any information on the source of event times $(T_k^N)_k$.

Our first result establishes the expression of the spectral density function for the superposition of two independent point processes as exhibited in Proposition 1.3.1.

Proposition 1.3.1. Let X and Y be two independent and stationary point processes, admitting respective spectral densities f^X and f^Y .

Then N = X + Y admits a spectral density function f^N and

$$f^N = f^X + f^Y \,. \tag{1.11}$$

As the spectral density of a Hawkes process is explicit (Daley et al. 2003, Example 8.2(e)), we obtain the general form of the spectral density of our noisy Hawkes process, which in turn makes it possible to establish an estimation procedure by maximising the spectral log-likelihood.

Our main contribution in the univariate setting is to establish conditions for the identifiability of f^N for different parametrisations of the kernel function h. The first result is related to the classical exponential kernel. We show that the model is identifiable if and only if one of the four parameters (three for the Hawkes process and one for the Poisson process) is fixed, even though the distribution of N is uniquely determined without this constraint. Furthermore, we prove that the noisy Hawkes process with a uniform kernel function defines an identifiable spectral model.

In the multivariate case, all tools from spectral theory exist and results are easily extended concerning the estimation procedure. Our contributions specifically concern the bivariate exponential Hawkes process noised by a bivariate Poisson process with common parameter λ_0 . We present sufficient conditions on the interaction matrix $(\alpha_{ij})_{1 \leq i,j \leq 2}$ that characterise either the identifiability or non-identifiability of the corresponding statistical model.

Our work is concluded by a numerical study in both the univariate and bivariate cases when identifiability conditions are verified. We show in both cases that our estimators perform particularly well, especially for processes with strong excitation effects and as the observation horizon T increases. Our final contribution is to propose an ad-hoc procedure in the bivariate case that allows to determine the support of the interaction matrix by considering the empirical 5% quantiles of a sample of estimations, which in turn enhances the performance of our methods.

1.3.3 A numerical exploration of thinned Hawkes processes through spectral theory

Finally, Chapter 6 turns to the analysis of thinning for point processes. Let H_p be a process issued from thinning a univariate Hawkes process with a probability 1 - p of erasing each point T_k , for $k \in \mathbb{Z}$. As previously mentioned, any method based on the conditional intensity function is unavailable, so we turn to spectral theory.

All the tools presented in the previous subsection can still be implemented here, in particular the expression of the periodogram (Equation (1.9)) remains unchanged. As a consequence, it can still be exactly computed even though we do not observe all points of the original process. In order to implement an inference procedure as defined by Equation (1.10), we need to obtain an expression of the *p*-thinned process H_p . This is presented in Proposition 1.3.2.

Proposition 1.3.2. Let H be a stationary point process admitting a spectral density function f^H and let $m_1 = \mathbb{E}[\lambda(0)]$ be its average intensity. For any $p \in (0,1]$, let H_p be a p-thinning of H.

Then, H_p admits a spectral density function, denoted f^{H_p} , such that for any $\omega \in \mathbb{R}$:

$$f^{H_p}(\omega) = p^2 f^H(\omega) + p(1-p)m_1.$$
(1.12)

This result is obtained by leveraging the spectral theory of marked point processes as presented in Brémaud et al. (2005).

We focus our study on the exponential kernel function and we prove that the derived statistical model is identifiable if and only if one parameter is fixed, similar to the aforementioned superposition scenario. We present numerical results under this condition by fixing the value of the thinning parameter p. In particular, we show that our estimation method is robust with respect to the level of thinning, with remarkable results when enough event times are available for inference.

Our last contribution in this work is to make use of the thinning operation as a subsampling procedure when a single observation of H is available in a small window of time. The use of subsampling for point processes has been done previously in the literature (Møller et al. 2003; Cronie et al. 2024) in order to improve estimation procedures. In our case, we focus on enhancing the quality of our spectral estimations by combining an ℓ_2 penalisation with an averaged estimator obtained by thinning the unique observation of H. We show that our proposed procedure provides in general the best estimations, in the sense of the relative ℓ_2 error, when compared to another subsampling procedure and displays a substantial improvement by reducing the high bias observed for the non-penalised estimator.

1.4 Outline of the manuscript

Chapter 2 proposes a succinct presentation of point processes theory, from a random measure approach, and the formalism of Hawkes processes.

The first part of this thesis concerns the study of the inhibition.

• Chapter 3 presents a parametric estimation method for univariate Hawkes processes that accounts for both self-exciting or self-inhibiting interactions. This is a joint work with Anna Bonnet and Maxime Sangnier, published in *Statistics and Probability Letters* (Bonnet et al. 2021).

Code is freely available at https://github.com/migmtz/hawkes-inhibition-expon

• Chapter 4 presents a parametric estimation method for multivariate exponential Hawkes processes with exciting and inhibiting interactions along with a model selection procedure. This is a joint work with Anna Bonnet and Maxime Sangnier, published in *Statistics and Computing* (Bonnet et al. 2023).

Code is freely available at https://github.com/migmtz/multivariate-hawkes-inhibition

The second part consists in the study of imperfect observations of a Hawkes process realisation, similar to missing data formulations.

• Chapter 5 presents a parametric estimation method for exciting Hawkes processes whose event times are noised by those of a homogeneous point process, leveraging point process spectral theory. This is a joint work with Anna Bonnet, Felix Cheysson and Maxime Sangnier and has been submitted for publication (Bonnet et al. 2024).

Code is available at https://github.com/migmtz/noisy-hawkes-process

• Chapter 6 presents the analysis of an inference method for thinned univariate Hawkes processes through spectral theory. Additionally, it presents an improvement of ℓ_2 penalisation through thinning subsampling for the spectral estimator.

This chapter is an ongoing joint work with Felix Cheysson.

An introduction to Hawkes processes study

Outline of the current chapter

2.1	Introduction	16
2.2	Random measures and point processes	17
	2.2.1 Random measures	17
	2.2.2 Point processes	18
	2.2.3 Integrals and moment measures	19
	2.2.4 The conditional intensity function	20
2.3	Hawkes process	22
	2.3.1 The self-exciting Hawkes process	22
	2.3.2 The multivariate Hawkes process	23
	2.3.3 Non-linear Hawkes processes	23
2.4	Inference for Hawkes processes	24
	2.4.1 MLE for Hawkes process with exponential kernel	24
	2.4.2 Least-squares minimisation	26
	2.4.3 A Bayesian estimation approach	26
2.5	Simulation	27
	2.5.1 Ogata's thinning simulation	27
	2.5.2 Simulation through branching theory	28

2.1 Introduction

Point processes are a fundamental concept in stochastic modelling, providing a framework for analysing events that occur randomly over time. Traditionally, a temporal point process is presented through the concept of an ordered collection of random variables in the real line, each one representing the occurrence of an event. A standard approach to study point processes is by defining random measures, which are random variables with realisations on the space of boundedly finite measures. In particular, this framework gives access to many results from set theory and measure theory as largely shown in the literature (Cox et al. 1980; Baddeley 2006; Daley et al. 2008; Baccelli et al. 2020).
In this chapter, we present a succinct introduction to these concepts in order to present the submodel of past-dependent point processes: the Hawkes model (Hawkes 1971). Our goal here is twofold: the first one is to introduce numerous notations and definitions that are used throughout this manuscript, serving as a general reference for further chapters. The second is to present a short review of the original self-exciting Hawkes processes and some numerical procedures from the literature.

We introduce the concept of temporal point processes through random measure theory in Section 2.2 with a particular interest on the concept of conditional intensity functions. With this background, we formally introduce the univariate self-exciting Hawkes processes in Section 2.3 with the two classical extensions to multivariate and non-linear settings, the latter being often used to model inhibition effects. We follow up with a general presentation of some classic inference methods in Section 2.4 and we conclude with Section 2.5 by presenting two methods of simulation that are commonly used in the context of Hawkes processes.

2.2 Random measures and point processes

2.2.1 Random measures

Let $\mathcal{X} \subseteq \mathbb{R}$ denote either the real half-line $\mathbb{R}_{\geq 0}$ or the real line \mathbb{R} and $\mathcal{B}_{\mathcal{X}}$ be the Borel σ -algebra of \mathcal{X} .

Definition 2.2.1. A measure ν on \mathcal{X} is said to be *boundedly finite* if for any bounded $B \in \mathcal{B}_{\mathcal{X}}$, $\nu(B) < +\infty$ and we define $\mathcal{M}_{\mathcal{X}}$ as the set of boundedly finite measures on \mathcal{X} .

For $\mathcal{M}_{\mathcal{X}}$, we may obtain Borel σ -algebras generated by the mappings $\nu \mapsto \nu(B)$, for all $B \in \mathcal{B}_{\mathcal{X}}$, that we note $\mathcal{B}(\mathcal{M}_{\mathcal{X}})$.

Let $(\Omega, \mathcal{F}, \mathbb{P})$ be a probability space.

Definition 2.2.2 (Random measure). A random measure ξ on \mathcal{X} is a measurable mapping from Ω into $\mathcal{M}_{\mathcal{X}}$. The distribution of a random measure ξ is the probability measure induced by the probability measure \mathbb{P} .

An equivalent definition of a random measure is obtained through the following result.

Proposition 2.2.1. (Baccelli et al. 2020, Proposition 1.1.7)

For a mapping $\xi : (\Omega, \mathcal{F}) \to (\mathcal{M}_{\mathcal{X}}, \mathcal{B}(\mathcal{M}_{\mathcal{X}}))$, we define for any $B \in \mathcal{B}_{\mathcal{X}}$ the mapping $\xi(B) : (\Omega, \mathcal{F}) \to (\mathcal{X}, \mathcal{B}_{\mathcal{X}})$ as:

$$(\xi(B))(\omega) = (\xi(\omega))(B)$$

Then, ξ is a random measure if and only if, for all $B \in \mathcal{B}_{\mathcal{X}}, \xi(B)$ is measurable.

With this result, it means that we can work with a random measure ξ in two ways:

- For any $\omega \in \Omega$, a realisation $\xi(\omega)$ is a measure on \mathcal{X} .
- For any $B \in \mathcal{B}_{\mathcal{X}}, \xi(B)$ is a random variable on \mathcal{X} . Furthermore, $\{\xi(B)\}_{B \in \mathcal{B}_{\mathcal{X}}}$ is a stochastic process.

In practice, as with other random variables we tend to omit the term ω when working with random measures and using ξ either as a measure or random variable is to be understood by the context. Another direct result from Proposition 2.2.1 is that, for any finite sequence $(B_k)_{k=1:K}$ of Borel sets, we can define the random vector:

$$(\xi(B_1),\ldots,\xi(B_K)).$$

It turns out that a characterisation of any random measure is given by the distribution of all of these vectors.

Proposition 2.2.2. (Daley et al. 2008, Corollary 9.2.IV)

Let ξ be a random measure on \mathcal{X} . The finite-dimensional distributions of ξ are the probability distribution of the vectors $(\xi(B_1), \ldots, \xi(B_K))$ for all integers $K \ge 1$ and all sequences of Borel sets $(B_k)_{k=1:K}$.

The distribution of ξ is fully given by its finite-dimensional distributions.

With this result, a way of defining any random measure is by defining its finite-dimensional distributions. However, not any set of finite-dimensional distributions corresponds to those of a random measure, but a set of necessary and sufficient conditions is proposed in Daley et al. (2008, Conditions 9.2.V-VI), namely Kolmogorov extension theorem conditions and additivity and continuity conditions. Finally, let us introduce the notion of stationarity. For any $B \in \mathcal{B}_{\mathcal{X}}$, for any $x \in \mathcal{X}$, we define B + x as the set:

$$B + x = \{b + x \mid b \in B\}.$$

Definition 2.2.3. A random measure ξ is said to be *stationary* if for any $B \in \mathcal{B}_{\mathcal{X}}$ and for any $x \in \mathcal{X}$, N(B) and N(B+x) follow the same distribution.

2.2.2 Point processes

Let us now turn to the study of point processes. For this, we need the definition of a counting measure.

Definition 2.2.4. A measure ν on \mathcal{X} is a counting measure if for any $B \in \mathcal{B}_{\mathcal{X}}, \nu(B) \in \mathbb{N} \cup \{+\infty\}$. We note $\mathcal{N}_{\mathcal{X}}$ the set of all boundedly finite counting measures on \mathcal{X} .

As done previously, we can equip the space $\mathcal{N}_{\mathcal{X}}$ with a Borel σ -algebra $\mathcal{B}(\mathcal{N}_{\mathcal{X}})$ generated by the mappings $\nu \mapsto \nu(B)$, for all $B \in \mathcal{B}_{\mathcal{X}}$.

Definition 2.2.5 (Point processes). A *point process* N on \mathcal{X} is a measurable mapping from a $(\Omega, \mathcal{F}, \mathbb{P})$ into $(\mathcal{N}_{\mathcal{X}}, \mathcal{B}(\mathcal{N}_{\mathcal{X}}))$.

Intuitively, a point process N is a random measure (Baccelli et al. 2020, Corollary 1.6.4.) such that, for any $\omega \in \Omega$, the realisation $N(\omega)$ is a counting measure and so in particular all previous results from random measures hold for a point process. For instance, for any $B \in \mathcal{B}_{\mathcal{X}}$, N(B) will denote the measure of B by N, which allows us to introduce the definition of a simple point process.

Definition 2.2.6. A point process N is simple if, for any $x \in \mathcal{X}$, $N(\{x\}) \in \{0, 1\}$, a.s.

It is very common in the literature to work uniquely with simple point processes, but before further discussing this, let us exhibit the following result from the theory of point processes.

Theorem 2.2.1. (Last et al. 2017, Corollary 6.5)

Let N be a point process on \mathcal{X} and let δ_x be the Dirac measure on x, for any $x \in \mathcal{X}$. There exists a random variable $\kappa \in \mathbb{N} \cup \{+\infty\}$ and a sequence of random variables (T_1, T_2, \ldots) such that:

$$N = \sum_{k=1}^{n} \delta_{T_k} \,, \quad a.s.$$

We refer to the variables T_k as the event times or points of process N.

This theorem reunites the definition of a point process as a random measure with other formalisms of point processes as random points observed in a space \mathcal{X} . A simple point process is then a point process without coinciding points, in other words, for any $i \neq j$, $T_i \neq T_j$ almost surely. From now on, we will assume that all point processes are simple.

Let us now turn to some particularities of working in a temporal setting. As \mathcal{X} is a wellordered set, it is practical to order the event times of N. The indexation of the sequence of event times slightly differs whether we work on $\mathbb{R}_{\geq 0}$ or \mathbb{R} . Unlike Theorem 2.2.1 may suggest, it is a common adopted convention, when working on the entire real line, that the event times are indexed by \mathbb{Z} such that

$$\ldots < T_{-2} < T_{-1} < T_0 \le 0 < T_1 < T_2 < \ldots$$

This way, positive indices will always denote points *after* the origin of time t = 0. Another usual convention is to denote N(t) = N([0, t]) the number of points in the interval [0, t] especially used in inference contexts. This allows also to see $(N(t))_{t \in \mathcal{X}}$ as a right-continuous stochastic process.

2.2.3 Integrals and moment measures

We can define the stochastic integral of a measurable function $f : \mathcal{X} \to \mathbb{R}_{\geq 0}$ against a point process N with event times $(T_k)_{k \in \mathbb{Z}}$ as:

$$\int_{\mathcal{X}} f(t) N(\mathrm{d}t) = \sum_{k \in \mathbb{Z}} f(T_k) \,,$$

which is similar to the Lebesgue-Stieljes integral. We can see that the integral defined this way is a random variable and by taking $f = \mathbb{1}_B$ for any $B \in \mathcal{B}_X$, we can retrieve the measure of B:

$$\int_{\mathcal{X}} \mathbb{1}_B(t) N(\mathrm{d}t) = \sum_{k \in \mathbb{Z}} \mathbb{1}_{T_k \in B} = N(B) \,.$$

Let us now define the moment measures of N.

Definition 2.2.7. Let N be a point process. Whenever it exists, we note M_k the k-th moment measure of N defined, for any $(B_1, \ldots, B_k) \in \mathcal{B}^k_{\mathcal{X}}$, as:

$$M_k(B_1,\ldots,B_k) = \mathbb{E}[N(B_1)\ldots N(B_k)].$$

In particular, the first moment measure M_1 is also known as the *intensity measure* of N, also denoted Λ .

With the intensity measure, we are able to introduce the most classical point process, the Poisson process.

Definition 2.2.8. Let Λ be a measure on \mathcal{X} absolutely continuous with respect to the Lebesgue measure ℓ and let $\lambda : \mathcal{X} \to \mathbb{R}_{>0}$ be its Radon-Nikodym derivative.

A point process N is a *Poisson process* with intensity function λ if it verifies the following conditions:

- 1. For any $B \in \mathcal{B}_{\mathcal{X}}$, N(B) follows a Poisson distribution with parameter $\Lambda(B) = \int_{B} \lambda(t) dt$.
- 2. For any integer K and any set of disjoint Borel sets $(B_k)_{k=1:K}$, the variables $(N(B_k))_{k=1:K}$ are independent.

If λ is a constant function, then N is called a *homogeneous* Poisson process.

The Poisson process is sometimes referred to as a process with independent increments because of the second property. In particular, when the intensity function λ is constant, we can further characterise the process by its inter-arrival times.

Theorem 2.2.2. (Last et al. 2017, Theorem 7.2)

N is a homogeneous Poisson process with intensity $\lambda > 0$ if and only if the inter-arrival times $(T_{k+1} - T_k)_{k \in \mathbb{Z}}$ are i.i.d, all following an exponential distribution with parameter λ .

2.2.4 The conditional intensity function

In the previous subsection, we defined the Poisson process through its intensity function, which allows to account for different time-dependent dynamics. The concept that will allow us to properly introduce the Hawkes process is the conditional intensity function. For this, let us note \mathcal{H}_t the *history* of a point process N up to the instant $t \in \mathcal{X}$. \mathcal{H}_t is the σ -algebra generated by the event times T_k :

$$\mathcal{H}_t = \sigma(\{T_k \mid k \in \mathbb{Z}, T_k \le t\}).$$

Definition 2.2.9. Let N be a simple point process on \mathcal{X} , and $(\mathcal{H}_t)_t$ the histories of N. The conditional intensity function λ of N is defined, for any $t \in \mathcal{X}$, as:

$$\lambda(t \mid \mathcal{H}_t) = \lim_{h \to 0} \frac{\mathbb{E}[N([t, t+h]) \mid \mathcal{H}_t]}{h} \,. \tag{2.1}$$

Let us remark that by Proposition 7.2.IV in Daley et al. (2003), a point process is entirely determined by its conditional intensity function. In general, the conditional intensity function λ is a random variable for each $t \in \mathcal{X}$ as it is expressed through a conditional expectation. However, if λ is a deterministic function, then it coincides with the intensity function of a Poisson process as given in Definition 2.2.8, which justifies keeping the same notation λ .

It is a common practice to omit the term \mathcal{H}_t in Equation (2.1) as a conditional intensity is usually understood as having access to the entire past history (unless it is to avoid ambiguity). Intuitively λ quantifies the instantaneous probability of observing a point on an interval of infinitesimal size h. As N is a simple point process, the expectation in Equation (2.2.8) corresponds to $\mathbb{P}(N([t, t+h])) = 1$ for h small enough. Another common interpretation (see Hawkes (1971), for an example) of the conditional intensity function for simple point processes is through the following properties:

$$\begin{cases} \mathbb{P}(N([t,t+h]) = 1) = \lambda(t)h + o(h) \\ \mathbb{P}(N([t,t+h]) = 0) = 1 - \lambda(t)h + o(h) \\ \mathbb{P}(N([t,t+h]) \ge 2) = o(h) \,. \end{cases}$$

A very formal presentation of the conditional intensity function can be found in Daley et al. (2003, Chapter 7) through the concept of Janossy densities. In a nutshell, given a number K of observed points in a set B, the Janossy densities describe the probability distributions of points $(T_k)_{k=1:K}$ inside B.

Definition 2.2.10. Let N be a simple point process and λ its conditional intensity function. The *compensator* Λ of N is the random measure defined, for any $B \in \mathcal{B}_{\mathcal{X}}$ as:

$$\Lambda(B) = \int_B \lambda(t) \, \mathrm{d}t \, .$$

In particular, the expectation of Λ is the first-order moment measure of N. If N is stationary, then this measure is a constant multiple of the Lebesgue measure ℓ .

Again, the concept of compensator coincides with the intensity measure in Definition 2.2.8 whenever λ is deterministic. Similarly to the notation of point processes, we note $\Lambda(t) = \Lambda([0, t])$ for any $t \in \mathcal{X}$. The compensator is a highly studied quantity for point processes defined through conditional intensity functions. An important result concerning Λ is the time change theorem:

Theorem 2.2.3. (Daley et al. 2003, Theorem 7.4.IV)

Let $(X_k)_{k\in\mathbb{Z}}$ be an a.s. increasing sequence of random variables in \mathcal{X} and $\mathcal{F}_t = \sigma(\{X_k \mid k \in \mathbb{Z}, X_k \leq t\})$, for all $t \in \mathbb{R}$. Let $\bar{\lambda}(\cdot \mid \mathcal{F})$ be a non-negative function such that, for all $t \in \mathbb{R}$, $\bar{\lambda}(t \mid \mathcal{F}_t)$ is \mathcal{F}_t -measurable and let $\bar{\Lambda}$ be defined, for any $t \in \mathbb{R}$, as:

$$\bar{\Lambda}(t) = \int_{-\infty}^{t} \bar{\lambda}(u \mid \mathcal{F}_u) \,\mathrm{d}u \,.$$

Then, the process $\sum_{k \in \mathbb{Z}} \delta_{X_k}$ is a point process with conditional intensity function $\bar{\lambda}$ if and only if $\sum_{k \in \mathbb{Z}} \delta_{\bar{\Lambda}(X_k)}$ is a homogeneous Poisson process with unit intensity.

It is then possible to rescale the event times of any observed point process N in order to obtain a realisation of a homogeneous Poisson process. This is a very useful result in order to establish goodness-of-fit procedures for point processes when an estimation $\hat{\Lambda}$ of Λ is available. For example, by transforming the observed event times $(T_k)_k$ with $\hat{\Lambda}$, we can then test whether the inter-arrival times $(\hat{\Lambda}(T_{k+1}) - \hat{\Lambda}(T_k))_k$ are exponentially distributed (see Theorem 2.2.2).

Lastly, we conclude this section by introducing the expression of the likelihood.

Proposition 2.2.3. (Daley et al. 2003, Theorem 7.2.III)

Let N be a point process with conditional intensity function λ . Let $(T_k)_{k=1:N(T)}$ be the realisation of N in the interval [0,T].

Then the likelihood L_T of N reads:

$$L_T = \left(\prod_{k=1}^{N(T)} \lambda(T_k^-)\right) e^{-\Lambda(T)}, \qquad (2.2)$$

where $\lambda(T_k^-) = \lim_{t \to T_k^-} (\lambda(t))$. The log-likelihood of N is:

$$\ell_T = \sum_{k=1}^{N(T)} \log(\lambda(T_k^-)) - \Lambda(T) \,. \tag{2.3}$$

The likelihood can be introduced by means of the Janossy densities. Let us note that, in the case of point processes on the entire real line \mathbb{R} , it is necessary to know all the points $T_k \leq T$ in order to compute the likelihood, which is usually unavailable. In practice, it is often assumed that no points occured in $(-\infty, 0)$ and so an adapted version of λ is used in Equation 2.2 (see Ogata (1978)).

2.3 Hawkes process

2.3.1 The self-exciting Hawkes process

The Hawkes process, as introduced in Hawkes (1971), is defined as a point process in \mathbb{R} as follows:

Definition 2.3.1. Let $\mu > 0$ and $h : \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$ be a measurable function such that

$$||h||_1 = \int_0^{+\infty} |h(t)| \, \mathrm{d}t < 1.$$

A Hawkes process N is a point process on \mathbb{R} defined by the conditional intensity function:

$$\forall t \in \mathbb{R}, \qquad \lambda(t) = \mu + \int_{-\infty}^{t} h(t-s) N(\mathrm{d}s) \,. \tag{2.4}$$

 μ is known as the baseline intensity and h as the interaction or kernel function.

This process was initially referred to as a self-exciting point process due to the integral in Equation (2.4) that represents the positive contribution of all past points up to time t to the intensity function λ . Let us recall that:

$$\int_{-\infty}^{t} h(t-s) N(\mathrm{d}s) = \sum_{T_k \leq t} h(t-T_k) \,,$$

and so each event time T_k increases the baseline intensity by an additive factor of $h(t - T_k)$, hence the self-excitation effect.

The condition

$$\int_0^{+\infty} h(t) \,\mathrm{d}t < 1\,,$$

for the interaction function ensures that N is boundedly finite a.s. and so that N is a point process. Another result of this condition is that $\lim_{t\to+\infty} h(t) = 0$, meaning that the effect of any point T_k dissipates the further they are in the past.

Another implication of this condition is that a Hawkes process defined in \mathbb{R} is a stationary point process (Hawkes et al. 1974, Lemma 1) which allows to establish the following result.

Proposition 2.3.1. (Hawkes 1971, Equation (9)) Let N be a stationary Hawkes process. Then the average intensity of N reads:

$$\forall t \in \mathbb{R}, \qquad \mathbb{E}[\lambda(t)] = \frac{\mu}{1 - \|h\|_1}$$

The stationarity of process N is verified for processes only in \mathbb{R} and so this result does not hold for a process in $\mathbb{R}_{\geq 0}$. On the one hand, it is often more practical for theoretical reasons to work in \mathbb{R} , as stationarity tends to facilitate establishing certain results (see Hawkes (1971) for example in the context of spectral theory). On the other hand, working with a process in $\mathbb{R}_{\geq 0}$ is often preferred for application purposes, as it is unrealistic to suppose that the practitioner has access to an infinite number of points in the past before t = 0. To define such a process, it suffices to replace the lower bound in the integral of Equation (2.4) by 0.

2.3.2 The multivariate Hawkes process

The multivariate version of a self-exciting Hawkes process was introduced in Hawkes (1971) as follows.

Definition 2.3.2. A multivariate Hawkes process $N = (N_1, \ldots, N_d)$ of dimension d is defined by d point processes $(N_i)_{i=1:d}$. For each integer i, we note $(T_k^i)_k$ the event times of process N_i and its conditional intensity function λ^i reads:

$$\lambda^{i}(t) = \mu_{i} + \sum_{j=1}^{d} \int_{-\infty}^{t} h_{ij}(t-s) N_{j}(\mathrm{d}s) = \mu_{i} + \sum_{j=1}^{d} \sum_{T_{k}^{j} \leq t} h_{ij}(t-T_{k}^{j}).$$

For any integers $i, j, \mu_i > 0$ is the baseline intensity of N_i and $h_{ij} \colon \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$ is a measurable function.

For a multivariate Hawkes process, each intensity function λ^i is influenced by every point T_k^j of process N_j , for all integers j. Function h_{ij} represents the influence of a point from N_j to process N_i . We define the matrix $S = (\|h_{ij}\|_{1})_{1 \ge i,j \ge d}$ and a condition for all N^i to be proper point processes is to control the spectral radius $\rho(S) < 1$ (Bacry et al. 2015). This condition ensures again the stationarity of all processes when defined in \mathbb{R} .

When d = 1, we retrieve the original expression of a Hawkes process from Definition 2.3.1 and so we talk about a univariate Hawkes process. Defining a multivariate Hawkes process as a tuple is the more common formulation for any kind of point process, but it is sometimes practical to consider process N as the superposition of all subprocesses N_i . N is then characterised by the ordered union of all event times $(T_k^i)_{k\in\mathbb{Z}}$ for all integers i, forming a single point process in \mathbb{R} with event times $(T_{(k)})$ and conditional intensity function:

$$\lambda(t) = \sum_{i=1}^{d} \lambda^{i}(t)$$

This perspective allows to extend all results from general point process theory to multivariate processes, including inference methods.

2.3.3 Non-linear Hawkes processes

A more general formulation of a past-dependent point process inspired by the Hawkes process was proposed in Brémaud et al. 1996 and known as the non-linear Hawkes process.

Definition 2.3.3. The univariate *non-linear Hawkes process* N on \mathbb{R} is defined by the conditional intensity function:

$$\lambda(t) = \Phi\left(\mu + \int_{-\infty}^{t} h(t-s) N(ds)\right), \qquad (2.5)$$

where $\mu > 0$ and $\Phi : \mathbb{R} \to \mathbb{R}_{\geq 0}$ and $h : \mathbb{R}_{\geq 0} \to \mathbb{R}$ are two measurable functions. Φ is known as the activation function and h as the *interaction* or *kernel function*.

The usual formulation of the self-exciting Hawkes process in Definition 2.3.1 can be retrieved by taking the identity function $\Phi: x \mapsto x$ and by adding a positivity constraint on h. Because of this, it is often referred to as the *linear* Hawkes process.

By allowing h to take negative values in Equation (2.5), the non-linear Hawkes process can model an *inhibiting* effect between points where each point decreases the chances of other points occuring, in opposition to the excitation effect. To guarantee the existence of such a process, either Φ needs to be upper-bounded or Φ to be *L*-lipschitz, for L > 0, such that:

$$L \|h^+\|_1 < 1$$
,

where $(\cdot)^+$: $x \mapsto \max(0, x)$ denotes the positive part function.

A multivariate version of this process can be defined as for the classical Hawkes process in Definition 2.3.2. In this setting, existence can be retrieved either by upper-boundness of Φ or by imposing that the spectral radius of matrix $S^+ = (L \| h_{ij}^+ \|_1)_{1 \ge i,j \ge d}$ is strictly smaller than 1 (Sulem et al. 2024, Lemma 2.1). For instance, if Φ is the positive part function, this condition can be retrieved by upper-bounding the intensity function by the intensity of the linear process N^+ with interaction functions $(h_{ij}^+)_{i,j}$, for all integers i, j (Deutsch et al. 2024, Theorem 2).

This function, often called ReLU activation, is the most common choice in the literature for the non-linear function (Lemonnier et al. 2014; Lu et al. 2018; Costa et al. 2020), with other examples include clipped exponential (Chornoboy et al. 1988; Carstensen et al. 2010), sigmoid (Menon et al. 2018) and softplus functions (Mei et al. 2017). This model of inhibiting interactions for the Hawkes process is often referred to as *additive* inhibition and other works in the literature have proposed alternative versions such as the self-limiting Hawkes process with a multiplicative exponential term dependent on the number of points in a moving interval (Olinde et al. 2020), and the mean-field Hawkes process with multiplicative inhibition (Duval et al. 2022).

2.4 Inference for Hawkes processes

In this section we will review some classical methods for the estimation of the baseline intensity μ and interaction function h of the linear Hawkes process N. The literature on inference procedures in the non-linear setting is scarcer and some will be mentioned in Section 2.4.3.

2.4.1 MLE for Hawkes process with exponential kernel

A common method is maximisation of the log-likelihood:

$$\ell_T = \sum_{k=1}^{N(T)} \log(\lambda(T_k^-)) - \Lambda(T) \,,$$

as shown in Ozaki (1979). A multivariate version of it is presented in Embrechts et al. (2011) and Guo et al. (2018) with expected mathematical properties including consistency and asymptotic normality (Simon et al. 2017), mainly developed in parametric settings.

We present here the implementation of the maximum likelihood estimation method for univariate self-exciting Hawkes processes. Let N be a Hawkes process in $\mathbb{R}_{\geq 0}$ and we assume that the function h is parametrised by an exponential distribution such that:

$$h(t) = \alpha e^{-\beta t}$$
, for $t \ge 0$

with $\alpha > 0$ and $\beta > 0$. The existence condition in Definition 2.3.1 reads:

$$\|h\|_1 = \frac{\alpha}{\beta} < 1\,,$$

and so we assume that $\alpha < \beta$. By defining the following parametric model:

$$\mathcal{P} = \left\{ \lambda_{\theta} \mid \theta = (\mu, \alpha, \beta) \in \mathbb{R}^3_{>0}, \alpha < \beta \right\} ,$$

the log-likelihood (2.3) becomes a function of parameter θ :

$$\ell_T(\theta) = \sum_{k=1}^{N(T)} \log(\lambda_\theta(T_k^-)) - \Lambda_\theta(T) + \frac{1}{2} \log(\lambda_\theta(T_k^-)) - \frac{1$$

with Λ_{θ} the compensator of λ_{θ} .

The choice of the exponential kernel function for N implies that the intensity λ is Markovian in the sense that in each interval $[T_k, T_{k+1})$ the expression is solely dependent on the value of $\lambda(T_k)$. For any integer $k \geq 1$ and any $t \in [T_k, T_{k+1})$, the intensity reads:

$$\lambda(t) = \mu + \int_{0}^{+\infty} \alpha e^{-\beta(t-s)} N(ds) = \mu + \sum_{j=1}^{k} \alpha e^{-\beta(t-T_{j})}$$
$$= \mu + e^{-\beta(t-T_{k})} \sum_{j=1}^{k} \alpha e^{-\beta(T_{k}-T_{j})}$$
$$= \mu + e^{-\beta(t-T_{k})} \left(\lambda(T_{k}) - \mu\right).$$
(2.6)

We can then easily integrate λ in $[T_k, T_{k+1})$:

$$\int_{T_k}^{T_{k+1}} \lambda(t) \, \mathrm{d}t = \mu(T_{k+1} - T_k) + (\lambda(T_k) - \mu) \int_{T_k}^{T_{k+1}} \mathrm{e}^{-\beta(t - T_k)} \, \mathrm{d}t$$
$$= \mu(T_{k+1} - T_k) + (\lambda(T_k) - \mu)\beta^{-1}(1 - \mathrm{e}^{-\beta(T_{k+1} - T_k)})$$

Then, for any interval [0, T], the compensator Λ can be computed piecewise:

$$\begin{split} \Lambda(T) &= \int_0^T \lambda(t) \, \mathrm{d}t \\ &= \int_0^{T_1} \mu \, \mathrm{d}t + \sum_{k=1}^{N(T)-1} \int_{T_k}^{T_{k+1}} \lambda(t) \, \mathrm{d}t + \int_{T_N(T)}^T \lambda(t) \, \mathrm{d}t \\ &= \mu T + \beta^{-1} \left(\sum_{k=1}^{N(T)-1} (\lambda(T_k) - \mu)(1 - \mathrm{e}^{-\beta(T_{k+1} - T_k)}) + (\lambda(T_N(T)) - \mu)(1 - \mathrm{e}^{-\beta(T - T_k)}) \right) \,. \end{split}$$

The consequence of this expression is that the terms $\lambda(T_k)$ (and $\lambda(T_k^-)$) can be computed recursively by Equation 2.6 and so Λ has a computational complexity of O(N(T)). The log-likelihood also has a complexity of O(N(T)) and can be efficiently computed as described in Algorithm 1:

In practice, we obtain an estimation $\hat{\theta}$ of a parameter $\theta = (\mu, \alpha, \beta)$ by maximising the loglikelihood.

Other methods implemented in parametric settings are the method of moments (Da Fonseca et al. 2013), by leveraging the spectral theory of point processes to maximise a spectral version of the log-likelihood (Adamopoulos 1976) or through an Expectation-Maximisation procedure to estimate the branching structure of the Hawkes process (Veen et al. 2008).

Algorithm 1: Computation of ℓ_T

Input Parameters μ , α , β , time horizon T and sequence of event times $(T_k)_{k=1:N(T)}$; Initialization Initialize $\ell_T = \log(\mu) - \mu T$, $\lambda(T_1) = \mu + \alpha$; for $k \ge 1$ do Compute $\lambda(T_{k+1}^-) = \mu + (\lambda(T_k) - \mu)e^{-\beta(T_{k+1} - T_k)}$; Update $\ell_T = \ell_T + \log(\lambda(T_{k+1}^-)) - \beta^{-1}(\lambda(T_k) - \mu)(1 - e^{-\beta(T_{k+1} - T_k)})$; Compute $\lambda(T_{k+1}) = \lambda(T_{k+1}^-) + \alpha$; end Update $\ell_T = \ell_T - \beta^{-1}(\lambda(T_k) - \mu)(1 - e^{-\beta(T - T_k)})$; return Log-likelihood ℓ_T

2.4.2 Least-squares minimisation

Inference procedures in non-parametric settings tend to be based on the minimisation of a leastsquares contrast. Let us assume that we want to estimate an intensity function λ in an interval [l, L] through a function $\hat{\lambda}$. We define the squared error of $\hat{\lambda}$ as:

$$\|\hat{\lambda} - \lambda\|_2^2 = \int_l^L (\hat{\lambda}(t) - \lambda(t))^2 \,\mathrm{d}t \tag{2.7}$$

$$= \int_{l}^{L} \hat{\lambda}^{2}(t) \,\mathrm{d}t - 2 \int_{l}^{L} \hat{\lambda}(t) \lambda(t) \,\mathrm{d}t + \int_{l}^{L} \lambda^{2}(t) \,\mathrm{d}t \,.$$
(2.8)

In order to minimise this quantity with respect to $\hat{\lambda}$, we need to remove the dependency on λ as this quantity is unavailable in estimation contexts. We may remark that the last term can be completely ignored as it does not depend on $\hat{\lambda}$ but the middle term has to be treated differently.

As done in Reynaud-Bouret et al. (2014), we can leverage the fact that a valid approximation of $\lambda(t) dt$ is N(t). This is justified by the fact that locally $\mathbb{E}[N(t)] = \mathbb{E}[\Lambda(t)]$ and that λ is the Stieltjes-Lebesgue derivative of Λ . We can then minimise the following expression:

$$\int_{l}^{L} \hat{\lambda}^{2}(t) \,\mathrm{d}t - 2 \int_{l}^{L} \hat{\lambda}(t) N(\mathrm{d}t) \,,$$

for an observation of process N.

Non-parametric approaches to minimise this quantity tend to approximate the interaction function h by histograms (Lemonnier et al. 2014; Reynaud-Bouret et al. 2014) or by using autoregressive models (Kirchner 2017).

Other non-parametric methods include solving Wiener-Hopf equations (Bacry et al. 2016), optimising a penalised log-likelihood through an EM algorithm (Lewis et al. 2011) and fitting second and third-order cumulants (Achab et al. 2016).

2.4.3 A Bayesian estimation approach

The Bayesian approach for the inference of Hawkes processes consists in estimating the posterior distribution for the parameters of the intensity function. The method we present here follows the procedure by Rasmussen (2013) which leverages the expression of the likelihood. We present an application for the exponential kernel function $h(\cdot) = \alpha e^{-\beta}$.

Let $\theta = (\mu, \alpha, \beta) \sim \Pi$ be a random vector of parameters with density π and, conditionally on

 θ , let N be a linear Hawkes process with intensity λ_{θ} parametrised by θ . We define the likelihood L_T (Equation (2.2)) of N, in a time window [0, T] and for a history \mathcal{H}_T , as:

$$L_T(\theta) = \left(\prod_{k=1}^{N(T)} \lambda_{\theta}(T_k^-)\right) e^{-\Lambda_{\theta}(T)}$$

By the Bayes rule, the posterior distribution of θ conditional on the observation of N in [0, T]admits a density $\pi(\cdot | \mathcal{H}_T)$ proportional to the prior π and the likelihood L_T :

$$\pi(\theta \mid \mathcal{H}_T) \propto \pi(\theta) L_T(\theta)$$
.

An estimator $\hat{\theta} = \mathbb{E}[\theta \mid \mathcal{H}_T]$ can be approximated by sampling the posterior distribution of $\theta \mid \mathcal{H}_T$, which has the advantage of providing credibility intervals.

For this, we implement a Metropolis-Hastings algorithm with multivariate isotropic normal distribution as the proposition kernel. Let us illustrate the principle with the baseline intensity μ : for a current parameter μ_l , a candidate $\tilde{\mu}_l \mid \mu_l \sim \mathcal{N}(\mu_l, \sigma^2)$ is sampled and the acceptance ratio reads:

$$H_{\mu} = \frac{\pi(\tilde{\mu}_{l}, \alpha, \beta)}{\pi(\mu_{l}, \alpha, \beta)} \left(\prod_{k=1}^{N(T)} \frac{\lambda_{(\tilde{\mu}_{l}, \alpha, \beta)}(T_{k}^{-})}{\lambda_{(\mu_{l}, \alpha, \beta)}(T_{k}^{-})} \right) e^{-(\tilde{\mu}_{l} - \mu_{l})T}.$$

This allows us to obtain approximatively a sample according to the posterior distribution and provides an estimation $\hat{\theta}$ by averaging.

Further works in Bayesian contexts in the multivariate setting include Blundell et al. (2012) for Hawkes processes for the study of a social interaction graph, Donnet et al. (2020) with theoretical results concerning the concentration rates of posterior distributions. These methods have been recently adapted to include inhibition through non-linear Hawkes processes. In Deutsch et al. (2024), the authors propose to compute the log-likelihood in a multivariate parametric setting with a time-dependent baseline μ . They provide an exact computation of the log-likelihood for exponential kernels with shared decaying parameters ($\beta_{ij} = \beta \in \mathbb{R}$), and an approximation of it by a Simpson's rule suited for more general cases. Additionally, the model is reparametrised with respect to the expected total number of offsprings in order to provide normal priors better adapted for higher dimensional processes. A non-parametric estimation method is proposed in Sulem et al. (2024) which derives posterior concentration rates and in Sulem et al. (2023) through a variational Bayes procedure with a sparsity-inducing paradigm.

2.5 Simulation

2.5.1 Ogata's thinning simulation

The most classic method to simulate point processes with an intensity function λ is obtained through a rejection algorithm. An important condition to implement this method is for the intensity function to be upper-bounded by a constant λ^* . We may then simulate N by first simulating a homogeneous Poisson process with intensity λ^* and then keeping each point with probability $\lambda(t)/\lambda^*$.

In general, it is not possible to find an upper bound for the intensity of a Hawkes process. Additionally, the intensity function is dependent on each simulated point so each time a point appears, the intensity has to be updated. In order to account for these particularities of a Hawkes process, one of the most used methods is Ogata's thinning algorithm (Ogata 1981). To simulate the points of a process N in a window of time [0, T] (with the convention that $T_0 = 0$), we implement the following paradigm.

- 1. Assume that event time $T_k < T$ has been simulated.
- 2. Compute an upper bound λ^* of $t \mapsto \lambda(t \mid \mathcal{H}_{T_k})$ on the interval $[T_k, T]$.
- 3. Simulate a candidate point t_{cand} according to an exponential distribution with mean $1/\lambda^*$.
- 4. Compute the intensity $\lambda(t_{cand})$.
- 5. Accept the candidate with probability $\lambda(t_{cand})/\lambda^{\star}$.

This algorithm generalises the case when a global upper bound is known by allowing for upper bounds to be local instead.

For the simulation of a Hawkes process, we recall that the interaction function h is such that:

$$||h||_1 < 1$$
,

so in particular it is upper-bounded and attains its upper bound noted $||h||_{\infty}$. We can then sequentially determine a local upper bound. Without any other constraints on h, for a set of simulated event times (T_1, \ldots, T_k) , a local upper bound for $t \mapsto \lambda(t \mid \mathcal{H}_{T_k})$ on the interval $[T_k, +\infty]$ (assuming no other points) is given by:

$$\lambda^{\star} = \mu + k \|h\|_{\infty} \,.$$

This is clearly a very rough bound and can be easily improved under certain conditions, for example if h is a decreasing function. In this case, then λ is piecewise decreasing and so the upper bound becomes:

$$\lambda^{\star} = \lambda(T_k) + h(0)$$

Choosing a smaller upper-bound improves the simulation time by reducing the overall number of candidates and rejections.

Algorithm 2 displays the adaptation of Ogata's thinning algorithm to Hawkes processes for a non-increasing interaction function.

2.5.2 Simulation through branching theory

Another simulation algorithm is based on what is called the branching structure of a Hawkes process. By Definition 2.3.1, the Hawkes process can be seen as a branching Poisson process as described below.

- 1. Let N_c be a homogeneous Poisson process with intensity μ . All event times $(T_k^c)_{k \in \mathbb{Z}}$ are known as parents.
- 2. Each parent T_k^c generates a subsidiary process of descendants C_k of event times as follows.
 - A first generation $C_{k,1}$ is generated as an inhomogenous Poisson process with intensity $h(\cdot T_k^c)$.
 - Each child point t in $C_{k,1}$ generates a new generation $C_{k,2}$ inhomogenous Poisson process with intensity $h(\cdot t)$.

Algorithm 2: Thinning algorithm for monotone self-exciting Hawkes process.

Input Parameters μ , h a non-increasing function, and a stopping criteria (end-time T or maximal number of jumps N_{max}); **Initialization** Initialize $\lambda_k = \mu$, $t_k = 0$ and list of times $\mathcal{T} = \emptyset$; while Stopping criteria not fulfilled **do** Set $\lambda^* = \lambda_k$; Generate candidate time $t_{cand} = t_k - \frac{\log(U_1)}{\lambda^*}$, $U_1 \sim U([0, 1])$; Compute intensity $\lambda_k = \lambda(t_{cand})$ using sequence of times \mathcal{T} ; Sample $U_1 \sim U([0, 1])$; if $U_1 \leq \frac{\lambda_k}{\lambda^*}$ then Add t_{cand} to sequence of times \mathcal{T} ; Update $\lambda_k = \lambda_k + h(0)$; end Set $t_k = t_{cand}$; end return the sequence of jumps \mathcal{T} .

- This is repeated for each generated point until no new points are born. C_k is formed by the union of all points in each subprocess $C_{k,j}$.
- 3. The process N composed of the ordered union of all parents $(T_k^c)_k$ and all points in every subprocess $(C_k)_k$ is a Hawkes process.

Branching Poisson processes were introduced in Bartlett (1963) and Lewis (1964) and allow to leverage the theory of branching processes to study Hawkes processes. In particular, this procedure can be used to simulate a Hawkes process N as presented in Algorithm 3.

Algorithm 3: Branching simulation algorithm for self-exciting Hawkes process

Input Parameters μ , h a positive function, and an end-time T; **Initialization** Initialize list of times \mathcal{T} , $t_k = 0$ and auxiliary empty list L_{aux} ; Generate the number of parents N_0 according to a Poisson distribution with parameter $\mu T;$ Generate parent event times $(T_k^c)_k$ as N_0 independent and uniformly distributed points in [0, T], that are then sorted; Add all points $(T_k^c)_k$ to L_{aux} and \mathcal{T} ; while L_{aux} is not empty do for t in L_{aux} do Generate a number of children with Poisson distribution of parameter $||h||_1$; Generate the children times distributed according to the probability density function $h(\cdot - t)/||h||_1$.; Add all points inside the simulation window [0,T] to L_{aux} and \mathcal{T} ; end end return Ordered sequence of jumps \mathcal{T} .

The fact that such an algorithm will end, in other words that each subsidiary process will end up by not generating any point, is a direct consequence of the condition $||h||_1 < 1$. This is a result of Galton-Watson process theory, as each point tends to generate on average less than a children, the branches will "die out" eventually.

This algorithm presents some advantages when compared to Ogata's procedure. First, it does not require to compute an upper-bound for the intensity function, which as seen previously, is an important step in Algorithm 2. Second, this procedure does not have a rejection step and there is no need to compute the value λ each time that a point is simulated, which can greatly reduce the overall computation time. Overall, the only constraint on h imposed by Algorithm 3 is to be able to simulate according to the probability density function $h(\cdot - t)/||h||_1$. Nevertheless, this method is not longer valid to simulate non-linear Hawkes processes that lack the branching structure of the linear Hawkes process.

Maximum Likelihood Estimation for Hawkes Processes with self-excitation or inhibition

In this chapter, we present a maximum likelihood method for estimating the parameters of a univariate Hawkes process with self-excitation or inhibition. Our work generalizes techniques and results that were restricted to the self-exciting scenario. The proposed estimator is implemented for the classical exponential kernel and we show that, in the inhibition context, our procedure provides more accurate estimations than current alternative approaches.

Outline of the current chapter

3.1	Introduction	31
3.2	The Hawkes process	32
3.3	Maximum likelihood estimation and the exponential model	33
3.4	Goodness-of-fit	36
3.5	Numerical Results	36
3.6	Discussion	37
3.A	Proof of Proposition 3.3.2	38
3.B	Simulation algorithm	39

3.1 Introduction

The Hawkes model is a point process observed on the real line, which generally corresponds to the time, where any previously encountered event has a direct influence on the chances of future events occurring. This past-dependent mathematical model was introduced in Hawkes (1971) and its first application was to model earthquakes occurrences (Ogata 1988; Ogata 1998). Since then, Hawkes processes have been widely used in various fields, for instance finance (Bacry et al. 2013), social media (Rizoiu et al. 2017; Mishra et al. 2016), epidemiology (Rizoiu et al. 2018), sociology (Linderman et al. 2014) and neuroscience (Reynaud-Bouret et al. 2014).

The main advantage of Hawkes processes is their ability to model different kinds of relationships between phenomena through an unknown kernel or transfer function. The Hawkes model was originally introduced as a self-exciting point process where the appearance of an event increases the chances of another one triggering. Several estimation procedures have been proposed for the kernel function, both in parametric (Ogata 1988; Da Fonseca et al. 2013; Ozaki 1979) and nonparametric (Reynaud-Bouret et al. 2014; Bacry et al. 2016) frameworks.

However, the inhibition setting, where the presence of an event decreases the chance of another occurring, has drawn less attention in the literature, although it can be of great interest in several fields, in particular in neuroscience (Lambert et al. 2018). In this inhibition context, the cluster representation (Hawkes et al. 1974) on which is based the construction of a self-exciting Hawkes process, is no longer valid. While the existence and the construction of such nonlinear processes can be found in recent works for the univariate (Costa et al. 2020) and multivariate (Chen et al. 2017) cases, statistical estimation of the kernel function has been hardly addressed. A first approach consists in computing an approximation of the likelihood as if the intensity function could take negative values, and optimizing it to get a maximum likelihood estimator (Lemonnier et al. 2014). Alternatively, the type of interaction (excitation or inhibition) can be considered as a hidden variable, giving rise to a very practical estimation method (Mei et al. 2017).

In this chapter, we propose a maximum likelihood procedure that can handle both excitation and inhibition scenarios for a univariate Hawkes process. Our approach is based on an explicit computation of the likelihood for any type of monotone kernel functions, which is facilitated by the introduction of the natural concept of restart points. The latter are the times when the intensity function, that can be null on some intervals, become strictly positive again. We show that these restart points have a closed-form expression when the kernel is exponential, which allows us to rewrite and maximize the likelihood without approximations that are proposed for instance in Lemonnier et al. (2014). Our estimator is implemented in Python (the code is freely available online¹). We also propose a numerical study which shows the good performance of our exact estimation procedure compared to approximated approaches, especially when the intensity function is frequently equal to zero.

To outline the chapter, besides a quick introduction to self-regulating Hawkes processes (also referred to as self-correcting Hawkes processes or Hawkes processes with inhibition), Section 3.2 introduces the concepts of underlying intensity function and restart points. General results concerning the compensator and the exact maximum likelihood estimation procedure are described in Section 3.3. At last, after a brief discussion about goodness-of-fit in Section 3.4, Section 3.5 concludes with a numerical study of the estimation error.

3.2 The Hawkes process

Let N be a point process on \mathbb{R}^*_+ , here $\mathbb{R}^*_+ = \{x > 0 : x \in \mathbb{R}\}$, and $(T_k)_{k \ge 1}$ its associated event times (with convention $T_0 = 0$). For any $t \ge 0$, let us note $N(t) = \sum_{k \ge 1} \mathbb{1}_{T_k \le t}$ the number of events in [0, t] (where $\mathbb{1}$. stands for the indicator function), and λ its conditional intensity function (Daley et al. 2003):

$$\lambda(t) = \lim_{h \to 0} \frac{\mathbb{P}(N(t+h) - N(t) > 0)}{h} \,.$$

¹https://github.com/migmtz/hawkes-inhibition-expon

A univariate Hawkes process is a point process defined by the conditional intensity function:

$$\lambda(t) = \left(\lambda_0 + \int_0^t h(t-s) \,\mathrm{d}N(s)\right)^+ = \left(\lambda_0 + \sum_{T_k \le t} h(t-T_k)\right)^+, \tag{3.1}$$

where $x^+ = \max(0, x)$ denotes the positive part of any real value $x, \lambda_0 \in \mathbb{R}^*_+$ is the baseline intensity and $h : \mathbb{R}_+ \to \mathbb{R}$ is the kernel, which is assumed to be a monotone measurable function with $\lim_{t\to\infty} h(t) = 0$. The kernel function h is the key component of a Hawkes process: it translates the influence (generally assumed to fade away over time) of a past event over the process. Here, h is allowed to take negative values, meaning that it can model both self-exciting and self-regulating Hawkes processes.

Working with such Hawkes processes may prove to be difficult as the positive part function is non-linear. In particular, while computing the compensator function (Daley et al. 2003)

$$\Lambda(t) = \int_0^t \lambda(t) \, \mathrm{d}t \,, \quad \forall t \ge 0, \tag{3.2}$$

is very easy in the self-exciting case (by linearity of the intensity), it becomes more challenging for the self-regulating Hawkes process. As it is the keystone to derive the likelihood function (and then to obtain a parametric estimation method), our first contribution is to provide an exact expression of the compensator.

For this purpose, let us first introduce the *underlying intensity function* and the *restart time*, two quantities which will allow us to derive the computation of the likelihood of a monotone Hawkes process, in a framework unifying self-correcting and self-exciting Hawkes processes.

Definition 3.2.1. Let the underlying intensity function of N be:

$$\lambda^{\star}(t) = \lambda_0 + \int_0^t h(t-s) \,\mathrm{d}N(s) \,.$$

In addition, let the restart time T_k^{\star} be, for any positive integer k:

$$T_k^{\star} = \inf \left\{ t \ge T_k \mid \lambda(t) > 0 \right\},$$

along with its corresponding cooldown interval $\tau_k^* = T_k^* - T_k$.

As illustrated in Figure 3.1, λ^* corresponds to the intensity λ as if it were allowed to take negative values. Moreover, as the kernel is assumed to be monotone, the restart time associated to one occurrence can be interpreted as the first moment after this occurrence from which λ and λ^* become equal (in particular, the restart time and the occurrence time coincide if the intensity function is nonnegative at this time, see Figure 3.1):

$$T_k^{\star} = \inf \left\{ t \ge T_i \mid \forall t \in (T_k^{\star}, T_{k+1}), \, \lambda(t) = \lambda^{\star}(t) \right\}.$$

3.3 Maximum likelihood estimation and the exponential model

Assume a parametric model $\mathcal{P} = \{\lambda_{\theta}, \theta \in \Theta\}$ for the conditional intensity function λ , where θ contains unknown quantities such as the baseline λ_0 and the kernel h. Then, with convention $log(t) = -\infty$ for $t \leq 0$, the log-likelihood ℓ_t of any $\theta \in \Theta$ with respect to the observations



Figure 3.1: Example of the intensity (red curve) and underlying intensity (blue curve) for a self-regulating Hawkes process, with the associated restart times. We only see the negative values of the blue curve since they precisely correspond to the values for which the two intensity functions are not equal.

 $T_1, \ldots, T_{N(t)}$ in the time interval [0, t] is (Daley et al. 2003, Proposition 7.2.III.), (Ozaki 1979):

$$\ell_t(\theta) = \sum_{k=1}^{N(t)} \log\left(\lambda_\theta(T_k^-)\right) - \Lambda_\theta(t), \qquad (3.3)$$

where the compensator Λ_{θ} is defined as in Equation (3.2) and $\lambda_{\theta}(T_k^-) = \lim_{t \to T_k^-} \lambda_{\theta}(t)$.

Equation (3.3) reveals the importance of being able to compute the compensator Λ (equivalently Λ_{θ}) in order to provide a practical implementation of the maximum likelihood estimator of λ . Thus, a first contribution of this work lies in Proposition 3.3.1, which establishes a decomposition of the compensator Λ using the underlying intensity function λ^* and the restart times $T_1^*, \ldots, T_{N(t)}^*$.

Proposition 3.3.1. For any t > 0, the compensator Λ can be expressed as:

$$\Lambda(t) = \begin{cases} \lambda_0 t & \text{if } t < T_1 \\ \lambda_0 T_1 + \sum_{k=1}^{N(t)-1} \int_{T_k^{\star}}^{T_{k+1}} \lambda^{\star}(u) \, \mathrm{d}u + \int_{T_{N(t)}^{\star}}^{t} \lambda^{\star}(u) \, \mathrm{d}u & \text{if } t \ge T_1 \,, \end{cases}$$
(3.4)

with the conventions that the sum is equal to 0 if N(t) = 1 and the last integral is equal to 0 if $t < T^*_{N(t)}$.

Proof. This comes directly from splitting the integral of $\Lambda(t) = \int_0^t \lambda(t) dt$ on the intervals $[T_k, T_{k+1})$ $(k \in \{0, \dots, N(t) - 1\})$ and $[T_{N(t)}, t]$, and by remarking that, since h is monotone, $\forall t \in [T_k, T_{k+1}), \lambda(t) = \lambda^*(t) \mathbb{1}_{[T_k^*, T_{k+1})}(t).$

In order to give an explicit computation of the quantity $\int_{T_k^{\star}}^{T_{k+1}} \lambda^{\star}(u) du$ (equivalently $\int_{T_{N(t)}}^{t} \lambda^{\star}(u) du$) which appears in Proposition 3.3.1, we focus on the classical scenario where we consider an ex-

ponential kernel $h(t) = \alpha e^{-\beta t}$, for some $\alpha \in \mathbb{R}$ and $\beta \in \mathbb{R}^*_+$. Let us notice that α can be either positive or negative, meaning that the process may be either self-exciting or self-regulating.

Then, the underlying intensity function can be written as:

$$\lambda^{\star}(t) = \lambda_0 + \int_0^t \alpha \mathrm{e}^{-\beta(t-s)} \,\mathrm{d}N(s) \,. \tag{3.5}$$

The forthcoming proposition steps forward in computing the compensator for an exponential kernel.

Proposition 3.3.2 (Compensator for exponential kernel). Let t > 0 and $k \in \{1, ..., N(t)\}$. The restart times read:

$$T_k^{\star} = T_k + \beta^{-1} \log \left(\frac{\lambda_0 - \lambda^{\star}(T_k)}{\lambda_0} \right) \mathbb{1}_{\lambda^{\star}(T_k) < 0},$$

and the compensator is expressed as in Equation (3.4), with, for any $\tau \in [T_k^*, T_{k+1}]$:

$$\int_{T_k^*}^{\tau} \lambda^*(u) \, \mathrm{d}u = \lambda_0 (\tau - T_k^*) + \beta^{-1} (\lambda^*(T_k) - \lambda_0) (\mathrm{e}^{-\beta(T_k^* - T_k)} - \mathrm{e}^{-\beta(\tau - T_k)})$$

Proof. The proof is in 3.A.

Corollary 3.3.1 (Log-likelihood for exponential kernel). Let

$$\mathcal{P} = \left\{ \lambda_{\theta} = \bar{\lambda_0} + \int_0^t \bar{\alpha} e^{-\bar{\beta}(t-s)} \, \mathrm{d}N(s) : \theta = (\bar{\lambda_0}, \bar{\alpha}, \bar{\beta}) \in \Theta \right\} \,, \tag{3.6}$$

be a parametric exponential model for the conditional intensity function λ with $\Theta = \mathbb{R}^*_+ \times \mathbb{R} \times \mathbb{R}^*_+$, along with the candidate compensator Λ_{θ} , the underlying intensity function λ^*_{θ} and the restart times $T^*_{\theta,1}, \ldots, T^*_{\theta,N(t)}$ associated to λ_{θ} (see Equation (3.2) and Definition 3.2.1).

For any $\theta = (\overline{\lambda_0}, \overline{\alpha}, \overline{\beta}) \in \Theta$, by denoting

.....

$$\Lambda_{\theta,k} = \bar{\lambda}_0 (T_k - T_{\theta,k-1}^{\star}) + \bar{\beta}^{-1} (\lambda_{\theta}^{\star}(T_{k-1}) - \bar{\lambda}_0) (e^{-\bar{\beta}(T_{\theta,k-1}^{\star} - T_{k-1})} - e^{-\bar{\beta}(T_k - T_{k-1})})$$

the log-likelihood reads (with convention $\log(x) = -\infty$ for $x \leq 0$):

$$\ell_{t}(\theta) = \log \bar{\lambda_{0}} - \bar{\lambda_{0}}T_{1} + \sum_{k=2}^{N(t)} \left[\log \left(\bar{\lambda_{0}} + (\lambda_{\theta}^{\star}(T_{k-1}) - \bar{\lambda_{0}}) e^{-\bar{\beta}(T_{k} - T_{k-1})} \right) - \Lambda_{\theta,k} \right] \\ - \left[\bar{\lambda_{0}}(t - T_{\theta,N(t)}^{\star}) + \bar{\beta}^{-1} (\lambda_{\theta}^{\star}(T_{N(t)}) - \bar{\lambda_{0}}) \left(e^{-\bar{\beta}(T_{\theta,N(t)}^{\star} - T_{N(t)})} - e^{-\bar{\beta}(t - T_{N(t)})} \right) \right] \mathbb{1}_{t > T_{\theta,N(t)}^{\star}}.$$
(3.7)

Proof. By Equation (3.8) in the proof of Proposition 3.3.2,

$$\lambda_{\theta}^{\star}(T_k^-) = \begin{cases} \bar{\lambda}_0 & \text{if } k = 1, \\ \bar{\lambda}_0 + (\lambda_{\theta}^{\star}(T_{k-1}) - \bar{\lambda}_0) e^{-\bar{\beta}(T_k - T_{k-1})} & \text{if } k \ge 2. \end{cases}$$

Combining this expression with Propositions 3.3.1 and 3.3.2 leads to the result.

Corollary 3.3.1 exhibits that the log-likelihood for self-regulating Hawkes processes with an exponential kernel can be evaluated in O(N(t)) operations (by computing iteratively the quantities

potential kernel can be evaluated in O(N(t)) operations (by computing iteratively the quantities $T_{\theta,k}^*$ and $\Lambda_{\theta,k}$ appearing in the summation of Equation (3.7)), as already known for self-exciting exponential Hawkes processes Laub (2014, Chapter 4.2). For other monotone kernels without the Markov property, evaluating the log-likelihood with the method proposed here requires $O(N(t)^2)$ operations, similarly to existing approaches for self-exciting Hawkes processes.

3.4 Goodness-of-fit

Even though computing the compensator Λ (equivalently Λ_{θ}) was clearly motivated by maximum likelihood estimation, it turns out that it is of great benefit to assess goodness-of-fit, and in particular to check the validity of a maximum likelihood estimation. This is possible thanks to the Time Change Theorem, a result originally stated for inhomogeneous Poisson processes.

Theorem 3.4.1 ((Daley et al. 2003, Theorem 7.4.IV)). Assume that Λ is continuous, monotone and $\Lambda(t) \xrightarrow[t \to \infty]{t \to \infty} \infty$ a.s. Then a.s., a sequence of event times $(U_k)_{k \ge 1}$ is a realization of N if and only if $(\Lambda(U_k))_{k \ge 1}$ is a realization of a homogeneous Poisson process with unit intensity.

Let us note that we can find applications of Theorem 3.4.1 to self-exciting Hawkes processes in the literature (Laub 2014, Chapter 5). Since for self-regulating Hawkes processes Λ is still monotone, this result can also be applied in our case.

To be more precise, let us consider $\theta \in \Theta$ and the null hypothesis: " $(U_k)_{k\geq 1}$ is a realization of an exponential Hawkes process with parameter θ ". This hypothesis can be tested by applying a Kolmogorov-Smirnov test between the empirical distribution of $(\Lambda_{\theta}(U_{k+1}) - \Lambda_{\theta}(U_k))_{k\geq 1}$ and an exponential distribution with parameter 1. This procedure is illustrated in Table 3.1, Section 3.5.

3.5 Numerical Results

This section is aimed at assessing the maximum likelihood estimation method for self-regulating Hawkes processes, based on the exact computation of the compensator Λ_{θ} in the exponential model (3.6) (Corollary 3.3.1). This procedure is compared to the approximated maximum likelihood estimation proposed in Lemonnier et al. (2014), which consists in approximating Λ_{θ} by:

$$\Lambda_{\theta}^{LM}(t) = \int_0^t \lambda_{\theta}^{\star}(u) \,\mathrm{d}u \,.$$

This optimization procedure is performed with the L-BFGS-B algorithm from the Scipy package (with (1,0,1) as a starting guess and a bounds argument such that $\lambda_0 \ge 0, \alpha \in \mathbb{R}, \beta \ge 0$). In other words, estimators are:

$$\hat{\theta} \in \arg \max_{\theta \in \Theta} \left\{ \ell_{T_{N_{max}}}(\theta) = \sum_{k=1}^{N_{max}} \log \left(\lambda_{\theta}(T_k^-) \right) - \Lambda_{\theta}(T_{N_{max}}) \right\},\$$

where $N_{max} = 200$ is the total number of jumps and Λ_{θ} can be replaced by Λ_{θ}^{LM} to obtain the approximated likelihood proposed in Lemonnier et al. (2014).

The comparison between the exact and the approximated estimation procedure is based on simulated data sets coming from self-correcting Hawkes processes of the form (3.6) with 6 different values of $\theta = (\bar{\lambda}_0, \bar{\alpha}, \bar{\beta}) \in \Theta$ (see Table 3.1) which have been chosen in order to explore different

scenarios, in particular depending on whether the intensity function is frequently null or not. Observations are sets of time jumps generated with a sampling algorithm (see the algorithm in 3.B and Python implementation online), which is a particular case of Ogata's thinning simulation method (Ogata 1981) that can handle Hawkes processes with either self-excitation or inhibition.



Figure 3.2: Top panel: relative absolute errors of estimations $\hat{\theta} = (\hat{\lambda}_0, \hat{\alpha}, \hat{\beta})$. Bottom panel: example of simulated intensities for each set of values $\theta = (\bar{\lambda}_0, \bar{\alpha}, \bar{\beta})$ with the corresponding average percentage of time when the intensities are equal to zero.

Figure 3.2 represents the relative absolute errors of estimations $\hat{\theta} = (\hat{\lambda}_0, \hat{\alpha}, \hat{\beta})$ for each of the 6 simulated models. We observe that the exact approach provides more accurate estimations than the approximated procedure (as illustrated in the boxplots of Figure 3.2 and by the *p*-values of the goodness-of-fit tests in Table 3.1). As expected, the more time the conditional intensity equals 0 (from left to right in Figure 3.2), the greater the differences between the two procedures. Furthermore, the leftmost boxplot confirms that when the underlying intensity is nonnegative both methods are mostly identical. Let us note that in this case the estimation of $\bar{\alpha}$ is rather wrong (the estimation of $\bar{\beta}$ is impacted consequently) probably because its value is close to 0 compared to the magnitude of $\bar{\lambda}_0$.

3.6 Discussion

In this chapter we proposed a maximum likelihood approach for Hawkes processes that can handle both self-exciting and self-regulating scenarios, the first case being already covered in the literature and the latter being our main contribution. For this purpose, we define the concepts

	Parameters			Estimations			
	$ar{\lambda}_0$	\bar{lpha}	$ar{eta}$	$\hat{\lambda}_0$	\hat{lpha}	\hat{eta}	p-value
Exact	0.5	0.001	0.4	0.52	0.03	2.13	0.38
Approx	0.5	-0.001		0.52	0.03	2.11	0.38
Exact	0.5	0.9	0.4	0.51	-0.21	0.45	0.42
Approx	0.5	-0.2	0.4	0.51	-0.22	0.47	0.42
Exact	t 1.05	0.75	0.8	1.06	-0.76	0.83	0.43
Approx	1.05	-0.75	0.8	1.12	-0.88	0.84	0.35
Exact	9 49	0.08	3 0.4	2.59	-1.00	0.38	0.53
Approx	2.43	-0.98		2.90	-1.21	0.40	0.45
Exact	rox 2.85	95	10	2.81	-2.56	1.87	0.36
Approx		-2.0	1.0	$1.55 imes 10^4$	$-1.63 imes10^7$	3.04	0.07
Exact	1.6	1.6 -0.75	0.1	1.62	-0.76	0.11	0.42
Approx			0.1	2.72×10^7	-2.31×10^{10}	0.27	2.14×10^{-05}

Table 3.1: Quantitative assessment of the numerical study: sets of true parameters (left), average estimations over 50 repetitions (middle) and average p-values over 50 independent realisations for the test of Section 3.4.

of underlying intensity function and restart times when working with monotone kernel functions. In particular we obtain exact expressions of the compensator for the exponential Hawkes process which is the key step of the estimation procedure. We present numerical results on synthetic data that show the efficiency of our procedure, with a substantial improvement compared to approximated approaches when the intensity function is frequently null.

From a theoretical point of view, future work will consist in adapting analytical results to study the convergence of our estimator in the self-regulating case. Regarding modeling, it would be of great interest to consider kernel functions outside the classical exponential scenario. Another important step is the extension of our concepts and algorithms to the multivariate version of the process, which is not straightforward since in the multivariate setting the expression of the restart times are no longer explicit. This last point is essential in order to target real-world datasets since in many applications, being limited to the univariate case will lead to detect selfexcitation. However, a model that accounts for potential inhibition effects is of great interest when considering interactions between events of different natures, which will typically be modeled by a multivariate process. This multidimensional extension is the object of a future work, with a further perspective to use our procedure in neuroscience applications in order to detect attraction and repulsion effects between neurons.

3.A Proof of Proposition 3.3.2

Let us begin by expressing the underlying intensity function between two event times. First, $\lambda^*(t) = \lambda_0$ for $t \in [0, T_1)$. Then, for any $k \in \mathbb{N}$, for all $t \in [T_k, T_{k+1})$, the underlying intensity is differentiable in t and

$$(\lambda^{\star})'(t) = -\beta(\lambda^{\star}(t) - \lambda_0),$$

with the left condition: $\lambda_k^\star := \lambda^\star(T_k)$. Solving this differential equation leads to

$$\lambda^{\star}(t) = \lambda_0 + (\lambda_k^{\star} - \lambda_0) e^{-\beta(t - T_k)} .$$
(3.8)

Now, by definition of the restart time $T_k^* = \inf \{t \ge T_k \mid \lambda(t) > 0\}$, we have that if $\lambda_k^* \ge 0$, then $T_k^* = T_k$. Otherwise, as λ^* is continuous on the interval $[T_k, T_{k+1})$, we obtain T_k^* by solving for t: $\lambda^*(t) = 0$. Thus, by Equation (3.8):

$$\lambda^{\star}(T_k^{\star}) = 0 \iff T_k^{\star} = T_k + \beta^{-1} \log\left(\frac{\lambda_0 - \lambda_k^{\star}}{\lambda_0}\right).$$

Gathering both situations, we obtain the first part of Proposition 3.3.2:

$$T_k^{\star} = T_k + \beta^{-1} \log \left(\frac{\lambda_0 - \lambda_k^{\star}}{\lambda_0} \right) \mathbb{1}_{\lambda_k^{\star} < 0}$$

Let now $k \in \{1, \ldots, N(t)\}$ and $\tau \in [T_k^{\star}, T_{k+1}]$. By Equation (3.8),

$$\int_{T_k^\star}^{\tau} \lambda^\star(u) \,\mathrm{d}u = \int_{T_k^\star}^{\tau} \left(\lambda_0 + (\lambda_k^\star - \lambda_0) \mathrm{e}^{-\beta(u - T_k)}\right) \,\mathrm{d}u$$
$$= \lambda_0(\tau - T_k^\star) + \beta^{-1}(\lambda_k^\star - \lambda_0) (\mathrm{e}^{-\beta(T_k^\star - T_k)} - \mathrm{e}^{-\beta(\tau - T_k)}),$$

which is the second part of Proposition 3.3.2.

3.B Simulation algorithm

Algorithm 4 builds upon Ogata's thinning simulation method (Ogata 1981, Proposition 1) in order to handle Hawkes processes with either self-excitation or inhibition.

Algorithm 4: Thinning algorithm for monotone Hawkes process.

Inference of multivariate exponential Hawkes processes with inhibition and application to neuronal activity

The multivariate Hawkes process is a past-dependent point process used to model the relationship of event occurrences between different phenomena. Although the Hawkes process was originally introduced to describe excitation interactions, which means that one event increases the chances of another occurring, there has been a growing interest in modelling the opposite effect, known as inhibition. In this chapter, we focus on how to infer the parameters of a multidimensional exponential Hawkes process with both excitation and inhibition effects. Our first result is to prove the identifiability of this model under a few sufficient assumptions. Then we propose a maximum likelihood approach to estimate the interaction functions, which is, to the best of our knowledge, the first exact inference procedure in the frequentist framework. Our method includes a variable selection step in order to recover the support of interactions and therefore to infer the connectivity graph. A benefit of our method is to provide an explicit computation of the log-likelihood, which enables in addition to perform a goodness-of-fit test for assessing the quality of estimations. We compare our method to standard approaches, which were developed in the linear framework and are not specifically designed for handling inhibiting effects. We show that the proposed estimator performs better on synthetic data than alternative approaches. We also illustrate the application of our procedure to a neuronal activity dataset, which highlights the presence of both exciting and inhibiting effects between neurons.

Outline of the current chapter

4.1	Introduction	41
4.2	The multivariate Hawkes process	43
	4.2.1 Definition	43
	4.2.2 Related work	44
4.3	Estimation and goodness-of-fit	45
	4.3.1 Introductive example	45
	4.3.2 Underlying intensity and restart times in the multivariate setting	46
	4.3.3 Identifiability and likelihood computation	48
	4.3.4 On identifiability conditions	49

	4.3.5 Recovering the graph of interactions	51 52
4.4	Illustration on synthetic datasets 4.4.1 Simulation procedure 4.4.2 Proposed methods and comparison to existing procedures 4.4.3 Robustness on misspecified models	53 54 55 62
4.5	Application on neuronal data	64
	4.5.1 Preprocessing and data description	64
	4.5.2 Resampling	64
	4.5.3 Goodness-of-fit and multiple testing procedure	65
	4.5.4 Estimation results	66
4.6	Discussion	67
4.A	Proof of Lemma 4.3.1	68
4.B	Proof of Proposition 4.3.1	70
4. C	Proof of Theorem 4.3.1	70
4.D	Proof of Corollary 4.3.1	73
4. E	Algorithm for computing the log-likelihood	74
4.F	Reconstructed interaction functions for synthetic data	74

4.1 Introduction

A Hawkes process is a point process in which each point is commonly associated with event occurrences in time. In this past-dependent model, every event time impacts the probability that other events take place subsequently. These processes are characterised by the conditional intensity function, seen as an instantaneous measure of the probability of event occurrences. Since their introduction in Hawkes (1971), Hawkes processes have been applied in a wide variety of fields, for instance in seismology (Ogata 1988), social media (Rizoiu et al. 2017), criminology (Olinde et al. 2020) and neuroscience (Lambert et al. 2018).

The multidimensional version of this model, referred to as the multivariate Hawkes process, describes the appearance of different types of events, the occurrences of which are influenced by all past events of all types. Each interaction between two types of events is encoded in kernel functions, also called interaction functions. Originally this model takes only into account mutually exciting interactions - an event increases the chances of others occurring - by assuming that all kernel functions are non-negative. A specificity of self-exciting Hawkes processes is their branching structure, also known as cluster structure. Introduced in Hawkes et al. (1974), this parallel between Hawkes processes and branching theory has provided the first theoretical background for the self-exciting Hawkes model, in particular existence and expected number of points on a finite interval. Estimation methods in the literature are vast including maximum likelihood estimators (Ozaki 1979; Guo et al. 2018) and method of moments (Da Fonseca et al. 2013). Nonparametric approaches include an EM procedure introduced in Lewis et al. (2011), estimations obtained via the solution of Wiener-Hopf equations (Bacry et al. 2016) or by approximating the process through autoregressive models (Kirchner 2017) or through functions in reproducing kernel Hilbert spaces (Yang et al. 2017).

Although the self-exciting Hawkes process remains widely studied, there has been a growing interest in modeling the opposite effect, known as inhibition, in which the probability of observing an event is lowered by the occurrence of certain events. In practice, this amounts to considering negative kernel functions. In order to maintain the positivity of the intensity function, a non-linear operator is added to the expression which in turns entails the loss of the cluster representation. This model known as the non-linear Hawkes process was first presented in Brémaud et al. (1996), where existence of such processes was proved via construction using bi-dimensional marked Poisson processes. Such approach of analysis has been used in the literature as in Chen et al. (2017), where a coupling process is established and leveraged to obtain theoretical guarantees on cross-analysis covariance. Another approach is presented in Costa et al. (2020), where renewal theory allows to obtain limit theorems for processes with bounded support kernel functions. Estimation methods focus mainly on nonparametric methods for general interactions and non-linear functions, as found in Bacry et al. (2016) and Sulem et al. (2024).

In the last years, alternative models have been designed in order to take into account inhibiting effects in Hawkes processes. An example is the neural Hawkes process, presented in Mei et al. (2017) and Zuo et al. (2020), which combines a multivariate Hawkes process and a recurrent neural network architecture. In Duval et al. (2022), a multiplicative model considers two sets of neuronal populations, one exciting and another inhibiting, and each intensity function is the product of two non-linear functions (one for each group). Another model is presented in Olinde et al. (2020) and called self-limiting Hawkes process. It includes the inhibition as a multiplicative term in front of a the traditional self-exciting intensity function.

In this chapter, we present a maximum likelihood estimation method for multivariate Hawkes processes with exponential kernel functions, that works for both exciting and inhibiting interactions, as modelled by Brémaud et al. (1996) and Chen et al. (2017). This work builds upon the methodology for the univariate case, presented in Bonnet et al. (2021), by focusing in the intervals where the intensity function is positive. We show that, under a weak assumption on the kernel functions, these intervals can be determined exactly. We can then write for each dimension the integral of the intensity function, known in the literature as the compensator, which in turn provides an explicit expression of the log-likelihood. This enables to build the corresponding maximum likelihood estimator and we complete our procedure with a variable selection step to recover the significant interactions within the whole process. This is of particular interest since it provides a graphical interpretation of the model and it can also be used a reduction dimension tool. Our numerical procedure is implemented in Python and freely available on GitHub.¹ As a by-product of our method, the closed-form expression of the compensator also allows to assess goodness-of-fit via the Time Change Theorem and multiple testing. We carry out a numerical study on simulated data and on a neuronal activity dataset (Petersen et al. 2016; Radosevic et al. 2019). The performance of our approach is compared to estimations obtained via approximations from Bacry et al. (2020) and Lemonnier et al. (2014), and we show that our method not only achieves better estimations but is capable of identifying correctly the interaction network of the process.

To outline this chapter, Section 4.2 presents the multivariate Hawkes process framework and reviews the literature regarding inference of non-linear Hawkes processes. In Section 4.3, we detail our procedure, including the maximum likelihood estimation, variable selection and goodness-of-fit test to assess the quality of the estimations. We also address the question of identifiability of the model, that we prove under a few sufficient conditions. The whole procedure is illustrated on simulated data in Section 4.4 and applied to a neuronal activity dataset in Section 4.5. In Section 4.6 we discuss our contributions and its current limitations along with interesting perspectives for future work.

¹https://github.com/migmtz/multivariate-hawkes-inhibition

4.2 The multivariate Hawkes process

4.2.1 Definition

A multivariate Hawkes process $N = (N^1, N^2, \dots, N^d)$ of dimension d is defined by d point processes on \mathbb{R}^*_+ , denoted $N^i \colon \mathcal{B}(\mathbb{R}^*_+) \to \mathbb{N}$, where $\mathcal{B}(\mathbb{R}^*_+)$ is the Borel algebra on the set of positive numbers.

Each process N^i can be characterised by its associated event times $(T_k^i)_k$ and its conditional intensity function, defined for all $t \ge 0$ by

$$\lambda^{i}(t) = \left(\mu_{i} + \sum_{j=1}^{d} \int_{0}^{t} h_{ij}(t-s) \, \mathrm{d}N^{j}(s)\right)^{+}$$
$$= \left(\mu_{i} + \sum_{j=1}^{d} \sum_{T_{k}^{j} \leq t} h_{ij}(t-T_{k}^{j})\right)^{+}, \qquad (4.1)$$

where $x^+ = \max(0, x)$. Here, the quantity $\mu_i \in \mathbb{R}^*_+$ is called the baseline intensity and each interaction or kernel function $h_{ij} \colon \mathbb{R}^*_+ \to \mathbb{R}$ represents the influence of the process N^j on the process N^i and T^j_k corresponds to the k-th event time of process N^j .

Remark. The positive-part function in Equation (4.1) is needed to ensure the non-negativity of λ^i in the presence of strong inhibiting effects, that is when some interaction functions h_{ij} are sufficiently negative compared to positive contributions. Concretely, the positive part does not affect the intensity function if inhibiting effects are in minority compared to the positive contributions (exciting effects or baseline intensities).

Remark. Equation (4.1) may question the reader for two reasons. First, it is the cadlag definition of a the conditional intensity of a Hawkes process. It's our choice to prefer it to the caglad version but all the results presented here can be written in this setting. Second, it is considered that the history is empty for t < 0. It is a common choice for statistical inference (a finite amount a times is observed) while the infinite history is preferred for a probabilistic analysis based on a stationary assumption.

For each process N^i and for all $t \ge 0$, let us note $N^i(t) = \sum_{k\ge 1} \mathbb{1}_{T^i_k \le t}$ the measure of (0, t] and the compensator

$$\Lambda^i(t) = \int_0^t \lambda^i(u) \,\mathrm{d} u \,.$$

The process N can be seen as a point process on \mathbb{R}^*_+ , where for any $B \in \mathcal{B}(\mathbb{R}^*_+)$, $N(B) = \sum_{i=1}^d N^i(B)$. Similarly to a univariate process, N can be characterised by its conditional intensity λ (also called total intensity):

$$\lambda(t) = \sum_{i=1}^{d} \lambda^{i}(t) , \qquad (4.2)$$

and by its compensator

$$\Lambda(t) = \int_0^t \lambda(u) \, \mathrm{d}u = \sum_{i=1}^d \Lambda^i(t)$$

From this point of view, the process N is associated to event times $(T_{(k)})_k = (T_{u_k}^{m_k})_k$

corresponding to the ordered sequence composed of $\bigcup_{i=1}^{d} \{T_k^i \mid k > 0\}$, and we may define, for every $t \ge 0$, $N(t) = \sum_{k\ge 1} \mathbb{1}_{T_{(k)} \le t} = \sum_{i=1}^{d} N^i(t)$. Here, $(u_k)_k$ is the random ordering sequence and $(m_k)_k$ the sequence of marks that make it possible to identify to which dimension each time corresponds. These marks can be written as

$$m_k = \sum_{j=1}^d j \mathbb{1}_{N^j \left(\{ T_{(k)} \} \right) = 1} \,.$$

Remark. A more detailed introduction of multivariate point processes via the concept of marked point processes can be found in Daley et al. (2003, Chapter 6.4)

As the aim of this chapter is to describe a practical methodology for estimating the conditional intensities $\lambda_1, \ldots, \lambda_d$ via maximising the log-likelihood, the latter quantity has to be made explicit. Let $t \geq 0$ and assume that event times $\{T_k^i : 1 \leq k \leq N_i(t), 1 \leq i \leq d\}$ are observed in the interval (0, t]. Then, given a parametric model $\mathcal{P} = \{(\lambda_{\theta_1}^1, \ldots, \lambda_{\theta_d}^d) : \theta = (\theta_1, \ldots, \theta_d) \in \Theta_1 \times \cdots \times \Theta_d\}$ (and associated compensators $\Lambda_{\theta_1}^1, \ldots, \Lambda_{\theta_d}^d$) for conditional intensity functions $\lambda^1, \ldots, \lambda^d$, for every $\theta \in \Theta$, the log-likelihood $\ell_t(\theta)$ reads Daley et al. (2003, Proposition 7.3.III.)

$$\ell_t(\theta) = \sum_{i=1}^d \ell_t^i(\theta_i) \,,$$

with

$$\ell_t^i(\theta_i) = \sum_{k=1}^{N^i(t)} \log \lambda_{\theta_i}^i(T_k^{i-}) - \Lambda_{\theta_i}^i(t), \qquad (4.3)$$

where $\lambda_{\theta_i}^i(T_k^{i-}) = \lim_{t \to T_k^{i-}} \lambda_{\theta_i}^i(t)$ and with convention $\log(x) = -\infty$ for $x \leq 0$.

The heart of the problem in deriving a maximum likelihood estimator for the conditional intensities λ^i is being able to evaluate exactly the compensator values $\Lambda^i_{\theta}(t)$ for every possible $\theta \in \Theta$, which requires to determine when the conditional intensities λ^i are non-zero. The forthcoming sections clear this point up.

4.2.2 Related work

Estimation methods for Hawkes processes have focused mainly on self-exciting interactions (by assuming $h_{ij} \geq 0$). In Ozaki (1979), the author presents the maximum likelihood estimation method for univariate processes with exponential kernel $h(t) = \alpha e^{-\beta t}$ ($\alpha > 0, \beta > 0$), the same method being established in Mishra et al. (2016) for the power law kernel function $h(t) = \frac{\alpha \beta}{(1+\beta t)^{1+\gamma}}$ ($\alpha > 0, \beta > 0, \gamma > 0$). In Chen et al. (2018) the maximum likelihood method is presented for the multivariate version with exponential kernel, while Bacry et al. (2020) proposed an inference method based on optimising a least-squares criterion. Other methods in the parametric setting include spectral analysis (Adamopoulos 1976), EM algorithm (Veen et al. 2008) and method of moments (Da Fonseca et al. 2013).

Estimators of the interaction functions h_{ij} are also presented in a nonparametric setting. For instance, Yang et al. (2017) proposed to estimate h_{ij} in a reproducing kernel Hilbert space. Reynaud-Bouret et al. (2014) proposed a decomposition of the interaction functions h_{ij} on a histogram basis with bounded support, the estimation of which are obtained by minimising a least-squares contrast. Hawkes processes with excitation have also been studied in a Bayesian context, with likelihood-based approaches, as in Rasmussen (2013) for the univariate case and in Donnet et al. (2020) for multivariate processes.

Although inhibiting effects in Hawkes processes were first mentioned in Brémaud et al. (1996), they have only met a growing interest in the last decade. Concerning inference, most of the known methods are not designed for handling the inhibiting case: nevertheless some are able in practice to estimate negative interactions by minimising a least-squares criterion, but without guaranteeing that the estimated intensity functions remain non-negative (Reynaud-Bouret et al. 2014; Bacry et al. 2020). A similar approach is proposed in Lemonnier et al. (2014) for maximum likelihood estimation, where the compensator Λ^i is approximated by integrating the conditional intensity λ^i without the positive part function (see Equation (4.1)). Obviously, these methods should perform well when the intensities remain mostly positive, but it is unclear how they will adapt to scenarios when the intensities are frequently equal to zero due to inhibiting terms. A similar remark is mentioned in Bacry et al. (2016) concerning their estimation method, that can provide negative estimations of the interactions only if there is a negligible chance of the intensities to be null.

Inference procedures that are specifically dedicated to Hawkes processes with inhibition are scarcer in the literature. Sulem et al. (2024) presents various results for non-linear Hawkes processes including inhibition effects: existence, stability and Bayesian estimation for kernel functions with bounded support. Deutsch et al. (2024) presents choices of priors for Bayesian estimation based on a new reparametrisation of the process.

Lastly, Bonnet et al. (2021) presents a maximum likelihood estimation adapted to the univariate Hawkes process with inhibition and monotone kernel functions. The decisive contribution of this work is to give, for an exponential kernel $h(t) = \alpha e^{-\beta t}$ ($\alpha \in \mathbb{R}, \beta > 0$), a closed-form expression of restart times, which are defined as the instants at which the single conditional intensity becomes non-zero. This makes possible to compute explicitly the compensator and then the log-likelihood. Yet, this study is limited to the univariate case.

This work goes a step forward in estimation of multivariate Hawkes processes with inhibition, by providing the first exact maximum likelihood method for exponential interactions $h_{ij}(t) = \alpha_{ij} e^{-\beta_{ij}t}$, combined with a variable selection procedure. As it will be explained in the next section, the proposed approach also enables to perform standard goodness-of-fit tests. It has to be noted that the concurrent work of Deutsch et al. (2024) proposed a similar approach for multivariate Hawkes processes to exactly compute the compensator, but was used neither for maximum likelihood estimation, nor for goodness-of-fit tests.

4.3 Estimation and goodness-of-fit

4.3.1 Introductive example

Before motivating and explaining the estimation procedure proposed in this chapter, we present an example of multivariate Hawkes process. Figure 4.1 depicts in red conditional intensities λ^1 and λ^2 for a realisation of a 2-dimensional Hawkes process (see the forthcoming section for the definition of underlying intensities). The existence of such a process (along with its stationarity) is ensured by controlling the spectral radius $\rho(S^+) < 1$ of the matrix $S^+ = (\|h_{ij}^+\|_1)_{ij}$ (Deutsch et al. 2024). Similar results with slightly different conditions can be found in Brémaud et al. (1996) and Sulem et al. (2024). The simulation has been carried out with baselines $\mu_1 = 0.5$ and $\mu_2 = 1.0$, and with exponential kernels $h_{ij}(t) = \alpha_{ij} e^{-\beta_{ij}t}$ parameterised by:

$$\begin{pmatrix} \alpha_{11} & \alpha_{12} \\ \alpha_{21} & \alpha_{22} \end{pmatrix} = \begin{pmatrix} -1.9 & 3.0 \\ 0.9 & -0.7 \end{pmatrix}, \quad \text{and} \quad \begin{pmatrix} \beta_{11} & \beta_{12} \\ \beta_{21} & \beta_{22} \end{pmatrix} = \begin{pmatrix} 2.0 & 20.0 \\ 3.0 & 2.0 \end{pmatrix}$$

These kernels have been chosen such that both processes are self-inhibiting $(\alpha_{11}, \alpha_{22} < 0)$ but inter-exciting $(\alpha_{12}, \alpha_{21} > 0)$.



Figure 4.1: Simulation of a 2-dimensional Hawkes process. Each cross corresponds to an event time, and each $T_{(k)}$ is shown in its corresponding process.

The goal of this work is to establish a parametric estimation method, via maximum likelihood estimation, that is able to handle both excitation and inhibition frameworks in the multivariate case. For this purpose, it is necessary to compute explicitly the log-likelihood $\ell_t(\theta)$ (see Equation (4.3)) and in particular to evaluate the compensator Λ^i_{θ} , expressed as an integral of λ^i_{θ} . For the latter, the main challenge is to determine when conditional intensities λ^i are non-zero, that is on which intervals they are tailored by the exponential interaction functions and not by the positive-part operator.

In Bonnet et al. (2021), the authors solved this challenge for univariate processes by remarking that the conditional intensity is monotone between two event times. Figure 4.1 illustrates that this is not necessarily true for multivariate processes (here, between $T_{(2)}$ and $T_{(3)}$). This constitutes the major difficulty we have to cope with.

4.3.2 Underlying intensity and restart times in the multivariate setting

From now on, let us focus on the exponential model (Hawkes 1971), where each interaction function h_{ij} is then defined as

$$h_{ij}(t) = \alpha_{ij} \mathrm{e}^{-\beta_{ij}t} \,,$$

with $\alpha_{ij} \in \mathbb{R}$ and $\beta_{ij} \in \mathbb{R}^*_+$ for $i, j \in \{1, \ldots, d\}$. For each $i \in \{1, \ldots, d\}$, the underlying intensity function λ^{i*} is defined as in Bonnet et al. (2021) for the univariate case:

$$\lambda^{i\star}(t) = \mu_i + \sum_{j=1}^d \int_0^t h_{ij}(t-s) \, \mathrm{d}N^j(s) \, .$$

This quantity coincides with the conditional intensity λ^i when it is non-zero, and is non-positive otherwise. In particular, we can observe that $\lambda^i(t) = (\lambda^{i\star}(t))^+$ (see Figure 4.1).

As explained in the previous section, the main difficulty of the multivariate exponential setting is the non-monotony of conditional intensities λ^i between two event times. Determining intervals where λ^i is non-zero (that is when $\lambda^{i\star}$ is positive) would require to numerically find the roots of a high-degree polynomial, which is expensive and inexact. To alleviate this problem, we introduce Assumption 2.

Assumption 2. For each $i \in \{1, \ldots, d\}$, there exists $\beta_i \in \mathbb{R}^*_+$ such that $\beta_{ij} = \beta_i$ for all $j \in \{1, \ldots, d\}$.

Remark. This model with constant recovery rates β_i has been studied before in the works of Ogata (1981) in the self-exciting version of a 2-dimensional Hawkes process. Intuitively, this assumption considers the situation where the rate of "dissipation" of any internal or external effect is dependent only on the receiving phenomenon. For instance, for neuronal interactions, each activation from neuron j will have an impact on a connected neuron i dependent on both neurons $(\alpha_{ij})_{ij}$ but the "recovery" time can be assumed to depend only on the connected neuron i $(\beta_i)_i$.

As we will see in Lemma 4.3.1, this assumption enables to recover the monotony of the conditional intensities between two times. It remains now to determine when the underlying intensity λ^{i*} is negative. To do so, we define the restart times in the multivariate framework, to be, for any k and i:



$$T_{(k)}^{i\star} = \min\left(\inf\{t \ge T_{(k)} : \lambda^{i\star}(t) \ge 0\}, T_{(k+1)}\right)$$

Figure 4.2: Illustration of restart times $(T_{(k)}^{\star})_{1 \le k \le 4}$ for each subprocess of a 2-dimensional process associated with event times $(T_{(k)})_{1 \le k \le 4}$.

As exemplified in Figure 4.2, the restart time $T_{(k)}^{i\star}$ (associated to the sub-process *i*) corresponds to the first instant between $T_{(k)}$ and $T_{(k+1)}$ from which $\lambda^{i\star}(t)$ becomes non-negative (or $T_{(k+1)}$ if this instant does not exist). Intuitively, it means that $\lambda^{i}(t) = \lambda^{i\star}(t)$ on $(T_{(k)}^{i\star}, T_{(k+1)})$ and $\lambda^i(t) = 0$ elsewhere on $(T_{(k)}, T_{(k+1)})$. This is formalised in Lemma 4.3.1. In particular, it appears that the restart time $T_{(k)}^{i\star}$ can be expressed as a function of

$$T_{(k)} + \beta_i^{-1} \log \left(\frac{\mu_i - \lambda^{i\star}(T_{(k)})}{\mu_i} \right)$$

which is the single root to the equation $\mu_i + (\lambda^{i*}(T_{(k)}) - \mu_i)e^{-\beta_i(t-T_{(k)})} = 0$ on the interval $[T_{(k)}, +\infty)$ when $\lambda^{i*}(T_{(k)}) < 0$ (see top panel of Figure 4.2). Then, Proposition 4.3.1 gives an explicit formulation of the compensator of each subprocess N^i , which is needed to compute the log-likelihood (see Equation (4.3)). The proofs of these two results are presented respectively in Appendix 4.A and Appendix 4.B.

Lemma 4.3.1. If Assumption 2 is granted, then for each $i \in \{1, ..., d\}$ and any $k \ge 1$:

$$T_{(k)}^{i\star} = \min\left(t_k^{\star}, T_{(k+1)}\right).$$
(4.4)

where

$$t_{k}^{\star} = \left(T_{(k)} + \beta_{i}^{-1} \log\left(\frac{\mu_{i} - \lambda^{i\star}(T_{(k)})}{\mu_{i}}\right) \mathbb{1}_{\{\lambda^{i\star}(T_{(k)}) < 0\}}\right)$$

Furthermore, for all $t \in (T_{(k)}, T_{(k+1)})$,

$$\lambda^{i}(t) = \begin{cases} \lambda^{i\star}(t) > 0 & \text{if } t \in (T_{(k)}^{i\star}, T_{(k+1)}) \\ 0 & \text{otherwise.} \end{cases}$$

Proposition 4.3.1. [Compensator for multivariate exponential kernels] Let us suppose that Assumption 2 is granted. For each $i \in \{1, ..., d\}$ the compensator Λ^i of the process N^i reads, $\forall t \geq 0$:

$$\Lambda^{i}(t) = \begin{cases} \mu_{i}t & \text{if } t < T_{(1)} \\ \mu_{i}T_{(1)} + \sum_{k=1}^{N(t)} J_{k} & \text{if } t \ge T_{(1)} , \end{cases}$$

$$(4.5)$$

where for all integer $k \in \{1, \ldots, N(t)\}$:

$$J_{k} = \mu_{i} \left[\min(t, T_{(k+1)}) - T_{(k)}^{i\star} \right] + \beta_{i}^{-1} \left(\lambda^{i\star}(T_{(k)}) - \mu_{i} \right) \left[e^{-\beta_{i}(T_{(k)}^{i\star} - T_{(k)})} - e^{-\beta_{i}(\min(t, T_{(k+1)}) - T_{(k)})} \right].$$

4.3.3 Identifiability and likelihood computation

As already mentioned in Section 4.2, let $t \ge 0$ and assume that event times $\{T_k^i : 1 \le k \le N_i(t), 1 \le i \le d\}$ are observed in the interval (0, t]. We consider the parametric exponential model for a multivariate Hawkes process of dimension d, defined by

$$\mathcal{P} = \left\{ (\lambda^1, \dots, \lambda^d) \colon \lambda^1 \in \mathcal{P}^1, \dots, \lambda^d \in \mathcal{P}^d \right\} \,,$$

where for each $i \in \{1, \ldots, d\}$, \mathcal{P}^i is the exponential parametric model for the process N^i :

$$\mathcal{P}^{i} = \left\{ \lambda_{\theta_{i}}^{i} = \left(\mu_{i} + \sum_{j=1}^{d} \int_{-\infty}^{t} \alpha_{ij} \mathrm{e}^{-\beta_{i}(t-s)} \, \mathrm{d}N^{j}(s) \right)^{+} : \theta_{i} = (\mu_{i}, \alpha_{i1}, \dots, \alpha_{id}, \beta_{i}) \in \Theta \right\},$$

where $\Theta = \mathbb{R}^*_+ \times \mathbb{R}^d \times \mathbb{R}^*_+$. For a candidate set of intensities $(\lambda^1_{\theta_1}, \ldots, \lambda^d_{\theta_d})$, the underlying intensity functions are denoted $\lambda^{i*}_{\theta_i}$ $(i \in \{1, \ldots, d\})$, the compensators $\Lambda^i_{\theta_i}$ and the restart times $(T^{i*}_{\theta_i,(k)})_k$.

Now, given a realisation $(T_{(k)})_{k>0}$ of a multivariate exponential Hawkes process, Theorem 4.3.1 establishes the correspondence between the conditional intensities and the parameters.

Theorem 4.3.1 (Identifiability). Let $N = (N^1, ..., N^d)$ be a multivariate Hawkes process defined by a set of intensity functions $(\lambda_{\theta_1}^1, ..., \lambda_{\theta_d}^d) \in \mathcal{P}$, for some $\theta = (\theta_1, ..., \theta_d) \in \Theta^d$. Let also $(T_{(k)})_{k>0}$ be a realisation of N and \mathcal{F}_t be the corresponding filtration.

Let us assume that a.s. for every $(i, j) \in \{1, \ldots, d\}^2$, $i \neq j$, there exist an event time τ from process N^j , and an event time $\tau_+ > \tau$ from process N^i , such that:

1. $\lambda_{\theta_i}^i(\tau^-) > 0;$

2. there are only events of process N^j in the interval $[\tau, \tau_+)$.

Then, for any $\theta' \in \Theta^d$,

 $\forall i \in \{1, \dots, d\}, \ \lambda_{\theta_i}^i(t \mid \mathcal{F}_t) = \lambda_{\theta'_i}^i(t \mid \mathcal{F}_t) \ a.e. \qquad \Longleftrightarrow \qquad \theta = \theta' \,.$

The proof is presented in Appendix 4.C. To the best of our knowledge, the only identifiability result for non-linear multivariate Hawkes processes is given by Sulem et al. (2024) but only if interaction functions h_{ij} have a bounded support. Their proof strongly relies on this assumption since they extend to the multivariate case the renewal properties proved by Costa et al. (2020) for non-linear univariate Hawkes processes with a bounded kernel. Their proof also requires some assumptions to ensure that one process is not totally inhibited, which is also a consequence of the assumptions that we propose, as discussed in Section 4.3.4.

As expected, Proposition 4.3.1 makes it possible to compute explicitly the log-likelihood expressed in Equation (4.3) for multivariate exponential Hawkes processes. This is formalised in Corollary 4.3.1 (and proved in Appendix 4.D).

Corollary 4.3.1. Let $i \in \{1, \ldots, d\}$ and $k \ge 2$ an integer. Let us denote

$$S_k^i := T_{(N(T_k^i) - 1)} = T_{(\max\{\ell \in \mathbb{N}^* : T_{(\ell)} < T_k^i\})},$$

the time preceding directly T_k^i , the k^{th} observation of process N^i . Then, for all $\theta \in \Theta$, the log-likelihood of process N^i reads:

$$\ell_t^i(\theta_i) = \log \mu_i + \sum_{k=2}^{N^i(t)} \log \left(\mu_i + (\lambda_{\theta_i}^{i\star}(S_k^i) - \mu_i) e^{-\beta_i(T_k^i - S_k^i)} \right) - \Lambda_{\theta_i}^i(t) , \qquad (4.6)$$

where $\Lambda_{\theta_i}^i$ is given by Equation (4.5) and with convention $\log(x) = -\infty$ for $x \leq 0$.

Algorithm 5 in Appendix 4.E presents the iterative computation of the likelihood using Equation (4.6). In particular, the complexity of the computation is $O(N(t) \times d)$. It is then possible to establish the Maximum Likelihood Estimator, which we will refer to as (MLE). These estimators will be denoted by a tilde: $(\tilde{\mu}_i)_i, (\tilde{\alpha}_{ij})_{ij}, (\tilde{\beta}_i)_i$ and $(\tilde{h}_{ij})_{ij}$.

4.3.4 On identifiability conditions

In the previous section we presented a result on the identifiability of multivariate Hawkes process with inhibition via Theorem 4.3.1. Let us mention that identifiability of parameters μ_i and β_i do not require any assumption. The challenge of the proof lies in identifying parameters α_{ij} , which is achieved thanks to Conditions 1 and 2 of Theorem 4.3.1. Condition 1 allows to control strong inhibition scenarios by ensuring that each subprocess' intensity is positive sufficiently often, and not only at its own event times, while Condition 2 enables to disentangle the contributions of each subprocess. We believe that these assumptions are only sufficient and could be weakened at the cost of a more intricate analysis.

In this section we will discuss this set of conditions by providing two examples of parameters α_{ij} that allow to apply this result along with one counter-example.

Examples for which the conditions are fulfilled

Let us begin with an example of a two-dimensional process. In this situation, as soon as both processes have an infinite amount of events, Conditions 1 and 2 boil down to finding indexes $k \ge 1$ and $k' \ge 1$ such that $\lambda^1(T_k^{2^-}) > 0$ and $\lambda^2(T_{k'}^{1^-}) > 0$. Indeed, it is then enough to consider $(\tau, \tau_+) = (T_k^2, T_{N(T_k^2)+1}^1)$ for (i, j) = (1, 2) and $(\tau, \tau_+) = (T_{k'}^1, T_{N(T_{k'}^1)+1}^2)$ for (i, j) = (2, 1).

Example 1. Let us assume that N is a two-dimensional Hawkes process with the following matrix for parameters α_{ij} :

$$\begin{pmatrix} \alpha_{11} & \alpha_{12} \\ \alpha_{21} & \alpha_{22} \end{pmatrix} = \begin{pmatrix} \alpha_{11} & 0.0 \\ \alpha_{21} & \alpha_{22} \end{pmatrix} ,$$

such as $\alpha_{11} \leq 0$, $\alpha_{21} \geq 0$ and $\alpha_{22} > 0$.

Both processes contain an infinite number of events as process N^1 can be seen as a onedimensional Hawkes process and the second one has a lower-bounded intensity $\lambda^2 > \mu^2$. Now, since $\lambda^2(t) > 0$ for all $t \ge 0$, we have $\lambda^2(T_1^{1-}) > 0$. Then, let us first remark that the event times of process N^1 occur independently of N^2 , as $\alpha_{12} = 0$. So, for any event time T_{ℓ}^1 of N^1 , the restart times can be written as if N^1 was a univariate Hawkes process (see Bonnet et al. (2021)):

$$T_{(\ell)}^{1\star} = T_{\ell}^{1} + \beta_{1}^{-1} \log \left(\frac{\mu_{1} - \lambda^{1\star}(T_{\ell}^{1})}{\mu_{1}} \right) \mathbb{1}_{\{\lambda^{1\star}(T_{\ell}^{1}) < 0\}} \,.$$

For $t > T_{(\ell)}^{1\star}$ small enough, both $\lambda^1(t)$ and $\lambda^2(t)$ are positive, meaning that the next event time can come either from N^1 or from N^2 . If we consider an infinite sequence of event times, we will eventually observe an event T_k^2 of N^2 such that $\lambda^1(T_k^{2-}) > 0$.

This gives a set of Hawkes processes with inhibition that verify the assumptions of Theorem 4.3.1. For higher dimensions, the multiplicity of all possible connections between processes complicates the study of general cases from a theoretical point of view. Example 2 illustrates a case where the conditions are fulfilled by considering identically distributed processes.

Example 2. Let us consider a d-dimensional Hawkes process, as well as $\mu, \alpha_+, \alpha_-, \beta$ such that for any *i* and for any $j \neq i$:

$$\mu_i = \mu > 0, \quad \alpha_{ii} = \alpha_- \le 0,$$

$$\beta_i = \beta > 0, \quad \alpha_{ij} = \alpha_+ \ge 0.$$

As each process has the same parameters for μ_i and β_i along with the exact same interactions, all processes are identically distributed and so in order to verify Conditions 1 and 2 it is enough to verify them for i = 1 and j = 2.

Fulfilling both conditions amounts to verifying that, with non-zero probability, we can find $k \ge 1$ such that:

1.
$$\lambda^1(T_k^{2^-}) > 0;$$

2. $T_{(N(T^2)+1)}$ is an event from process N^1 .

Furthermore, let us notice that as all processes are cross-exciting, if $\lambda^1(T_k^{2^-}) > 0$ for a certain k, then $\lambda^1(t) > 0$ for $t > T_k^2$ small enough, so $T_{(N(T_k^2)+1)}$ can come from process N^1 . It remains now to verify that with non zero probability $\lambda^1(T_k^{2^-}) > 0$ for a certain $k \ge 1$.

But the opposite would entail that almost surely, for all indexes $k \ge 1$, $\lambda^1(T_k^{2^-}) = 0$. Yet, as all processes are identically distributed, this means that, for all $j \ne 2$, $\lambda^j(T_k^{2^-}) = 0$ almost surely. It would then follow that, for every i and for every $j \ne i$, for all k > 1, $\lambda^j(T_k^{i^-}) = 0$ almost surely. Let us consider $T_k^{i_0}$ for a fixed k and i_0 . As each process N^j for $j \ne i$ is then excited with the same parameter α_+ and they are identically distributed, then $\lambda^j(T_k^{i_0})$ are identically distributed and independent conditionally on history $\mathcal{F}_{T_k^{i_0}}$. It follows then that the restart times associated to $T_k^{i_0}$ of each process are independent and identically distributed. Consequently, there is a non-

zero probability that at least two processes regenerate at roughly the same time before another time of process N^{i_0} (as it self-inhibits). Once that for j_0, j_1, λ^{j_0} and λ^{j_1} , are positive, the next event time may come from either process, and so either $\lambda^{j_1}(T^-_{(N(T^{i_0}_k)+1)}) > 0$ with $T_{(N(T^{i_0}_k)+1)}$ from process N^{j_0} , or the inverse, which contradicts the fact that for all i, j and for all k > 1, $\lambda^j(T^i_k) = 0$. So for at least one pair (i, j) and one k, we must have $\lambda^j(T^{i-}_k) > 0$ and so it has to occur for all pairs, in particular (1, 2).

In the next section we present Example 3 of a specific set of parameters for which both conditions are not necessarily met.

Example where the conditions are not fulfilled

Example 3. Let us consider the Hawkes process defined by the following parameters:

$$\begin{pmatrix} \mu_1 \\ \mu_2 \end{pmatrix} = \begin{pmatrix} 10^5 \\ 0.1 \end{pmatrix}, \quad \begin{pmatrix} \alpha_{11} & \alpha_{12} \\ \alpha_{21} & \alpha_{22} \end{pmatrix} = \begin{pmatrix} 1.0 & 1.0 \\ -10^6 & -10^6 \end{pmatrix}, \quad \begin{pmatrix} \beta_1 \\ \beta_2 \end{pmatrix} = \begin{pmatrix} 1.0 \\ 10^{-5} \end{pmatrix}$$

The probability that the first event time $T_{(1)}$ is from process N^1 is $10^5/(10^5 + 0.1) \approx 1$. If the first event is indeed from process N^1 , then process N^2 is strongly inhibited and in that case the restart time $T_{(1)}^{2\star}$ is equal to

$$T_{(1)}^{2\star} = T_1 + 10^5 \log(10^7)$$
,

which is very far from T_1 . As λ^1 is lower-bounded by 10^5 , the next candidate of process N^1 is roughly distributed as an exponential random variable with parameter 10^5 so the next event time is with high probability of process N^1 . If this is the case, process N^2 is further inhibited, and with probability going exponentially quickly to 1, all next event times will come from process N^1 , preventing us from granting Conditions 1 and 2.

4.3.5 Recovering the graph of interactions

The aim of this section is to describe methodologies able to estimate non-null interactions between subprocesses, which boils down to detecting parameters such that $\alpha_{ij} \neq 0$. Recovering interactions has an interest, first, in providing a graphical interpretation of the Hawkes model, as it describes which subprocesses are actually connected within the whole process. Moreover, the graph of interactions can also be used as a reduction dimension tool, for instance if we focus on the dynamic of one single subprocess, the activity of which can be impacted by a sub-network of surrounding processes that we want to identify, as investigated in Bonnet et al. (2022a) for neuronal activity.

Estimating non-null interactions is a challenging topic, which has been particularly studied for linear regression (see for instance Tibshirani (1996)). Regarding Hawkes processes with inhibition, this is even more demanding because of the non-differentiability of the log-likelihood. Therefore, we propose, in the subsequent sections, two post hoc techniques (i.e. after computing the MLE estimator) not related to numerical optimization, and additionally, having the benefit to scale easily to high-dimensional processes.

Thresholding

The first method that will be referred to as $(\text{MLE}-\varepsilon)$ is obtained by adding a thresholding step to the classic Maximum Likelihood Estimation (MLE). This is similar to the cumulative percentage of total variation approach presented in Principal Component Analysis Joliffe (2002, Section 6.1.1). All absolute estimated values $|\tilde{\alpha}_{ij}|$ are arranged in increasing order $(|\tilde{\alpha}_{(k)}|)_{k \in \{1,...,d^2\}}$. We compute then the cumulative sums $s_k = \sum_{l=1}^k |\tilde{\alpha}_{(l)}|$ and write $S := s_{d^2}$ the sum of all absolute estimated values. Then all estimations $\tilde{\alpha}_{(k)}$ such that

$$s_k < \varepsilon S$$
,

are set to zero, for a threshold $\varepsilon \in (0,1)$. Subsequently, all non-null estimations $\tilde{\alpha}_{ij}$ are then re-estimated by maximising the log-likelihood.

The choice of an optimal threshold level ε requires a way of comparing estimations, which is achieved thanks to the goodness-of-fit procedure described in Section 4.3.6, and which is illustrated in Section 4.4.2.

Confidence interval

The second method is applicable when a sample (N_1, \ldots, N_n) of n realisations of a multivariate Hawkes process is available. For every i, j, we average all estimations $\tilde{\alpha}_{ij}$ over the realisations N_1, \ldots, N_n to obtain $\bar{\alpha}_{ij}$ and then we determine a confidence interval around each estimation at a given confidence level $1 - \eta$. Then each estimation for which the confidence interval contains 0 is set to zero. Subsequently, all non-null estimations $\tilde{\alpha}_{ij}$ are re-estimated.

In this work we consider two different confidence intervals.

• (CfE) corresponds to the empirical interval

$$\left[\alpha_{\left(\lfloor \frac{\eta}{2}n \rfloor\right)}, \alpha_{\left(\left\lceil \left(1-\frac{\eta}{2}\right)n \right\rceil\right)}\right] ,$$

where, $(\alpha_{(k)})_{k \in \{1,...,n\}}$ is the sequence of the sorted estimations of α_{ij} , and $\lfloor \cdot \rfloor$ and $\lceil \cdot \rceil$ are respectively the floor and the ceiling functions.

• (CfSt) corresponds to

$$\left[\bar{\alpha}_{ij}-t_{1-\frac{\eta}{2}}s_n,\bar{\alpha}_{ij}+t_{1-\frac{\eta}{2}}s_n\right]\,,$$

where s_n is the empirical standard deviation of the sample and $t_{1-\frac{\eta}{2}}$ is the quantile of level $1-\frac{\eta}{2}$ of the Student distribution with n-1 degrees of freedom. This corresponds to a confidence interval obtained for normally distributed estimators. For Hawkes processes with exclusively exciting interactions, estimations obtained through the MLE procedure are asymptotically normal as proven in Ogata (1978, Theorem 5) and as discussed in Laub
(2014). For processes with inhibiting interactions, asymptotic normality is still an open question but is in all likelihood true. However, one has to be careful when using this estimator, in particular a small number of observations could imply that the asymptotic normality is not achieved. In practice, normality can be tested thanks to a Kolmogorov-Smirnov test.

This method of selection through confidence intervals can be seen as testing the following hypothesis at a confidence level $1 - \eta$ for every i, j:

$$\begin{cases} \mathcal{H}_0 : \alpha_{ij} = 0, \\ \mathcal{H}_1 : \alpha_{ij} \neq 0. \end{cases}$$

We can then compute the corresponding *p*-value for each test and set to zero all parameters for which the null hypothesis is not rejected. As we test d^2 different hypotheses, it is essential to incorporate multiple testing procedures. For this purpose, we choose the Benjamini-Hochberg method, consisting in adapting the rejection threshold of each *p*-value. This method enables to control the false discovery rate (FDR). If we denote V the number of rejected true null hypothesis and R the number of rejected true alternative hypotheses, the FDR is defined as

$$FDR = \mathbb{E}\left[\frac{V}{R+V}\right]$$
.

In other words, we control the expected number of true null hypotheses (i.e. parameter α_{ij} is equal to zero) rejected by our testing method. The B-H procedure considers the ordered *p*-values $(p_{(k)})_{k \in \{1,...,d^2\}}$ and compares each one to the adapted rejection threshold $(1 - \eta)\frac{k}{d^2}$. Then, we determine the largest $K \in \{1, \ldots, d^2\}$ such that $p_{(K)} < (1 - \eta)\frac{K}{d^2}$ and we reject all hypothesis such that $p_{(k)} \leq p_{(K)}$

4.3.6 Goodness-of-fit

As a benefit of our approach, it is possible to perform a goodness-of-fit test for assessing the quality of estimations. This is particularly useful when choosing between several estimations (such as those introduced before), in particular to choose an optimal level of thresholding for the (MLE- ε) method. The closed-form expression of the compensator given in Proposition 4.3.1 enables the use of the Time Change Theorem for inhomogeneous Poisson processes Daley et al. (2003, Proposition 7.4.IV). For any *i*, the sequence of transformed times $(\Lambda^i(T_k^i))_k$ is a realisation of a homogeneous Poisson process with unit-intensity if and only if $(T_k^i)_k$ is a realisation of a point process with intensity λ^i .

We can then define for any $\theta \in \Theta$ the null hypothesis

 $\mathcal{H}_i: (T_k^i)_k$ is a realisation of a point process with intensity $\lambda_{\theta_i}^i$.

The hypothesis is then tested via a Kolmogorov-Smirnov test between the empirical distribution $(\Lambda_{\theta}^{i}(T_{k+1}^{i}) - \Lambda_{\theta}^{i}(T_{k}^{i}))_{k\geq 1}$ and an exponential distribution with parameter 1. We obtain then *d* different tests with *p*-values $(p_{i})_{i\geq 1}$. Using multiple testing approaches can help in determining correctly estimated processes.

Lastly, we can obtain an additional test by considering the entire sequence of times $(T_{(k)})_{k\geq 1}$ and the total intensity λ . We obtain then the null hypothesis

 \mathcal{H}_{tot} : " $(T_{(k)})_k$ is a realisation of a point process with intensity λ_{θ} ",

with corresponding *p*-value p_{tot} . This value is obtained by way of a Kolmogorov-Smirnov test between the empirical distribution of $(\Lambda_{\theta}(T_{(k+1)}) - \Lambda_{\theta}(T_{(k)}))_{k\geq 1}$ and an exponential distribution with parameter 1, this time using the total compensator of the process.

In the forthcoming sections, this testing procedure is applied to several realisations of event times, that are independent of the considered estimator. This enables to assess properly the accuracy of estimations, without knowing the true conditional intensities. This is particularly interesting for real-world data.

Let us mention that the goodness-of-fit procedure is not only an assessment of the overall fit between the model and the observations but it provides also a tool to calibrate the threshold for the (MLE- ε) method. Indeed, for each threshold level ε chosen over a grid, we compute all p-values (one for each subprocess, one for p_{tot}) and we choose the value of ε that maximises the mean p-value. For the other two methods, (CfE) and (CfSt), the model selection procedure is described in Section 4.3.5 and the goodness-of-fit is only performed afterwards in order to establish the quality of the estimations.

4.4 Illustration on synthetic datasets

4.4.1 Simulation procedure

In order to assess the performance of the maximum likelihood estimation method, we simulate different data by using Ogata's thinning method (Ogata 1981). This method consists in defining a piecewise constant function λ^+ such that for any $k \ge 1$ and any $t \in [T_{(k)}, T_{(k+1)}), \lambda^+(t) \ge \lambda(t)$. For this, we define λ^+ for any $t \in [T_{(k)}, T_{(k+1)})$ as

$$\lambda^{+}(t) = \sum_{i=1}^{d} \left(\mu_{i} + \sum_{j=1}^{d} \int_{0}^{t} \alpha_{ij}^{+} \mathrm{e}^{-\beta_{i}(T_{(k)}-s)} \,\mathrm{d}N^{j}(s) \right),$$

which corresponds to considering only the positive interactions.

Four different parameter sets are considered: three sets for 2-dimensional Hawkes processes and a last one for a 10-dimensional process. Table 4.1 presents the parameters used in Dimension 2. All scenarios contain at least one negative interaction ($\alpha_{ij} < 0$). Scenario (1) is a Hawkes process where all parameters are non-null whereas Scenarios (2) and (3) are chosen to study the performance of our methods when estimating null interactions (α_{12} for Scenario (2) and α_{21} for Scenario (3)). All simulations have exactly 5000 event times in total.

Scenario	(1)	(2)	(3)
$\begin{pmatrix} \mu^1 \\ \mu^2 \end{pmatrix}$	$\begin{pmatrix} 0.5\\ 1.0 \end{pmatrix}$	$\begin{pmatrix} 0.7\\ 1.0 \end{pmatrix}$	$\begin{pmatrix} 1.2 \\ 1.0 \end{pmatrix}$
$\begin{pmatrix} \alpha_{11} & \alpha_{12} \\ \alpha_{21} & \alpha_{22} \end{pmatrix}$	$\begin{pmatrix} -1.9 & 3.0 \\ 1.2 & 1.5 \end{pmatrix}$	$\left \begin{array}{cc} 0.2 & 0.0\\ -0.6 & 1.2 \end{array}\right $	$\left \begin{array}{cc} -1.0 & 0.1\\ 0.0 & -0.8 \end{array}\right $
$\begin{array}{c} & \\ & \begin{pmatrix} \beta_1 \\ \beta_2 \end{pmatrix} \end{array}$	$\begin{pmatrix} 5.0 \\ 8.0 \end{pmatrix}$	$\begin{pmatrix} 3.0\\ 2.0 \end{pmatrix}$	$\left \begin{array}{c} \left(\begin{matrix} 0.3 \\ 0.5 \end{matrix} \right) \\ \end{array} \right $

Table 4.1: Parameters for simulations of two-dimensional Hawkes processes.

In order to carry out the hypothesis testing procedure, we simulate a sample of Hawkes

processes independent from the one used for the estimation. Each testing sample contains as many realisations as the estimation sample. All *p*-values presented in the chapter correspond to the average obtained over all realisations.

4.4.2 Proposed methods and comparison to existing procedures

The main focus of this chapter is to assess the performance of the maximum likelihood estimator to correctly detect the interacting functions of our processes without ambiguity, estimators are denoted with a tilde: $(\tilde{\mu}_i)_i, (\tilde{\alpha}_{ij})_{ij}, (\tilde{\beta}_i)_i$ and $(\tilde{h}_{ij})_{ij}$. In this chapter we propose four methods previously introduced in Section 4.3.5:

- (MLE) The estimator obtained by minimising the opposite of the log-likelihood $-\sum_{i=1}^{d} \ell_t^i(\theta)$ (see Equation (4.6)). The log-likelihood is computed via Algorithm 5 and the minimisation is done with the L-BFGS-B method (Byrd et al. 1995).
- (MLE- ε) The estimator obtained by adding a thresholding step to the previous method to determine the non-null estimations. The value of ε is chosen such that it maximises the mean over all *p*-values obtained.
- (CfE) The estimator whose support is obtained through the empirical confidence intervals.
- (CfSt) The estimator using Student distributed intervals after verification of normality of the estimations.

The latter three methods are specially interesting for Scenarios (2) and (3) of the 2-dimensional processes, and also for the 10-dimensional setting.

Remark. Another option considered for $(MLE - \varepsilon)$ is to use instead the values $|\tilde{\alpha}_{ij}/\tilde{\beta}_i|$ for the thresholding. Numerical results slightly differ between the two methods, with the retained method showing better overall p-values.

For an informative assessment of the proposed approaches, we compare their performance to estimation methods from the literature. However, up to our knowledge, there is no other parametric estimation methods designed for inhibiting processes, that is for handling negative values of α_{ij} . As a consequence, we chose to include estimation methods developed for exciting processes, that are nonetheless able to produce negative estimations of α_{ij} . This is the case for three popular approaches described below.

1. (Approx) The first one (Lemonnier et al. 2014) is obtained by approaching the compensator $\Lambda^{i}(t)$ (in each log-likelihood $\ell^{i}_{t}(\theta)$) by

$$\int_0^t \lambda^{i\star}(u) \,\mathrm{d} u \,.$$

In the case where all interactions are positive, this integral is equal to the compensator. The difference is when interactions are negative as this integral takes into account the negative values of the underlying intensity function. The minimisation is done in the same way as for (MLE) using the L-BFGS-B method.

2. The other two methods minimise the least-squares loss approximation defined in Reynaud-Bouret et al. (2014) and Bacry et al. (2020) as:

$$R_t(\theta) = \int_0^t (\lambda_{\theta}(u))^2 \,\mathrm{d}u - \frac{2}{t} \sum_{k=1}^{N(t)} \lambda_{\theta}^{m_k}(T_{(k)}^-) \,,$$

which is an observable approximation of $\|\lambda_{\theta} - \lambda\|_t^2 = \int_0^t (\lambda_{\theta}(u) - \lambda(u))^2 \, du$ up to a constant term. In Bacry et al. (2020), all interactions are assumed to be positive, however the implemented version of this method in the package tick Bacry et al. (2018) allows to retrieve negative values. For this, we consider two different kernel functions from this implementation:

- (Lst-sq) $h_{ij}(t) = \alpha_{ij}\beta_{ij}e^{-\beta_{ij}t}$, where β_{ij} is fixed beforehand by the practitioner. In practice, we fix $\beta_{ij} = \beta_i$ to be consistent with our model (see Assumption 2). The only solver in the implementation that provides negative values is BFGS, which is limited to work with an ℓ_2 -penalty. The grid of values $\{1, 10, \ldots, 10^6\}$ is considered for the regularisation constant. To obtain the best estimation for this method, we choose the constant that minimises the relative squared error over all estimated parameters.
- (Grid-lst-sq) $h_{ij}(t) = \sum_{u=1}^{U} \alpha_{ij}^{u} \beta^{u} e^{-\beta^{u}t}$, with $(\beta^{u})_{u}$ a fixed grid of parameters. In our case, we choose U = d and the grid contains each parameter β_{i} . Intuitively, by applying an ℓ_1 -penalty, this method would be able to retrieve the corresponding parameter β_i for each process. However, in practice, the implementation uses BFGS as optimiser and is limited to work with an ℓ_2 -penalty. As for (Lst-sq), the regularisation parameter is chosen over the grid of values by minimising the relative squared error.

Results on bivariate Hawkes processes

We generate 25 realisations for each parameter set given in Table 4.1 and we estimate the parameters for each individual simulation. We begin this comparison by competing the proposed methods with both (Approx) and (Lst-sq) which are the two methods with the same kernel functions considered in this chapter. Figure 4.3 displays the relative squared errors for each group of parameters (baselines $(\mu_i)_i$, interaction terms $(\alpha_{ij})_{i,j}$ and delay factors $(\beta_i)_i$) by considering vector norms.

First, we observe that delay factors $(\beta_i)_i$ (last column of Figure 4.3) are similarly estimated by all approaches. Let us recall that (Lst-sq) is not included in the comparison of delay factors: since it requires to provide a value for these parameters (they are not estimated), it was given the true values of $(\beta_i)_i$ as input. An alternative offered by tick is to provide a grid of values, but this approach, denoted (Grid-lst-sq), is included in the comparison at the end of the section because of its difference with the exponential model considered here.

Then, regarding the baseline intensities $(\mu_i)_i$ and the interaction factors $(\alpha_{ij})_{ij}$, the proposed methods outperform the two other approaches. In all Scenarios, (MLE- ε), (CfE) and (CfSt) appear to perform almost identically as they retrieve the same supports and from then, the reestimations are the same. In Scenario (2), all estimation methods perform reasonably well. This can be explained by the weak inhibiting effect of the interaction $1 \rightarrow 2$, leaving the intensity almost always positive. The slight difference between (MLE- ε) and the confidence intervals comes from the fact that (MLE- ε) is applied individually to each estimation so for some estimations it does not set any values to zero.

In Scenario (1), the performance of (Approx) and (Lst-sq) is altered, in particular for the $(\tilde{\alpha}_{ij})_{ij}$ estimations, because the inhibiting effect is stronger than in Scenario (2). The major changes appear in Scenario (3), where both (Approx) and (Lst-sq) obtain very high relative errors. More precisely, they fail to explain the interactions between the two processes (see the estimations $(\tilde{\alpha}_{ij})_{ij}$ in the middle column of Figure 4.3), which is compensated by a wrong estimation $(\tilde{\mu}_i)_i$ of baseline intensities. This is not surprising since Scenario (1), and even more Scenario (3), were designed so that the intensity functions are frequently equal to zero, which induces major differences between true and underlying intensities. Since (Approx) and (Lst-sq) are both based on assuming that these two functions are almost equal, the violation of



Figure 4.3: Boxplots of the relative squared error for each group of parameters $((\mu_i)_i, (\alpha_{i,j})_{i,j})$ and $(\beta_i)_i$ for 25 realisations of a two-dimensional Hawkes processes. (Lst-sq) does not appear in the last column because it is provided with the true values of $(\beta_i)_i$. The proposed methods are (MLE), (MLE- ε), (CfE) and (CfSt).

this assumption causes large estimation errors. As expected, the proposed methods, which are developed to handle such inhibiting scenarios, provide accurate estimations.

These results are confirmed by the outcomes of the goodness-of-fit test displayed in Table 4.2. It shows indeed the averaged *p*-values for each scenario using both the true parameters and all four estimations from Figure 4.3 with 25 simulations different from the ones used for estimation. In particular, we can see that our methods obtain high *p*-values, being very close to those obtained using the true parameters. Table 4.2 also highlights when parameters are incorrectly estimated. For instance, in Scenario (1), (Approx) correctly estimate Process 2 but provides less accurate estimations for Process 1 (the *p*-value is almost half the one obtained with the true parameters), which is the one characterised by a self-inhibiting behaviour. In addition, at least one of the proposed methods obtains the highest value for p_{tot} in each scenario, which illustrates the ability of these procedures to reconstruct the complete process *N*. Let us note that the very low *p*-values obtained by (Approx) and (Lst-sq) for Scenario (3) confirm the ability of the goodness-of-fit procedure to detect when the parameter estimations strongly differ from the true parameters.

Lastly, let us investigate the estimations obtained via (Grid-lst-sq), which can be used in practice as a way to estimate the parameters β_i by providing a grid of possible parameters. Let us mention that both of the previous comparisons (boxplots and *p*-values) cannot be done here due to the difference in the number of parameters, but we can compare the methods in terms of reconstructions \tilde{h}_{ij} of the interaction functions h_{ij} . For this purpose, we analyse Figure 4.4, which represents the estimated interaction functions \tilde{h}_{ij} for all methods in Scenario (3). Interestingly, we see that (Grid-lst-sq) performs similarly to (Lst-sq), while the latter is fed with all true values $(\beta_i)_i$ for each interaction. However, we see that (Grid-lst-sq) suffers from the same difficulties

	Scenario (1)			Scenario (2)			Scenario (3)		
<i>p</i> -value	p_1	p_2	p_{tot}	p_1	p_2	p_{tot}	p_1	p_2	p_{tot}
True	0.492	0.438	0.430	0.535	0.468	0.479	0.510	0.623	0.338
MLE	0.440	0.442	0.398	0.483	0.461	0.485	0.549	0.638	0.357
$\begin{array}{c} \text{MLE-}\varepsilon\\ \text{CfE}\\ \text{CfSt} \end{array}$	0.440	0.442	0.398	0.488	0.461	0.491	0.549	0.574	0.327
Approx	0.257	0.442	0.358	0.483	0.452	0.459	0.0	0.007	0.0
Lst-sq	0.154	0.438	0.392	0.534	0.463	0.478	0.0	0.0	0.0

Table 4.2: Average p-values for estimations of two-dimensional Hawkes processes for all scenarios. The values are averaged over 25 simulations. In bold the p-values correspond to a rejected hypothesis at a confidence level of 0.95.

than (Approx) and (Lst-sq), which was expected since it relies on the same unvalid assumption. Let us note that we chose to display the results for Scenario (3) since it highlights the main differences between the compared approaches but the reconstructions for Scenarios (1) and (2) can be found in Appendix 4.F (Figures 4.13 and 4.14).



Figure 4.4: Reconstruction of interaction functions h_{ij} for Scenario (3) of two-dimensional Hawkes processes along with all estimated functions \tilde{h}_{ij} . The real function is plotted in red and 25 estimations are averaged for each method.

A 10-dimensional Hawkes process

A 10-dimensional Hawkes process is simulated based on a set of parameters corresponding to the quantities $(\text{sign}(\alpha_{ij}) \|h_{ij}\|_1)_{ij} = (\alpha_{ij}/\beta_i)_{ij}$ displayed in Figure 4.5. The chosen parameters fulfil



Figure 4.5: Heatmap of real parameters $(\text{sign}(\alpha_{ij}) \|h_{ij}\|_1)_{ij} = (\alpha_{ij}/\beta_i)_{ij}$ for the 10-dimensional simulation.

the existence condition $\|\rho(S^+)\| < 1$.

The corresponding estimations $\tilde{\alpha}_{ij}$ and $\tilde{\beta}_i$ are averaged over 25 realisations and displayed in Figure 4.6. The heatmap representation is convenient for high-dimensional processes as it allows us to see whether the signs of each interaction are well-estimated and whether the nullinteractions are correctly detected.

In this example we decided to keep only (Approx) and (Lst-sq) as comparison methods as these are the ones with the same parameterisation for the kernel functions. Among the four methods considered, (Approx) is the only one that wrongly estimates the sign of some interactions, represented by the black boxes in the second row matrix. (MLE) and (Lst-sq) correctly retrieve the sign of each interactions but are unable to detect the null interactions: this is not surprising since (MLE) does not contain a regularisation step and the (Lst-sq) estimator is implemented with a ℓ_2 -penalty which does not provide a sparse solution. On the one hand, $(MLE-\varepsilon)$ is in this case quite conservative by setting a single value equal to 0 compared to (MLE). On the other hand, both confidence intervals methods improve the number of interactions whose sign is correctly estimated. Interestingly, we see that (CfE) sets more values equal to zero than it should (purple boxes) whereas (CfSt) denotes the opposite effect by not detecting null interactions (orange boxes). Overall, (CfSt) obtains the best results in terms of support recovery and sign estimations by committing only two errors. Table 4.3 summarises the *p*-values for each hypothesis as described in Section 4.3.6. All of the proposed methods obtain overall better pvalues with no particularly low values, which is not the case for (Approx) (see p_4 and p_{tot}) and for (Lst-sq) (see p_8 and p_{10}). Although the *p*-values all exceed 5%, they remain substantially smaller than those obtained with the alternative methods.

Finally, we compare the relative squared errors for each group of parameters (see Figure 4.8).



Figure 4.6: Top row corresponds to the heatmap for each estimation method. Bottom row corresponds to errors made with respect to real parameters from Figure 4.5. A value of 1 (orange) shows an undetected 0 (non-null estimation for $\alpha_{ij} = 0$), a value of -1 (purple) shows a non-null value set to 0 and a value of -2 (black) shows a non-null value whose sign is wrongly estimated. Each of the compared approaches is described in Section 4.4.2.

<i>p</i> -value	p_1	p_2	p_3	p_4	p_5	p_6	p_7	p_8	p_9	p_{10}	p_{tot}
True	0.437	0.608	0.435	0.517	0.534	0.45	0.43	0.47	0.533	0.509	0.464
MLE	0.451	0.619	0.399	0.466	0.506	0.464	0.424	0.386	0.45	0.483	0.434
MLE-0.05	0.454	0.622	0.392	0.466	0.505	0.427	0.418	0.392	0.495	0.482	0.432
CfE	0.424	0.532	0.393	0.528	0.532	0.301	0.444	0.427	0.516	0.505	0.439
CfSt	0.452	0.633	0.375	0.474	0.527	0.462	0.431	0.422	0.488	0.493	0.465
Approx	0.411	0.376	0.475	0.077	0.485	0.411	0.3	0.384	0.285	0.436	0.085
Lst-sq	0.422	0.63	0.344	0.456	0.416	0.439	0.411	0.096	0.579	0.157	0.423

Table 4.3: *p*-values for estimations of a ten-dimensional Hawkes process. The values are averaged over 25 simulations. p_{tot} corresponds to testing whether the estimated intensity function corresponds to a multivariate Hawkes process N as defined in Section 4.3.6.

Similarly to the two-dimensional case, all proposed methods perform significantly better than alternative approaches regarding the estimation of all parameters. In addition, it can be noticed that inaccurate estimations of the parameters α_{ij} tend to deteriorate the estimations of μ_i , which suggests an effect of compensation between these parameters. Regarding the estimator (CfSt), which shows the best averaged performance, it can be remarked that it also exhibits a large variance, in particular when estimating β_i .

Figure 4.7 illustrates for (CfST) the ordered *p*-values for hypothesis $\mathcal{H}_0: \alpha_{ij} = 0$.

This can be explained by this estimator providing a very sparse solution (as seen in Figure 4.6) and therefore taking into account less observations for estimating the coefficients β_i .

An important question for any inference method, especially in a high-dimensional setting, is its computational cost. Table 4.4 shows the average estimation time (over 25 realisations), all times being total estimation time. More precisely, for our 3 model selection methods (MLE- ε), (CfE) and (CfSt), it takes into account the total times, including the first (MLE) estimation in addition to the re-estimation over the support.

Although the difference between (Lst-sq) and all other methods is substantial, let us recall



Figure 4.7: Ordered *p*-values corresponding to support estimation of method (CfSt). Red points correspond to rejected null hypothesis $\mathcal{H}_0: \alpha_{ij} = 0$ and blue points to non-rejected ones.



Figure 4.8: Boxplots of the relative squared error for each group of parameters $((\mu_i)_i, (\alpha_{ij})_{i,j})$ and $(\beta_i)_i$ for 25 realisations of a ten-dimensional Hawkes processes. (Lst-sq) does not appear in the last row because it is provided with the true values of $(\beta_i)_i$. The proposed methods are (MLE), (MLE- ε), (CfE) and (CfSt).

that (Lst-sq) requires to be provided with parameters β_i , which offers two numerical advantages: it does not need to optimise for parameters β_i which are the more difficult parameters to estimate and it includes a pre-computation step (of exponential terms) that accelerates all internal

	MLE	$\text{MLE-}\varepsilon$	CfE	CfSt	Approx	Lst-sq	Grid-lst-sq
Computing time	87.6	178.2	147.2	152.2	51.4	1.32	47.7

Table 4.4: Average computing time in seconds for all estimation methods, averaged over 25 realisations of a ten-dimensional Hawkes process. The time shown for (MLE- ε) is the total time of estimation for 7 different values of ε . Similarly for tick methods, the time is for 7 different levels of penalisation (as done in the estimations of Section 4.4.2). For the (Grid-lst-sq), we provided a search grid for β_i that contains 12 values, including the true one.

computations. In order to provide a fairer comparison, we include the computing time of the (Grid-lst-sq) method which can be considered as an alternative for estimating these parameters when given a search grid for β_i (here with 12 values, including the true parameters). As expected, the computational cost of the MLE method is higher than the alternative approaches (Approx) and (Grid-lst-sq), designed for the linear model. It remains nevertheless that (Lst-sq) and (Grid-lst-sq) are both implemented in a compiled language (C++), which is always faster than an interpreted language such as that used in the proposed package (Python). However, all computation times remain quite reasonable, even for the methods that include a selection model step.

4.4.3 Robustness on misspecified models

In this section, we address the question of the robustness of our estimator regarding the misspecification of the kernel function. More precisely, we generate a two-dimensional Hawkes process with power-law kernels, which are commonly used in the literature (Mishra et al. 2016; Ogata 1988) as an alternative to the exponential kernel for modelling a slower convergence to zero. For all i, j we define the power-law kernel as:

$$h_{ij}(t) = \frac{\alpha_{ij}\beta_{ij}}{(1+\beta_{ij}t)^{1+\gamma}},$$

with $\beta_{ij} > 0$, $\gamma > 0$ and $\alpha_{ij} \in \mathbb{R}$ in order to allow inhibition effects. Let us remark that in the general case each kernel function could be given a different parameter γ_{ij} but here we fix the same parameter for all interactions, which is a similar condition as Assumption 2.

We propose to investigate different scenarios in order to model different behaviours. In all cases, we set the values of α_{ij} in order to have both excitation and inhibition effects and the values of μ_i as follows:

$$\begin{pmatrix} \mu_1 \\ \mu_2 \end{pmatrix} = \begin{pmatrix} 1.0 \\ 1.0 \end{pmatrix}, \quad \begin{pmatrix} \alpha_{11} & \alpha_{12} \\ \alpha_{21} & \alpha_{22} \end{pmatrix} = \begin{pmatrix} 0.1 & 1.5 \\ 1.0 & -0.5 \end{pmatrix}$$

• Scenarios γ : we set

$$\begin{pmatrix} \beta_{11} & \beta_{12} \\ \beta_{21} & \beta_{22} \end{pmatrix} = \begin{pmatrix} 1.0 & 1.1 \\ 1.2 & 1.0 \end{pmatrix}$$

all values being similar in order to be close to Assumption 2. Then we study the effect of parameter γ which controls how the kernel functions decrease to zero. We set

$$\gamma \in \{2.0, 4.0, 6.0, 8.0\}$$
.

• Scenario β : we keep the same values of μ_i , α_{ij} as in Scenarios γ , we fix $\gamma = 4.0$ and we choose very different values of $\beta_{i,j}$:

$$\begin{pmatrix} \beta_{11} & \beta_{12} \\ \beta_{21} & \beta_{22} \end{pmatrix} = \begin{pmatrix} 1.0 & 2.0 \\ 0.1 & 1.0 \end{pmatrix}$$

so that Assumption 2 is violated and therefore the intensity function of process N^2 is frequently non-monotonous between two event times.

We expect that our estimator should adapt better to Scenarios γ (in particular large values of γ would correspond to a fast convergence to zero) than to Scenario β .

Figure 4.9 represents the heatmap for $\operatorname{sign}(\alpha_{ij}) \|h_{ij}\|_1 = \alpha_{ij}/\gamma$ and $\operatorname{sign}(\tilde{\alpha}_{ij}) \|\tilde{h}_{ij}\|_1 = \tilde{\alpha}_{ij}/\tilde{\beta}_i$ as well as whether the type of each interaction is correctly estimated (excitation or inhibition). We first notice that in most cases, our method is robust enough to differentiate between exciting and inhibiting interactions, the only errors concerning parameter α_{11} that is close to zero. For Scenarios γ , we can observe as expected that the bigger differences are obtained for smaller values of γ . Although the signs of the interactions are correctly estimated in Scenario β , we can observe substantial errors regarding the estimations of both α_{21} and α_{22} .



Figure 4.9: Top row corresponds to the heatmap $\operatorname{sign}(\alpha_{ij}) \|h_{ij}\|_1 = \alpha_{ij}/\gamma$. Middle row corresponds to the estimated heatmap $\operatorname{sign}(\tilde{\alpha}_{ij}) \|\tilde{h}_{ij}\|_1 = \tilde{\alpha}_{ij}/\tilde{\beta}_i$. Bottom row corresponds to errors when estimating the nature of the interaction. A black box represents a value α_{ij} whose sign is wrongly estimated.

Table 4.5 shows the average *p*-values associated with the goodness-of-fit measure presented in Section 4.3.6. It confirms that for all Scenarios γ , the *p*-values are smaller for smaller values of γ , all *p*-values remaining greater than 5%. However, the *p*-values associated with Scenario β are always equal to zero, which means that the goodness-of-fit is able to detect that the interactions are not correctly estimated.

To conclude, if the true model is not too far from an exponential model and Assumption 2 holds, our procedure can adapt and provide reasonable estimations. If not, our estimator cannot adjust but we are able to detect incorrect estimations thanks to the goodness-of-fit procedure.

		p_1	p_2	p_{tot}
	$\gamma = 2.0$	0.140	0.275	0.068
	$\gamma = 4.0$	0.484	0.536	0.482
Scenario γ	$\gamma = 6.0$	0.381	0.503	0.408
	$\gamma = 8.0$	0.408	0.506	0.330
Scenar	io β	0.0	0.0	0.0

Table 4.5: Average p-values for estimations of two-dimensional Hawkes processes for all scenarios in the misspecified power-law model. The values are averaged over 25 simulations. In bold the p-values correspond to a rejected hypothesis at a confidence level of 0.95.

4.5 Application on neuronal data

4.5.1 Preprocessing and data description

In this section we present the results obtained by our estimation method applied to a collection of 10 trials consisting in the measurement of spike trains of 223 neurons from the lumbar spinal of a red-eared turtle. This data are first presented in Petersen et al. (2016) and then also analysed in Bonnet et al. (2022a) to study how the activity of a group of neurons impacts the membrane potential's dynamic of another neuron. In particular, recovering the connectivity graph allows to isolate a subnetwork which activity impacts the dynamics of one given neuron. Events were registered for 40 seconds and in order to take into account eventual stationarity we only consider the events that took place on the interval [11, 24] (see Bonnet et al. (2022a) for further details). Among all trials, each neuron recording contains between 54 and 4621 event times. Furthermore, we divide our samples in a training set consisting on all events in half the interval [11, 17.5] and a test set consisting on the remaining window [17.5, 24], in particular each neuron has at least 15 event times in each set. The training sets are used for obtaining the estimations and the test sets for performing the goodness-of-fit tests.

4.5.2 Resampling

As only ten realisations are available, this can obviously limit the performance of both confidence intervals methods. In order to counter this problem, we perform a resampling method obtained as follows:

- 1. We sample 3 realisations at random (N_1, N_2, N_3) , without replacement and by taking the order into account. From now on, we consider that each realisation takes place in the time interval [0, 6.5] (instead of [11, 17.5]).
- 2. We cumulate all 3 realisations by considering that process N_1 takes place in [0, 6.5] then process N_2 in [6.5, 13] and finally N_3 in [13, 19.5]. This creates a single realisation N in the interval [0, 19.5].

This approach is proposed in Reynaud-Bouret et al. (2014, Section 3.4). In our case, we repeat this process 20 times to obtain another sample of realisations. These new sample will be used for both the (MLE) method (presented as (resampled-MLE)) and for both (CfE) and (CfSt).

4.5.3 Goodness-of-fit and multiple testing procedure

Since we are dealing with a high-dimensional setting, it is crucial to account for a multiple testing correction which is performed through the Benjamini-Hochberg procedure as described in Section 4.3.5. In this case, the adapted rejection threshold corresponds to $\frac{0.05k}{d+1}$ and represented in Figure 4.10 by a blue line. This is particularly useful in order to determine the best value of ε for (MLE- ε) as we do not have prior knowledge regarding the sparsity of the neuronal connections.



Figure 4.10: Ordered *p*-values for all hypothesis tests \mathcal{H}_i and \mathcal{H}_{tot} . p_{tot} appears as a cross for each model (if there appears no cross for one given method, it means that the corresponding p_{tot} is equal to zero). The blue curve corresponds to the adapted rejection threshold from the B-H procedure, so all tests whose *p*-value are under the line are rejected.

Figure 4.10 shows the ordered *p*-values for each hypothesis \mathcal{H}_i along with hypothesis \mathcal{H}_{tot} displayed with a bold cross, all total *p*-values also being summarised in Table 4.6.

	MLE	MLE-0.40	MLE-0.60	MLE-0.90	resampled-MLE	CfE	CfSt	Diag
p_{tot}	0.0	0.002	0.004	0.048	0.0	0.053	0.0	0.0

Table 4.6: Values of p_{tot} for each estimation method for the neuronal dataset. In bold appear the *p*-values above the rejection threshold after Benjamini-Hochberg procedure.

We first note that for most methods, the *p*-values p_{tot} associated with the \mathcal{H}_{tot} hypothesis are either equal to 0 or under the rejection threshold. In particular, it is the case for the MLE- ε approach for small values of ε (i.e. weak sparsity scenarios) but as we increase the threshold ε , the *p*-values appear to increase, with the best estimation being achieved for $\varepsilon = 0.90$.

This suggests that the simpler the model the better *p*-values we obtain so we decided to include another approach, named "Diag", consisting in setting all $\alpha_{ij} = 0$ for $i \neq j$. This corresponds to a model where there exists no interaction between neurons and we keep only self-interactions: in other words, each neuron is seen as a univariate Hawkes process with three

parameters $(\mu_i, \alpha_{ii}, \beta_i)$. Although most hypotheses \mathcal{H}_i are not rejected by the method, the total p-value p_{tot} is zero which suggests that although such a model could explain each dimension individually, it is unable to explain the neurons' interactions as a whole interconnected process. Although the total p-values are generally quite low, they remain above the rejection threshold for (MLE-0.90) and (CfE). Let us notice that the high value of the best threshold (0.90) is consistent with the (CfE) estimator that provides an even sparser solution with 4.26% non-null interactions.

Let us also recall that the (CfSt) method relies on the assumption that the MLE estimator is asymptotically normal. Here we have a low number of repetitions (10 trials) in a high-dimensional setting where some neurons are rarely observed which could explain why this estimator does not perform well. Moreover, in this context, a Kolmogorov-Smirnov test of normality is likely to provide high p-values for such small-sized samples even for non-normal distributions.

Finally, the model that best describes the complete process N is (CfE) with the highest value for p_{tot} and with almost all hypotheses, including \mathcal{H}_{tot} , not rejected. This suggests indeed that the estimations provided by (CfE) are the best fit for explaining the entire process as well as each individual subprocess.

4.5.4 Estimation results

Figure 4.11 illustrates the obtained estimation for all parameters for the (CfE) method. Let us recall that the estimation for (CfE) is obtained by using the resampled trials: the support is determined by using the empirical quantiles confidence intervals after an estimation through (MLE) and then all parameters are re-estimated over each trial. A single estimation is obtained by averaging over all trials.

Although the heatmap matrix corresponding to $(\text{sign}(\tilde{\alpha}_{ij}))_{ij}$ contains only 4.26% of nonnull entries, there remain many significant interactions. Interestingly, among them we detect all types of interactions: mutual excitation, mutual inhibition, self-excitation, self-inhibition. This supports the relevance of carefully accounting for inhibition when developing inference procedures.

We also notice that the diagonal contains mostly non-null entries (all but 6), which highlights the major effect of self-interactions, among which some are negative and some are positive. Although it is possible that different neurons actually show different patterns, some being selfexciting and other self-inhibiting, there exists another hypothesis. We might indeed observe a combination of effects from which we cannot distinguish: on the one hand, a self-exciting behaviour and on the other hand, a refractory period following a spike during which a neuron cannot spike again. This could also explain why the order of magnitude of the β_i estimations, which describe the duration until an effect vanishes, is different from a neuron to another. It would be of great interest to propose another modelling that could account for both effects and thereby helping us to provide additional information to support or refute this hypothesis.

Another striking phenomenon is the behaviour of neuron 13, which seems to interact with many other neurons: it contains indeed 69% non-null receiving interactions (row) and 57% non-null giving interactions (column). Further analysis shows that this neuron spikes only in one out of ten trials so that it could indicate an inaccurate estimation. However, the p-value associated with this neuron's subprocess is not rejected by our goodness-of-fit procedure, which suggests that the corresponding estimation is actually accurate. Therefore, this neuron could either play central role among the whole network or be connected to an unobserved neuron with a central role. On the opposite side, some neurons exhibit only a few connections, in particular there is one neuron that only receives interactions without giving, while another one gives without receiving.



Figure 4.11: Heatmap $(sign(\alpha_{ij}))_{ij}$ of (CfE) estimation on 223 neurons.

4.6 Discussion

In this chapter, we proposed a methodology for estimating the parameters of multivariate exponential Hawkes processes with both exciting and inhibiting effects. Our first contribution was to provide a few sufficient conditions to ensure the identifiability of a commonly used model. Then we developed and implemented a maximum likelihood estimator combined with a variable selection procedure that enables to detect the significant interactions inside the whole process. While our framework is more general than the usual linear Hawkes model, there remain two main limitations, the first one being the exponential distribution of the kernels, the second one the assumption that the delay factors β_{ij} only depend on the receiving process N^i . If it is essential to assume a parametric form for the kernel functions in order to maintain the concepts of our approach, it would be of great interest to consider some extensions to account for potential combined effects, as already mentioned in Section 4.5.4. It would be notably relevant to include multi-scale effects or to consider a potential lag between an event time and its actual impact. Regarding the assumption on the delay factors, while it is quite standard, it could be a limitation of our approach when considering heterogeneous phenomena.

Going over this assumption would lead us to explore numerical integration methods and would considerably increase the computational time of the estimation procedure. This is obviously detrimental since, in practice, time sequences are increasingly abundant and large. On the other hand, improving the computational effectiveness of estimation procedures for Hawkes processes is a current direction of research (Bompaire et al. 2018).

Our work focuses on the computational aspects of both maximum likelihood estimation and variable selection. It is of natural interest to provide further theoretical study of the asymptotic behaviour of our estimator, as done for exciting Hawkes processes (Guo et al. 2018). This work is currently under investigation.

Let us also highlight that, because of the physical constraints of the experiment, only a fraction of the neuronal network is observed, which raises the question of interpretability of the estimated interactions. Indeed, the latter do not take into account the interactions with neurons that are outside the observed network. Very recent results tackle the consistency of estimated interactions in a partially observed network (Reynaud-Bouret et al. 2021). A necessary condition to recover interactions in the subnetwork requires in particular to have a large number of interactions within the full network. Regarding the neuronal application, it could be of great interest to further investigate the interpretability of the inferred interactions and connectivity graph in light of the aforementioned work.

4.A Proof of Lemma 4.3.1

In order to prove Lemma 4.3.1, let us first state a preliminary result.

Lemma 4.A.1. If Assumption 2 is granted, then for each $i \in \{1, ..., d\}$ and any $k \ge 1$:

$$\forall t \in [T_{(k)}, T_{(k+1)}), \\ \lambda^{i*}(t) = \mu_i + (\lambda^{i*}(T_{(k)}) - \mu_i) e^{-\beta_i (t - T_{(k)})}.$$

Proof. Let $i \in \{1, \ldots, d\}$. For any $k \ge 1$, the underlying intensity function λ^{i*} in the interval $[T_{(k)}, T_{(k+1)})$ can be written:

$$\lambda^{i\star}(t) = \mu_i + \sum_{j=1}^d \sum_{\ell=1}^{N^j(t)} \alpha_{ij} e^{-\beta_{ij}(t-T_{\ell}^j)} \,.$$

This function is differentiable in the open interval $(T_{(k)}, T_{(k+1)})$ and we obtain:

$$(\lambda^{i\star})'(t) = -\sum_{j=1}^{d} \beta_{ij} \sum_{\ell=1}^{N^{j}(t)} \alpha_{ij} \mathrm{e}^{-\beta_{ij}(t-T_{\ell}^{j})}.$$

By using Assumption 2 that for all $j \in \{1, ..., d\}$, $\beta_{ij} = \beta_i \in \mathbb{R}^*_+$, we obtain the following differential equation:

$$(\lambda^{i\star})'(t) = -\beta_i \left(\lambda^{i\star}(t) - \mu_i\right) ,$$

which by solving on the interval gives:

$$\lambda^{i\star}(t) = \mu_i + \left(\lambda^{i\star}(T_{(k)}) - \mu_i\right) e^{-\beta_i(t - T_{(k)})}.$$

Proof of Lemma 4.3.1. Let $i \in \{1, ..., d\}$ and $k \ge 1$. By Lemma 4.A.1, $\forall t \in [T_{(k)}, T_{(k+1)})$:

$$\lambda^{i\star}(t) = \mu_i + \left(\lambda^{i\star}(T_{(k)}) - \mu_i\right) \mathrm{e}^{-\beta_i(t - T_{(k)})} \,.$$

In particular, the derivative of the underlying intensity function is of opposite sign as $(\lambda^{i*}(T_{(k)}) - \mu_i)$. Let us distinguish two cases, referring to Figure 4.12 for a better understanding:



Figure 4.12: Illustration of three possible scenarios for restart times $T_{(k)}^{i\star}$ depending on the sign of $\lambda^{i\star}(T_{(k)})$ and $\lambda^{i\star}(T_{(k+1)}^-) = \lim_{t \to T_{(k+1)}^-} \lambda^{i\star}(t)$. The dotted line in the last scenario shows the equation $\mu_i + (\lambda^{i\star}(T_{(k)}) - \mu_i) e^{-\beta_i(t-T_{(k)})} = 0$ and the term $T_{(k)} + \beta_i^{-1} \log\left(\frac{\mu_i - \lambda^{i\star}(T_{(k)})}{\mu_i}\right)$ as its only root.

• If $\lambda^{i\star}(T_{(k)}) \ge 0$, then,

$$T_{(k)}^{i\star} = T_{(k)} = \min(T_{(k)}, T_{(k+1)})$$
$$= \min(t_{k}^{\star}, T_{(k+1)}).$$

If $(\lambda^{i*}(T_{(k)})-\mu_i) \geq 0$, then λ^{i*} is decreasing and lower-bounded by μ_i . If $(\lambda^{i*}(T_{(k)})-\mu_i) < 0$ then λ^{i*} is increasing and lower-bounded by zero. In both cases, for any $t \in (T_{(k)}^{i*}, T_{(k+1)})$, $\lambda^{i*}(t) > 0$ and then $\lambda^i(t) = \lambda^{i*}(t)$.

• If $\lambda^{i\star}(T_{(k)}) < 0$, then $(\lambda^{i\star}(T_{(k)}) - \mu_i) < 0$ so $\lambda^{i\star}$ is strictly increasing and by continuity and by Lemma 4.A.1, there exists a unique $t^* > T_{(k)}$ such that $\mu_i + (\lambda^{i\star}(T_{(k)}) - \mu_i) e^{-\beta_i(t^* - T_{(k)})} = 0$.

We obtain:

$$t^{\star} = T_{(k)} + \beta_i^{-1} \log \left(\frac{\mu_i - \lambda^{i^{\star}}(T_{(k)})}{\mu_i} \right).$$

By denoting $\lambda^{i\star}(T^-_{(k+1)}) := \lim_{t \to T^-_{(k+1)}} \lambda^{i\star}(t)$:

- If $\lambda^{i\star}(T_{(k+1)}^-) > 0$, then $t^{\star} < T_{(k+1)}$ by strict increasingness and so by definition $T_{(k)}^{i\star} = t^{\star} = t_k^{\star}$. Lastly, for any $t \in (T_{(k)}, T_{(k+1)})$, if $t \in (T_{(k)}, T_{(k)}^{i\star}]$, $\lambda^{i\star}(t) \leq 0$ and then $\lambda^i(t) = 0$, while if $t \in (T_{(k)}^{i\star}, T_{(k+1)})$, $\lambda^{i\star}(t) > 0$ and then $\lambda^i(t) = \lambda^{i\star}(t)$.

- If $\lambda^{i\star}(T^-_{(k+1)}) \leq 0$, then by strict increasingness $t^{\star} > T_{(k+1)}$ and so $T^{i\star}_{(k)} = T_{(k+1)}$. In this case, for all $t \in (T_{(k)}, T_{(k+1)})$, $\lambda^{i\star}(t) < 0$ so $\lambda^i(t) = 0$. Moreover, $(T^{i\star}_{(k)}, T_{(k+1)}) = \emptyset$ so we never have $\lambda^i(t) = \lambda^{i\star}(t)$.

Combining all scenarios achieves the proof.

4.B Proof of Proposition 4.3.1

Proof. For each $i \in \{1, \ldots, d\}$ and $\forall t \ge 0$, with convention $T_{(0)} = 0$ and $T_{(0)}^{i\star} = 0$:

$$\Lambda^{i}(t) = \int_{0}^{t} \lambda^{i}(u) \, \mathrm{d}u = \sum_{k=0}^{N(t)} \int_{T_{(k)}}^{T_{(k+1)}} \lambda^{i}(u) \mathbb{1}_{u \le t} \, \mathrm{d}u$$
$$= \sum_{k=0}^{N(t)} \int_{T_{(k)}^{i*}}^{T_{(k+1)}} \lambda^{i*}(u) \mathbb{1}_{u \le t} \, \mathrm{d}u \,,$$

where the last equation comes from Lemma 4.3.1. Then, for k = 0:

$$\int_{T_{(0)}^{i_{\star}}}^{T_{(1)}} \lambda^{i_{\star}}(u) \mathbb{1}_{u \leq t} \, \mathrm{d}u = \int_{T_{(0)}^{i_{\star}}}^{T_{(1)}} \mu^{i} \mathbb{1}_{u \leq t} \, \mathrm{d}u$$
$$= \mu_{i} \min(t, T_{(1)}) \,,$$

and for every $k \in \{1, \ldots, N(t)\}$, by Lemma 4.A.1:

$$\begin{split} &\int_{T_{(k)}^{i\star}}^{T_{(k+1)}} \lambda^{i\star}(u) \mathbb{1}_{u \leq t} \, \mathrm{d}u \\ &= \int_{T_{(k)}^{i\star}}^{T_{(k+1)}} \left[\mu_i + \left(\lambda^{i\star}(T_{(k)}) - \mu_i \right) \mathrm{e}^{-\beta_i(t-T_{(k)})} \right] \mathbb{1}_{u \leq t} \, \mathrm{d}u \\ &= \mu_i \left(\min(t, T_{(k+1)}) - T_{(k)}^{i\star} \right) \\ &+ \beta_i^{-1} \left(\lambda^{i\star}(T_{(k)}) - \mu_i \right) \\ &\left(\mathrm{e}^{-\beta_i(T_{(k)}^{i\star} - T_{(k)})} - \mathrm{e}^{-\beta_i(\min(t, T_{(k+1)}) - T_{(k)})} \right) \,. \end{split}$$

4.C Proof of Theorem 4.3.1

Proof. We only need to prove that if $\lambda_{\theta_i}^i(t \mid \mathcal{F}_t) = \lambda_{\theta'_i}^i(t \mid \mathcal{F}_t)$ a.e. for every $i \in \{1, \ldots, d\}$ then $\theta = \theta'$. In this proof, both intensities are considered with respect to the same filtration \mathcal{F}_t so we will omit it from the rest of the proof.

Let $\theta, \theta' \in \Theta$. Let us assume that $\lambda_{\theta_i}^i(t) = \lambda_{\theta'_i}^i(t)$ for all t < T. In order to prove equality between the two parameters we will first prove that $\mu_i = \mu_i'$, then $\beta_i = \beta'_i$ and lastly that $\alpha_{ij} = \alpha'_{ij}$ for every i, j.

• For any *i*, as $\lambda_{\theta_i}^i(t) = \lambda_{\theta'_i}^i(t)$ a.e., then

$$\Lambda^{i}_{\theta_{i}}(T_{(1)}) = \Lambda_{\theta'_{i}}(T_{(1)})$$

$$\Longrightarrow \qquad \mu_{i}T_{(1)} = \mu_{i}'T_{(1)}$$

$$\Longrightarrow \qquad \mu_{i} = \mu_{i}' .$$

• For any i, let us choose $T_{(k+1)}$ such that it is an event of process N^i . As

$$\mathbb{P}(T_{(k+1)} \text{ is an event of } N^i \mid \mathcal{F}_{T_{(k+1)}^-}) = \frac{\lambda_{\theta_i}^i(T_{(k+1)}^-)}{\lambda_{\theta}(T_{(k+1)}^-)},$$

then $\lambda_{\theta_i}^i(T^-_{(k+1)}) > 0$. By definition of $T^{i\star}_{(k),\theta}$,

$$T_{(k),\theta}^{i\star} < T_{(k+1)}$$
 a.s.

Furthermore, $\lambda_{\theta_i}^i(t) = 0$ for $t \in (T_{(k)}, T_{(k),\theta_i}^{i\star})$ and $\lambda_{\theta_i}^i(t) > 0$ for $t \in (T_{(k),\theta_i}^{i\star}, T_{(k+1)})$. Then, as we assumed that $\lambda_{\theta_i}^i(t) = \lambda_{\theta'_i}^i(t)$ a.e., we can conclude that $T_{(k),\theta_i}^{i\star} = T_{(k),\theta'_i}^{i\star}$. By differentiating the intensity functions on the interval $(T_{(k),\theta_i}^{i\star}, T_{(k+1)})$ as in the proof of Lemma 4.3.1 we obtain:

$$\begin{aligned} &(\lambda_{\theta_i}^i)'(t) = (\lambda_{\theta_i'}^i)'(t) \text{ a.e.} \\ &\implies -\beta_i(\lambda_{\theta_i}^i(t) - \mu_i) = -\beta_i'(\lambda_{\theta_i'}^i(t) - \mu_i) \text{ a.e} \\ &\implies (\beta_i - \beta_i')(\lambda_{\theta_i}^i(t) - \mu_i) = 0 \text{ a.s.} \\ &\implies \int_{T_{(k),\theta}}^{T_{(k+1)}} (\beta_i - \beta_i')(\lambda_{\theta_i}^i(t) - \mu_i) \, \mathrm{d}t = 0 \\ &\implies (\beta_i - \beta_i') \int_{T_{(k),\theta}^i}^{T_{(k+1)}} (\lambda_{\theta_i}^i(t) - \mu_i) \, \mathrm{d}t = 0. \end{aligned}$$

Additionnally, $|\lambda_{\theta_i}^i(t) - \mu_i| > 0$ a.s. for all $t \in (T_{(k),\theta_i}^{i*}, T_{(k+1)})$ as $\lambda_{\theta_i}^i$ is monotone and converges to μ_i . Then it follows that the integral is non-zero and so $\beta_i = \beta'_i$.

- Let us prove the equality $\alpha_{ij} = \alpha'_{ij}$. By the assumption made on the event times, for any i, j with $i \neq j$, there exists two event times $\tau < \tau_+$ with τ an event time from process N_j and τ_+ an event time from process N_i such that:
 - 1. $\lambda_{\theta_i}^i(\tau^-) > 0;$
 - 2. there are only events of process N^{j} in the interval $[\tau, \tau_{+})$.

Let τ and τ_+ be two such event times. As a reminder, $T_{(N(\tau)-1)}$ corresponds to the event time before τ and similarly for τ_+ . For $t \in [T_{(N(\tau)-1)}, \tau)$, as $\lambda^i_{\theta_i}(t) = \lambda^i_{\theta'_i}(t)$ a.e. and by Condition 1:

$$\lambda_{\theta_i}^{i\star}(\tau^-) = \lambda_{\theta'_i}^{i\star}(\tau^-) \,.$$

By using the following equalities (proven in the previous points of the proof)

$$\mu_{i} = \mu_{i}', \qquad \beta_{i} = \beta_{i}', T_{(N(\tau)-1),\theta_{i}}^{i\star} = T_{(N(\tau)-1),\theta_{i}'}^{i\star},$$

it follows that

$$\lambda_{\theta_i}^{i\star}(\tau^-) = \lambda_{\theta_i'}^{i\star}(\tau^-)$$

$$\implies \sum_{l=1}^d \alpha_{il} \sum_{T_k^l < \tau} e^{-\beta_i(\tau - T_k^l)} = \sum_{l=1}^d \alpha_{il}' \sum_{T_k^l < \tau} e^{-\beta_i(\tau - T_k^l)}$$

$$\implies \sum_{l=1}^d (\alpha_{il} - \alpha_{il}') A^l = 0, \qquad (4.7)$$

where

$$A^{l} = \sum_{T_{k}^{l} < \tau} \mathrm{e}^{-\beta_{i}(\tau - T_{k}^{l})} \,.$$

We can then write Equation (4.7) by replacing τ by τ_+ as $\lambda^i_{\theta_i}(\tau^-_+) > 0$ because τ_+ is an event time of process N^i . We obtain then

$$\sum_{l=1}^{d} (\alpha_{il} - \alpha'_{il}) B^{l} = 0, \qquad (4.8)$$

where

$$B^l = \sum_{T_k^l < \tau_+} \mathrm{e}^{-\beta_i(\tau_+ - T_k^l)} \, .$$

By definition of event times τ and τ_+ , all events on the interval $[\tau, \tau_+)$ are from process N^j so we obtain for all $l \neq j$,

$$B^l = A^l \mathrm{e}^{-\beta_i (\tau_+ - \tau)} \,.$$

For l = j,

$$B^{j} = A^{j} e^{-\beta_{i}(\tau_{+}-\tau)} + \sum_{\tau \leq T_{k}^{j} < \tau_{+}} e^{-\beta_{i}(\tau_{+}-T_{k}^{j})},$$

where the second term of the right hand side is positive as interval $[\tau, \tau_+)$ contains at least one event, τ , from process N^j . We can rewrite then Equation (4.8):

$$\sum_{l=1}^{d} \left(\alpha_{il} - \alpha_{il}' \right) B^l = 0$$

$$\implies e^{-\beta_i(\tau_+ - \tau)} \sum_{l=1}^d (\alpha_{il} - \alpha'_{il}) A^l + (\alpha_{ij} - \alpha'_{ij}) \sum_{\tau \le T_k^j < \tau_+} e^{-\beta_i(\tau_+ - T_k^j)} = 0.$$

By Equation (4.7), the first term is null and the second sum is non-zero as interval $[\tau, \tau_+)$ contains at least an event from process N^j . It follows that:

$$(\alpha_{ij} - \alpha'_{ij}) \sum_{\tau \le T_k^j < \tau_+} e^{-\beta_i(\tau_+ - T_k^j)} = 0$$
$$\implies (\alpha_{ij} - \alpha'_{ij}) = 0.$$

It follows that for every $j \neq i$, $\alpha_{ij} = \alpha'_{ij}$. It remains to prove that $\alpha_{ii} = \alpha'_{ii}$. For this, let T_2^i be the second event time of process N^i . We can the write the equality

$$\begin{split} \lambda_{\theta_i}^{i\star}(T_2^i) &- \lambda_{\theta_i'}^{i\star}(T_2^i) = 0 \,, \\ \Longrightarrow \, \sum_{l=1}^d \left(\alpha_{il} - \alpha_{il}' \right) \sum_{T_k^l < T_2^i} \mathrm{e}^{-\beta_i (\tau - T_k^l)} = 0 \,, \\ \Longrightarrow \, \left(\alpha_{ii} - \alpha_{ii}' \right) \sum_{T_k^i < T_2^i} \mathrm{e}^{-\beta_i (\tau - T_k^l)} = 0 \,, \end{split}$$

as for $j \neq i$, $\alpha_{ij} = \alpha'_{ij}$. The sum is non-zero as it contains the event T_1^i and so we obtain $\alpha_{ii} = \alpha'_{ii}$.

This achieves the proof.

4.D Proof of Corollary 4.3.1

Proof. For all $i \in \{1, \ldots, d\}, \theta \in \Theta$ and $k \in \mathbb{N}^*$,

$$\begin{split} \log \lambda_{\theta_i}^i(T_k^{i-}) &= -\infty \mathbb{1}_{\lambda_{\theta_i}^i(T_k^{i-})=0} \\ &+ \log \lambda_{\theta_i}^i(T_k^{i-}) \mathbb{1}_{\lambda_{\theta_i}^i(T_k^{i-})>0} \\ &= -\infty \mathbb{1}_{\lambda_{\theta_i}^{i\star}(T_k^{i-})\leq 0} \\ &+ \log \lambda_{\theta_i}^{i\star}(T_k^{i-}) \mathbb{1}_{\lambda_{\theta_i}^{i\star}(T_k^{i-})>0} \\ &= \log \lambda_{\theta_i}^{i\star}(T_k^{i-}) \\ &= \log \lim_{t \to T_k^{i-}} \lambda_{\theta_i}^{i\star}(t). \end{split}$$

Now, for k = 1, $\lambda_{\theta_i}^{i^*}(T_k^{i-}) = \mu_i$, and for $k \ge 2$, let us note $q = N(T_k^i) - 1$. Then,

$$[T_{(q)}, T_{(q+1)}) = [T_{(N(T_k^i)-1)}, T_{(N(T_k^i))}) = [S_k^i, T_k^i),$$

and by Lemma 4.A.1, $\forall t \in [T_{(q)}, T_{(q+1)})$:

$$\lambda_{\theta}^{i\star}(t) = \mu_i + \left(\lambda_{\theta_i}^{i\star}(T_{(q)}) - \mu_i\right) e^{-\beta_i(t - T_{(q)})}$$
$$= \mu_i + \left(\lambda_{\theta_i}^{i\star}(S_k^i) - \mu_i\right) e^{-\beta_i(t - S_k^i)}.$$

Thus,

$$\lim_{t \to T_k^{i^*}} \lambda_{\theta_i}^{i^*}(t) = \lim_{t \to T_{(q)}^{-}} \lambda_{\theta_i}^{i^*}(t)$$
$$= \mu_i + \left(\lambda_{\theta_i}^{i^*}(S_k^i) - \mu_i\right) e^{-\beta_i(T_k^i - S_k^i)}$$

To conclude, by Equation (4.3),

$$\ell_t^i(\theta_i) = \sum_{k=1}^{N^i(t)} \log \lambda_{\theta_i}^i(T_k^{i-}) - \Lambda_{\theta_i}^i(t)$$

= $\log \lambda_{\theta_i}^i(T_1^{i-}) + \sum_{k=2}^{N^i(t)} \log \lambda_{\theta_i}^i(T_k^{i-}) - \Lambda_{\theta_i}^i(t)$
= $\log \mu_i + \sum_{k=2}^{N^i(t)} \log \left(\mu_i + \left(\lambda_{\theta_i}^{i\star}(S_k^i) - \mu_i\right) e^{-\beta_i(T_k^i - S_k^i)}\right)$
- $\Lambda_{\theta_i}^i(t)$.

4.E Algorithm for computing the log-likelihood

This section presents Algorithm 5 for computing the log-likelihood $\ell_t(\theta)$ by leveraging the results from Corollary 4.3.1.

4.F Reconstructed interaction functions for synthetic data

This section presents the reconstruction of interaction functions h_{ij} along with the estimated functions \tilde{h}_{ij} from the two-dimensional Hawkes processes simulations as described in Section 4.4.2. Figure 4.13 and Figure 4.14 correspond respectively to the estimations for Scenarios (1) and (2) from Table 4.1. Algorithm 5: Computation of the log-likelihood $\ell_t(\theta)$ of a multivariate exponential Hawkes process.

$$\begin{split} & \text{Input Parameters } \mu_i, \, \alpha_{ij}, \, \beta_i \text{ for } i, j \in \{1, \ldots, d\}, \text{ list of event times and marks } \\ & (T_{(k)}, m_k)_{k=1:N(t)}; \\ & \text{Initialisation Initialise for all } i, \, \Lambda_k^i = \mu_i T_{(1)}, \, \lambda^{i\star}(T_{(k)}^-) = \mu_i, \, \lambda_k^{i\star} = \mu_i + \alpha_{im_1} \text{ and } \\ & \ell_t(\theta) = \log(\lambda^{m_1\star}(T_{(k)}^-)) - \sum_{i=1}^d \Lambda_k^i; \\ & \text{for } k = 2 \text{ to } N(t) \text{ do} \\ & \text{Compute for all } i, \, T_{(k-1)}^{i\star} = \min\left(T_{(k-1)} + \beta_i^{-1}\log\left(\frac{\mu_i - \lambda_k^{i\star}}{\mu_i}\right)\mathbbm{1}_{\{\lambda_k^{i\star} < 0\}}, T_{(k)}\right); \\ & \text{Compute for all } i, \\ & \Lambda_k^i = \mu_i(T_{(k)} - T_{(k-1)}^{i\star}) + \beta_i^{-1}(\lambda_k^{i\star} - \mu_i)(\mathrm{e}^{-\beta_i(T_{(k-1)}^{i\star} - T_{(k-1)})} - \mathrm{e}^{-\beta_i(T_{(k)} - T_{(k-1)})}); \\ & \text{Compute for all } i, \, \lambda^{i\star}(T_{(k)}^-) = \mu_i + (\lambda_k^{i\star} - \mu_i)\mathrm{e}^{-\beta_i(T_{(k)} - T_{(k-1)})}; \\ & \text{Update } \ell_t(\theta) = \ell_t(\theta) + \log(\lambda^{m_k\star}(T_{(k)}^-)) - \sum_{i=1}^d \Lambda_k^i; \\ & \text{Compute for all } i, \, \lambda_k^{i\star} = \lambda^{i\star}(T_{(k)}^-) + \alpha_{im_k}; \\ & \text{end} \\ \\ & \text{Compute for all } i, \, T_{(N(t))}^{i\star} = \min\left(T_{(N(t))} + \beta_i^{-1}\log\left(\frac{\mu_i - \lambda_k^{i\star}}{\mu_i}\right)\mathbbm{1}_{\{\lambda_k^{i\star} < 0\}}, t\right); \\ & \text{Compute for all } i, \\ & \Lambda_k^i = \left[\mu_i(t - T_{(N(t))}^{i\star}) + \beta_i^{-1}(\lambda_k^{i\star} - \mu_i)(\mathrm{e}^{-\beta_i(T_{(k)}^{i\star} - T_{(N(t))})} - \mathrm{e}^{-\beta_i(t - T_{(N(t))})})\right] \mathbbm{1}_{\{t > T_{(N(t))}^{i\star}\}}; \end{aligned}$$

Update $\ell_t(\theta) = \ell_t(\theta) - \sum_{i=1}^d \Lambda_k^i;$ **return** Log-likelihood $\ell_t(\theta).$



Figure 4.13: Reconstruction of interaction functions h_{ij} for Scenario (1) of two-dimensional Hawkes processes along with all estimated functions \tilde{h}_{ij} . The real function is plotted in red and 25 estimations are averaged for each method.



Figure 4.14: Reconstruction of interaction functions h_{ij} for Scenario (2) of two-dimensional Hawkes processes along with all estimated functions \tilde{h}_{ij} . The real function is plotted in red and 25 estimations are averaged for each method.

Spectral analysis for the inference of noisy Hawkes processes

Classic estimation methods for Hawkes processes rely on the assumption that observed event times are indeed a realisation of a Hawkes process, without considering any potential perturbation of the model. However, in practice, observations are often altered by some noise, the form of which depends on the context. It is then required to model the alteration mechanism in order to infer accurately such a noisy Hawkes process. While several models exist, we consider, in this work, the observations to be the indistinguishable union of event times coming from a Hawkes process and from an independent Poisson process. Since standard inference methods (such as maximum likelihood or Expectation-Maximisation) are either unworkable or numerically prohibitive in this context, we propose an estimation procedure based on the spectral analysis of second order properties of the noisy Hawkes process. Novel results include sufficient conditions for identifiability of the ensuing statistical model with exponential interaction functions for both univariate and bivariate processes. Although we mainly focus on the exponential scenario, other types of kernels are investigated and discussed. A new estimator based on maximising the spectral log-likelihood is then described, and its behaviour is numerically illustrated on synthetic data. Besides being free from knowing the source of each observed time (Hawkes or Poisson process), the proposed estimator is shown to perform accurately in estimating both processes.

Outline of the current chapter

5.1	Introduction	78
	5.1.1 Mathematical setting	79
	5.1.2 Related works	80
5.2	Spectral analysis	81
	5.2.1 The Bartlett spectrum	81
	5.2.2 Superposition of processes and noisy Hawkes process	82
5.3	The univariate noisy Hawkes process	84
	5.3.1 General setting	84
	5.3.2 Exponential model	84
	5.3.3 Beyond the exponential model	86
5.4	The bivariate noisy Hawkes process	86

5.5	Numerical results	89
	5.5.1 Univariate setting \ldots	89
	5.5.2 Bivariate setting \ldots	91
5.6	Discussion	96
5.A	Proof of Proposition 5.2.1	96
$5.\mathrm{B}$	Proof of Proposition 5.3.1	98
5.C	Proof of Proposition 5.3.2	100
5.D	Proof of Proposition 5.3.3	102
5.E	Proof of Proposition 5.4.1	104
5.F	Proof of Proposition 5.4.2, Situations 1 and 2	107
5.G	Proof of Proposition 5.4.2, Situations 3 and 4	109

5.1 Introduction

Hawkes processes, introduced in Hawkes (1971), are a class of point processes that have been originally used to model self-exciting phenomena and more recently other types of past-dependent behaviours. Their fields of applications are wide and include for instance seismology (Ogata 1988; Ogata 1998), neuroscience (Chornoboy et al. 1988; Lambert et al. 2018), criminology (Olinde et al. 2020), finance (Embrechts et al. 2011; Bacry et al. 2015) and biology (Gupta et al. 2018), to mention a few. Consequently, there has been a deep focus on estimation techniques for Hawkes processes. Among them, let us mention maximum likelihood approaches (Ogata 1978; Ozaki 1979; Guo et al. 2018), methods of moments (Da Fonseca et al. 2013), least-squares contrast minimisation (Reynaud-Bouret et al. 2014; Bacry et al. 2020), Expectation-Maximisation (EM) procedures (Lewis et al. 2011), and methods using approximations through autoregressive models (Kirchner 2017).

All of these methods are based on the assumption that the history of the process has been accurately observed, although this information is partial or noised in many contexts, in particular due to measurement or detection errors. Different models, described notably in Lund et al. (2000), have been proposed to handle such errors for spatial point processes, with inference methods based on approximating the likelihood depending on each noise scenario. The first scenario, called *displacement*, is when the event times are observed with a shift. If deconvolution methods to recover the unnoised process are standard approaches for simpler point processes such as Poisson processes (see for instance Antoniadis et al. (2006) and Bonnet et al. (2022b) or the review by Hohage et al. (2016)), the literature for Hawkes processes remains scarce and consists of the work of Trouleau et al. (2019) where the event times are observed with a delay and the work of Deutsch et al. (2020) where the shift follows a Gaussian distribution. The latter also explores the framework where some event times are undetected, which is similar to that studied in Mei et al. (2019). This setting can either be referred to as thinning when the observations are randomly missing, or *censoring* when complete regions are unobserved. The last scenario, called Superposition of ghost points by Lund et al. (2000), is the focus of this chapter and describes situations where additional points are coming from an external point process, in our case a Poisson process. A real-world application that motivated this work comes from spike trains analysis in neuroscience: the membrane potential, which is a continuous signal, is recorded and when it exceeds a certain threshold, one considers that an event time, called a spike, occurred. However, since the original signal is noised, it is possible to detect spikes that do not correspond to real events. Regarding inference of such a noisy process, let us highlight that exact likelihood approaches are intractable due to the unknown origin of each occurrence

(Hawkes process or noise process) while methods based on inferring this missing information, for instance Expectation-Maximisation algorithms, are computationally too demanding.

Inspired by the work of Cheysson et al. (2022) when the event times of a Hawkes process are aggregated, we turn to spectral analysis (Section 5.2.1) to propose a novel estimation procedure that allows to infer the noisy Hawkes process. On the way, we present a general characterisation of the Bartlett spectrum of the superposition of two independent processes (Section 5.2.2) and identifiability results of the statistical model in the univariate (Section 5.3) and the bivariate (Section 5.4) settings. Inference in these two settings is numerically illustrated in Section 5.5. Before starting, we present the mathematical setting (Section 5.1.1) and review the references related to spectral approaches (Section 5.1.2).

5.1.1 Mathematical setting

Let $H = (H_1, \ldots, H_d)$ be a stationary multivariate Hawkes process on \mathbb{R} defined by its conditional intensity functions λ_i^H $(i \in \{1, \ldots, d\})$: for all $t \in \mathbb{R}$,

$$\lambda_i^H(t) = \mu_i + \sum_{j=1}^d \int_{-\infty}^t h_{ij}(t-s) H_j(\mathrm{d}s) = \mu_i + \sum_{j=1}^d \sum_{T_k^{H_j} \le t} h_{ij}(t-T_k^{H_j}), \quad (5.1)$$

where $\mu_i > 0$ is the baseline intensity of process H_i , $h_{ij} : \mathbb{R} \to \mathbb{R}_{\geq 0}$ is the interaction or kernel function describing the effect of process H_j on process H_i , and $(T_k^{H_j})_{k\geq 1}$ denotes the event times of H_j .

By defining the matrix $S = (||h_{ij}||_1)_{1 \le i,j \le d}$, where

$$||h_{ij}||_1 = \int_{-\infty}^{+\infty} h_{ij}(t) \,\mathrm{d}t \,,$$

the stationarity condition of H reduces to controlling the spectral radius of S: $\rho(S) < 1$ (Brémaud et al. 1996).

The goal of this work is to study a noisy version of the Hawkes process where the sequences of event times $(T_k^{H_1})_{k\geq 1}, \ldots, (T_k^{H_d})_{k\geq 1}$ of H are contaminated by the event times from another process P. Since the latter process is aimed at modeling an external noise mechanism due to errors in the detection of event times, it is naturally assumed that each subprocess of H is perturbed uniformly with the same level of noise, such that P is chosen to be a multivariate Poisson process with same intensity across subprocesses. Formally, let $P = (P_1, \ldots, P_d)$ be a multivariate homogeneous Poisson process on \mathbb{R} , supposed to be independent from H, where each univariate process P_i has the same intensity $\lambda_0 > 0$. We note the event times $(T_k^{P_i})_{k\geq 1}$ ($i \in$ $\{1, \ldots, d\}$). We consider then the point process $N = (N_1, \ldots, N_d)$ defined as the superposition of H and P (Definition 5.2.1): the sequence of event times $(T_k^{N_i})_{k\geq 1}$ of N_i ($i \in \{1, \ldots, d\}$) is the ordered union of $(T_k^{H_i})_{k\geq 1}$ and $(T_k^{P_i})_{k\geq 1}$. Throughout this chapter we will refer to N as the noisy Hawkes process, and it will be

Throughout this chapter we will refer to N as the noisy Hawkes process, and it will be assumed that event times of N are observed without knowing their origin (Hawkes or Poisson process). Our goal is to estimate both processes (i.e. the baselines μ_i , the kernels h_{ij} and the shared Poisson intensity λ_0) from the sole observation of $(T_k^{N_i})_{k\geq 1}$, $i \in \{1, \ldots, d\}$.

Inference procedures for point processes often leverage the intensity functions in order to devise maximum likelihood and method of moment estimators (Ogata 1988; Ozaki 1979; Da Fonseca et al. 2013). Here, the process of interest N being a superposition of two independent point processes, the intensity of each subprocess N_i reads (for any integer $i \in \{1, \ldots, d\}$): for all

 $t \in \mathbb{R},$

$$\lambda_i^N(t) = \lambda_0 + \mu_i + \sum_{j=1}^d \int_{-\infty}^t h_{ij}(t-s) H_j(\mathrm{d}s) \,.$$
(5.2)

However, it appears from Equation (5.2) that usual estimators cannot be designed from intensity functions $\lambda_1^N, \ldots, \lambda_d^N$ since they are based on H, which is indistinguishable from P in our setting.

In order to estimate the Hawkes and the Poisson components of N, we propose to leverage the spectral analysis of point processes, recently advocated by Cheysson et al. (2022) for inference of aggregated Hawkes processes. It consists in considering, for a multivariate point process, its matrix-valued spectral density function, denoted $\mathbf{f} : \mathbb{R} \to \mathbb{C}^{d \times d}$, which is related to second-order measures (Bartlett 1963). Given some observed times $(T_k^{N_1})_{k\geq 1}, \ldots, (T_k^{N_d})_{k\geq 1}$, (in a prescribed time window [0, T]), the spectral density is linked to cross-periodograms, defined for all pairs $(i, j) \in \{1, \ldots, d\}^2$ and all $\omega \in \mathbb{R}$ by:

$$I_{ij}^{T}(\omega) = \frac{1}{T} \sum_{k=1}^{N_{i}(T)} \sum_{l=1}^{N_{j}(T)} e^{-2\pi i \omega (T_{k}^{N_{i}} - T_{l}^{N_{j}})}, \qquad (5.3)$$

where $N_i(t) = N_i([0, t))$. Indeed, considering the matrix-valued function $\mathbf{I}^T : \omega \in \mathbb{R} \mapsto (I_{ij}^T(\omega))_{1 \leq i,j \leq d}$, the aforementioned link is to be understood as $\mathbf{I}^T(\omega)$ (for all $\omega \in \mathbb{R}$) being asymptotically distributed according to a complex Wishart distribution with one degree of freedom and scale matrix $\mathbf{f}(\omega)$ (Tuan 1981; Villani et al. 2022). In particular, this implies that $\mathbb{E}[\mathbf{I}^T(\omega)] = \mathbf{f}(\omega)$. Moreover, it is noteworthy that the periodogram $\mathbf{I}^T(\omega)$ can be computed regardless of knowing the source of the event times. This paves the way for estimation.

As it happens, in the scope of statistical inference, a parametric model for the matrix-valued spectral density function is considered:

$$\mathcal{P} = \left\{ \mathbf{f}_{\theta}^{N} : \mathbb{R} \to \mathbb{C}^{d \times d}, \theta = (\mu, \gamma, \lambda_{0}) \in \Theta \right\},\$$

where γ is a parameter that characterises the interaction functions. Then, for $\omega_k = k/T$ ($k \in \{1, 2, ...\}$), it can be shown that $(\mathbf{I}^T(\omega_k))_{k\geq 1}$ are asymptotically independent, leading to the approximate spectral log-likelihood (Brillinger 2012; Düker et al. 2019; Villani et al. 2022):

$$\ell_T(\theta) = -\frac{1}{T} \sum_{k=1}^M \left\{ \log \left(\det \left(\mathbf{f}_{\theta}^N(\omega_k) \right) \right) + \operatorname{tr} \left(\mathbf{f}_{\theta}^N(\omega_k)^{-1} \mathbf{I}^T(\omega_k) \right) \right\},\tag{5.4}$$

where det and tr are respectively the determinant and trace of matrices. Then, the so-called Whittle (or spectral) estimator $\mathbf{f}_{\hat{\theta}}^{N}$ of \mathbf{f} (Whittle 1952) is obtained for $\hat{\theta} \in \Theta$ such that

$$\hat{\theta} \in \arg \max_{\theta \in \Theta} \ \ell_T(\theta) \,.$$

5.1.2 Related works

The spectral analysis of point processes was introduced in Bartlett (1963) and extended to 2-dimensional point processes in Bartlett (1964). Subsequent research works focusing on the theoretical properties of the Bartlett spectrum include Daley (1971), Daley et al. (2003), and Tuan (1981) for temporal settings and Mugglestone et al. (1996), Mugglestone et al. (2001), and Rajala et al. (2023) for spatial contexts.

Despite this, practical applications remain scarcer in the literature. Adamopoulos (1976)

studies earthquake arrivals through the analysis of Hawkes processes and Karavasilis et al. (2007) analyses the cross-correlation of bivariate point processes in the context of muscular stimulation.

In a recent contribution by Chevsson et al. (2022), the authors employ the spectral analysis of point processes to infer an aggregated Hawkes process. More precisely, the observations are assumed to come from a standard Hawkes process but only the counts of events on fixed intervals are available. By leveraging the properties of the Bartlett spectrum, they propose an estimator obtained by means of maximisation of the spectral log-likelihood.

The main advantage of the spectral approach is its obvious ability to handle different kinds of partial observations. That is why we propose an inference procedure for noisy Hawkes processes based on Bartlett's spectral density.

Spectral results concerning the Hawkes model are built on the linearity of the intensity function along with its branching properties. The first spectral analysis of linear Hawkes processes appears in the original paper Hawkes (1971) and was then developed in Daley et al. (2003) in both univariate and multivariate contexts. Explicit expressions of the spectral density of a Hawkes process are available as long as the Fourier transform of the kernel functions are known, which allows to work with a wide array of parametrisations.

In this chapter, we mainly focus on studying identifiability of the statistical model with the classic exponential kernel (and also give insights about identifiability with other kernels). We subsequently derive a parametric inference procedure for estimating the parameters of a process composed of the superposition of a linear Hawkes process and a homogeneous Poisson process.

5.2Spectral analysis

5.2.1The Bartlett spectrum

In this section, we formally introduce the concept of matrix-valued spectral measure $\Gamma : \mathbb{R} \to \mathbb{R}$ $\mathbb{C}^{d \times d}$ for a multivariate stationary point process $N = (N_1, \ldots, N_d)$. This is an extension of the Bartlett spectrum introduced by Bartlett (1963) for the analysis of univariate point processes. Let S be the space of real functions on \mathbb{R} with rapid decay (Daley et al. 2003, Chapter 8.6.1):

$$\mathcal{S} = \left\{ f \in C^{\infty}, \forall k \in \{1, 2, \dots\}, \forall r \in \{1, 2, \dots\}, \sup_{x \in \mathbb{R}} \left| x^r f^{(k)}(x) \right| < \infty \right\},\$$

where C^{∞} is the set of smooth functions from \mathbb{R} to \mathbb{R} and $f^{(k)}$ is the k^{th} derivative of $f \in C^{\infty}$. Then, the Bartlett spectrum of N is the matrix-valued function $\Gamma^N : \omega \in \mathbb{R} \mapsto (\Gamma^N_{ij}(\omega))_{1 \leq i,j \leq d} \in \mathbb{R}$ $\mathbb{C}^{d \times d}$, such that for all $1 \leq i, j \leq d, \Gamma_{ij}^N$ is a measure on \mathbb{R} verifying (Daley et al. 2003, Equation 8.4.13):

$$\forall (\varphi, \psi) \in \mathcal{S} \times \mathcal{S} : \quad \operatorname{cov}\left(\int_{\mathbb{R}} \varphi(x) \, N_i(\mathrm{d}x), \int_{\mathbb{R}} \psi(x) \, N_j(\mathrm{d}x)\right) = \int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \, \Gamma_{ij}^N(\mathrm{d}\omega) \,,$$

where for all $f \in S$, $\tilde{f} : \mathbb{R} \to \mathbb{C}$ denotes the Fourier transform of f:

$$\forall \omega \in \mathbb{R} : \quad \tilde{f}(\omega) = \int_{\mathbb{R}} f(x) \mathrm{e}^{-2\pi \mathrm{i} x \omega} \, \mathrm{d} x \,.$$

If, for all $1 \leq i, j \leq d$, the measure Γ_{ij}^N is absolutely continuous, we can define the matrix-valued spectral density function of N, denoted $\mathbf{f}^N : \mathbb{R} \to \mathbb{C}^{d \times d}$, such that for all $\omega \in \mathbb{R}$, $\mathbf{f}^N(\omega) = (f_{ij}^N(\omega))_{1 \leq i,j \leq d}$ with $\Gamma_{ij}^N(d\omega) = f_{ij}^N(\omega) d\omega$.

From a practical point of view, the spectral density \mathbf{f}^N can be derived from the reduced covariance densities. Let $\mathcal{B}^c_{\mathbb{R}}$ be the collection of all bounded Borel sets on \mathbb{R} and $\ell_{\mathbb{R}} : \mathcal{B}^c_{\mathbb{R}} \to \mathbb{R}_{\geq 0}$ the Lebesgue measure on \mathbb{R} . For all $i \in \{1, \ldots, d\}$, the first moment measure of process N_i is defined as $A \in \mathcal{B}^c_{\mathbb{R}} \mapsto \mathbb{E}[N_i(A)]$ and, by stationarity of the process, it comes:

$$\forall A \in \mathcal{B}^c_{\mathbb{R}} : \quad \mathbb{E}[N_i(A)] = m_i^N \ell_{\mathbb{R}}(A) \,,$$

where $m_i^N = \mathbb{E}[N_i([0,1))]$ is the mean intensity of process N_i . Then, for all $(i,j) \in \{1,\ldots,d\}^2$ the second order moment measure $M_{ij}^N : \mathcal{B}_{\mathbb{R}}^c \times \mathcal{B}_{\mathbb{R}}^c \to \mathbb{R}_{\geq 0}$ is defined by (Daley et al. 2003, Section 5.4):

$$\forall (A,B) \in \mathcal{B}^c_{\mathbb{R}} \times \mathcal{B}^c_{\mathbb{R}} : \quad M^N_{ij}(A,B) = \mathbb{E}[N_i(A)N_j(B)] = \int_{A \times B} M^N_{ij}(\mathrm{d}x,\mathrm{d}y)$$

Now, as the process N is stationary, M_{ij}^N can be decomposed in a product of $\ell_{\mathbb{R}}$ and a so-called reduced measure $\check{M}_{ij}^N : \mathcal{B}_{\mathbb{R}}^c \to \mathbb{R}_{\geq 0}$, such that for any bounded measurable function $g : \mathbb{R}^2 \to \mathbb{R}$ with bounded support (Daley et al. 2003, Equation 8.1.1a):

$$\int_{\mathbb{R}^2} g(x,y) M_{ij}^N(\mathrm{d}x,\mathrm{d}y) = \int_{\mathbb{R}} \int_{\mathbb{R}} g(x,x+u) \ell_{\mathbb{R}}(\mathrm{d}x) \,\breve{M}_{ij}^N(\mathrm{d}u) \,, \tag{5.5}$$

which leads to the definition of the reduced covariance measure $\check{C}_{ij}^N : \mathcal{B}_{\mathbb{R}}^c \to \mathbb{R}_{>0}$:

$$\forall B \in \mathcal{B}^c_{\mathbb{R}} : \quad \check{C}^N_{ij}(B) = \check{M}^N_{ij}(B) - m^N_i m^N_j \ell_{\mathbb{R}}(B) \,. \tag{5.6}$$

Since the right-hand side is the difference of two positive, positive-definite measures (Daley et al. 2003, Section 8.6), we can define the Fourier transform of \check{C}_{ij} as the difference of their Fourier transforms (see for example (Pinsky 2008, Equation 5.2.1) for the Fourier transform of a measure). The resulting quantity comes out to correspond exactly to the spectral density function f_{ij}^N :

$$\forall \omega \in \mathbb{R} : \quad f_{ij}^N(\omega) = \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^N(\mathrm{d}x) - m_i^N m_j^N \delta(\omega) \,, \tag{5.7}$$

where δ is the Dirac delta function.

5.2.2 Superposition of processes and noisy Hawkes process

The model we study considers the superposition of two point processes that we define as follows.

Definition 5.2.1 (Superposition of processes). Let X and Y be two independent and stationary multivariate point processes with same dimension d. The superposition of X and Y, denoted N = X + Y, is the stationary multivariate point process defined for any integer $1 \le i \le d$ as:

$$\forall A \in \mathcal{B}^c_{\mathbb{R}} : \quad N_i(A) = X_i(A) + Y_i(A).$$

It comes from the definition that if X and Y have respectively event times $(T_k^{X_1})_{k\geq 1}, \ldots, (T_k^{X_d})_{k\geq 1}$ and $(T_k^{Y_1})_{k\geq 1}, \ldots, (T_k^{Y_d})_{k\geq 1}$, the sequences of event times of N = X + Y are the ordered unions of $(T_k^{X_1})_{k\geq 1}$ and $(T_k^{Y_1})_{k\geq 1}$ up to $(T_k^{X_d})_{k\geq 1}$ and $(T_k^{Y_d})_{k\geq 1}$. In addition, Proposition 5.2.1 below states that the spectral density of N can be obtained easily from those of X and Y.

Proposition 5.2.1. Let X and Y be two independent and stationary multivariate point processes with same dimension d, admitting respective spectral densities \mathbf{f}^X and \mathbf{f}^Y .

Then N = X + Y admits a matrix-valued spectral density function \mathbf{f}^N and

$$\mathbf{f}^N = \mathbf{f}^X + \mathbf{f}^Y \,. \tag{5.8}$$

Proof. The proof is given in Appendix 5.A.

We are now ready to define the noisy Hawkes model, which is the superposition of a Hawkes process and a homogeneous Poisson process.

Definition 5.2.2 (Noisy Hawkes process). Let H be a multivariate Hawkes process and $P = (P_1, \ldots P_d)$ be a multivariate homogenous Poisson process independent from H and with common intensity $\lambda_0 > 0$ (i.e. for any integer $1 \le i \le d$, P_i is a univariate Poisson process with constant intensity λ_0). The superposition N = H + P is called a noisy Hawkes process.

Let us remark that, since the process P is aimed at modeling some kind of background noise, it is naturally assumed that subprocesses of P share the same intensity λ_0 . However, if identifiability results from the forthcoming sections are specific to this assumption, the presentation done in this section can be trivially extended to a multivariate Poisson process with different intensities.

Now, let us recall that we aim at analysing the process N = H + P through an observation of event times $(T_k^{N_1})_{k\geq 1}, \ldots, (T_k^{N_d})_{k\geq 1}$ with the inability of distinguishing between the times of H and P. To do so from a statistical estimation perspective, we leverage the spectral loglikelihood (Equation (5.4)), which makes use of the spectral density of N. The latter can be computed thanks to Proposition 5.2.1 and Example 8.3(c) from Daley et al. (2003). Indeed, let $\tilde{h}: \mathbb{R} \to \mathbb{C}^{d \times d}$ be the matrix-valued Fourier transform of the interaction functions:

$$\forall w \in \mathbb{R}: \quad \tilde{h}(\omega) = \left(\tilde{h}_{ij}(\omega)\right)_{1 \le i,j \le d}$$

under stationarity conditions, the spectral density \mathbf{f}^H of the Hawkes process H (defined as in Equation (5.1)) is (Daley et al. 2003, Equation (8.3.11)):

$$\forall w \in \mathbb{R}: \quad \mathbf{f}^{H}(\omega) = \left(I_{d} - \tilde{h}(\omega)\right)^{-1} \operatorname{diag}\left(m^{H}\right) \left(I_{d} - \tilde{h}(-\omega)^{T}\right)^{-1}, \quad (5.9)$$

where I_d is the identity matrix of dimension d and diag (m^H) is the diagonal matrix formed by the vector of the mean intensities:

$$\begin{pmatrix} m_1^H \\ \vdots \\ m_d^H \end{pmatrix} = \left(I_d - \tilde{h}(0) \right)^{-1} \begin{pmatrix} \mu_1 \\ \vdots \\ \mu_d \end{pmatrix} .$$

Moreover, since the homogeneous Poisson process P (with common intensity λ_0) is a Hawkes process with null interactions, its spectral density \mathbf{f}^P results easily from Equation (5.9):

$$\forall \omega \in \mathbb{R} : \mathbf{f}^P(\omega) = \lambda_0 I_d \,,$$

which leads to the spectral density of N:

$$\forall w \in \mathbb{R}: \quad \mathbf{f}^{N}(\omega) = \mathbf{f}^{H}(\omega) + \lambda_{0} I_{d}, \qquad (5.10)$$

with \mathbf{f}^H expressed in Equation (5.9). Given this result, the inference procedure consists in maximising the spectral log-likelihood described in Equation (5.4).

In the forthcoming sections, we focus on this statistical estimation problem in low-dimensional settings (d = 1 and d = 2) and provide identifiability results when interactions are exponential.

5.3 The univariate noisy Hawkes process

5.3.1 General setting

Let us start with univariate processes (d = 1). In this case, Equation (5.10) simplifies, as stated in Corollary 5.3.1.

Corollary 5.3.1. Let N be a noisy Hawkes process defined by the superposition of a stationary Hawkes process H (with baseline intensity $\mu > 0$ and kernel function $h : \mathbb{R} \to \mathbb{R}_{\geq 0}$) and an independent homogeneous Poisson process P (with constant intensity $\lambda_0 > 0$). Then the spectral density f^N of N reads:

$$\forall \omega \in \mathbb{R}, \quad f^{N}(\omega) = \frac{\mu}{\left(1 - \|h\|_{1}\right) \left|1 - \tilde{h}(\omega)\right|^{2}} + \lambda_{0}.$$

Proof. This is straightforward from Equations (5.9) and (5.10). See also Daley et al. (2003, Example 8.2(e)) for the spectral density of a univariate exciting Hawkes process.

The estimation procedure also simplifies since the periodogram of N and the spectral loglikelihood (see Equations (5.3) and (5.4)) respectively read:

$$\forall \omega \in \mathbb{R}, \quad I^T(\omega) = \frac{1}{T} \left| \sum_{k=1}^{N(T)} e^{-2\pi i \omega T_k^N} \right|^2,$$

where $(T_k^N)_{k\geq 1}$ is the sequence of event times of the noisy Hawkes process N, and:

$$\forall \theta \in \Theta, \quad \ell_T(\theta) = -\frac{1}{T} \sum_{k=1}^M \left(\log\left(f_\theta^N(\omega_k)\right) + \frac{I^T(\omega_k)}{f_\theta^N(\omega_k)} \right). \tag{5.11}$$

As explained in Section 5.1.1, the Whittle estimator $\hat{\theta}$ is then obtained by maximising the function ℓ_T . In the next section, the exponential parametric model \mathcal{Q} for f^N is described and analysed.

5.3.2 Exponential model

Let us consider the classic exponential kernel for the Hawkes process H:

$$\forall t \in \mathbb{R}: \quad h(t) = \alpha \beta e^{-\beta t} \mathbb{1}_{t>0}, \qquad (5.12)$$

with $0 < \alpha < 1$ and $\beta > 0$. The kernel parameter is thus $\gamma = (\alpha, \beta)$ and the statistical model for a univariate noisy Hawkes process becomes:

$$\mathcal{Q} = \left\{ f_{\theta}^{N} : \mathbb{R} \to \mathbb{C}, \theta = (\mu, \alpha, \beta, \lambda_{0}) \in \mathbb{R}_{>0} \times (0, 1) \times \mathbb{R}_{>0} \times \mathbb{R}_{>0} \right\} \,.$$

The exponential kernel parametrisation has been widely studied from an inference point of view (see for instance Ozaki (1979) and Bacry et al. (2016)). In particular, its Fourier transform

is:

$$\forall \omega \in \mathbb{R} : \quad \tilde{h}(\omega) = \frac{\alpha \beta}{\beta + 2\pi i \omega} , \qquad (5.13)$$

and the spectral density f^H of a univariate Hawkes process with baseline intensity $\mu > 0$ and exponential kernel is (Hawkes 1971):

$$\forall \omega \in \mathbb{R}: \quad f^H(\omega) = m^H \left(1 + \frac{\beta^2 \alpha (2 - \alpha)}{\beta^2 (1 - \alpha)^2 + 4\pi^2 \omega^2} \right), \quad \text{where } m^H = \frac{\mu}{1 - \alpha}.$$

It results that the spectral density $f_{\theta}^{N} \in \mathcal{Q}$ of a univariate noisy Hawkes process is:

$$\forall \omega \in \mathbb{R}: \quad f_{\theta}^{N}(\omega) = \left(\frac{\mu}{1-\alpha}\beta^{2}\alpha(2-\alpha)\right)\frac{1}{\beta^{2}(1-\alpha)^{2}+4\pi^{2}\omega^{2}} + \left(\frac{\mu}{1-\alpha}+\lambda_{0}\right). \tag{5.14}$$

Now, we should be ready for implementing the inference procedure based on maximising the spectral log-likelihood as expressed in Equation (5.11). However, it appears that the model Q, as currently defined, is not identifiable (see Proposition 5.3.1). Hopefully, this model becomes identifiable when restricted to only three parameters, thus allowing the practicability of the proposed estimation method, as numerically illustrated in Section 5.5.1.

Proposition 5.3.1 (Identifiability in the univariate setting). The model Q is not identifiable. In particular, for any admissible parameter $\theta = (\mu, \alpha, \beta, \lambda_0)$ there exists an infinite number of admissible parameters θ' such that $f_{\theta}^N = f_{\theta'}^N$.

However, the four collections of models defined below are identifiable:

1. for all $\mu^{\circ} > 0$, $\mathcal{Q}_{\mu} = \left\{ f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} \in \mathcal{Q}, \mu = \mu^{\circ} \right\};$ 2. for all $\alpha^{\circ} \in (0,1), \ \mathcal{Q}_{\alpha} = \left\{ f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} \in \mathcal{Q}, \alpha = \alpha^{\circ} \right\};$ 3. for all $\beta^{\circ} > 0, \ \mathcal{Q}_{\beta} = \left\{ f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} \in \mathcal{Q}, \beta = \beta^{\circ} \right\};$ 4. for all $\lambda^{\circ}_{0} > 0, \ \mathcal{Q}_{\lambda_{0}} = \left\{ f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} \in \mathcal{Q}, \lambda_{0} = \lambda^{\circ}_{0} \right\}.$ *Proof.* The proof is given in Appendix 5.B. \Box

The previous discussion raises the question of whether this non-identifiability also extends to the distribution of the noisy Hawkes process. It turns out that, in the exponential case, Markov properties of the intensity function λ^H of the underlying Hawkes process help ensuring identifiability. Indeed, from the definition of the Hawkes process H and the stationarity of the Poisson process P, $(\lambda^N(t))_{t\geq 0}$ is a Markov process: it decreases with rate $\beta(\lambda^H(t) - \mu) =$ $\beta(\lambda^N(t) - \mu - \lambda_0)$, and the jumps occurring from the Hawkes process, with rate $\lambda^H(t) = \lambda^N(t) - \lambda_0$, are of size $\alpha\beta$, while the jumps occurring from the Poisson component, with rate λ_0 , have no impact on the intensity of the process. Then $(\lambda^N(t), N(t))_{t\geq 0}$ is also a Markov process, more specifically a piecewise deterministic Markov process (Davis 1984). This allows us to use results from Dassios et al. (2011) on the distribution of exponential Hawkes processes to show that the distribution of the exponential noisy Hawkes process is identifiable.

Proposition 5.3.2. Let $(\mu, \alpha, \beta, \lambda_0)$ and $(\mu', \alpha', \beta', \lambda'_0)$ be two admissible 4-tuples for the exponential noisy Hawkes model Q, and N and N' respectively defined by these two tuples (Equation (5.2) with kernel (5.12)). Then, if N and N' have same distribution and $\lambda^N(0)$ (respec-

tively $\lambda^{N'}(0)$ is distributed according to the stationary distribution of $(\lambda^{N}(t))_{t\geq 0}$ (respectively $(\lambda^{N'}(t))_{t\geq 0}$), then $(\mu, \alpha, \beta, \lambda_0) = (\mu', \alpha', \beta', \lambda'_0)$.

Proof. See Appendix 5.C.

A consequence of this result is that non-identifiability of our proposed method in the exponential case is a shortcoming of the spectral approach itself rather than an underlying property of the noisy Hawkes process, presumably stemming from the fact that the spectral density only encodes the second order moments of the process. In the forthcoming section, we briefly investigate whether this issue arises when considering other reproduction kernels.

5.3.3 Beyond the exponential model

Let N be a noisy Hawkes process defined by the superposition of a stationary Hawkes process with baseline intensity μ and kernel function αh , with $\alpha \in (0,1)$, $h : \mathbb{R} \to \mathbb{R}_{\geq 0}$ and $\|h\|_1 = 1$, and a homogeneous Poisson process with intensity λ_0 . Per Corollary 5.3.1, its spectral density is given by

$$\forall \omega \in \mathbb{R} : \quad f^N(\omega) = \frac{\mu}{\left(1 - \alpha\right) \left|1 - \alpha \tilde{h}(\omega)\right|^2} + \lambda_0 \,.$$

While it may be difficult to show that a model is identifiable from the spectral density expression, it may prove fruitful to look at its Taylor expansion around 0, and analyse the Taylor coefficients. For example, considering the uniform kernel and the corresponding Taylor expansion of the spectral density up to order 2, we get the following proposition.

Proposition 5.3.3. Let us consider a rectangle interaction function

$$h: t \in \mathbb{R} \mapsto \phi^{-1} \mathbb{1}_{0 \le t \le \phi},$$

for some kernel parameter $\phi > 0$, and the corresponding statistical model for a univariate noisy Hawkes process:

$$\mathcal{R} = \left\{ f_{\theta}^{N} : \mathbb{R} \to \mathbb{C}, \theta = (\mu, \alpha, \phi, \lambda_{0}) \in \mathbb{R}_{>0} \times (0, 1) \times \mathbb{R}_{>0} \times \mathbb{R}_{>0} \right\} .$$

Then \mathcal{R} is identifiable.

Proof. See Appendix 5.D.

This last proposition shows that non-identifiability of the spectral approach for the noisy exponential Hawkes process is more a consequence of the choice of the reproduction function h rather than a general shortcoming of the spectral approach. It is unexpected that the exponential reproduction function, usually chosen because the Markov properties for the resulting Hawkes intensity simplify calculations (Ozaki 1979; Da Fonseca et al. 2013; Duarte et al. 2019), seems to be here the main culprit of non-identifiability for our proposed spectral approach. Still, we will show how, by imposing some constraints on the modelling of multivariate noisy Hawkes processes, we are able to ensure identifiability of the model even in this case.

5.4 The bivariate noisy Hawkes process

This section addresses bivariate noisy Hawkes processes (d = 2). More precisely, for such a process N = H + P, where H is a bivariate Hawkes process (see Equation (5.1)) and P a Poisson

process with shared intensity λ_0 , Corollary 5.4.1 gives the closed-form expression of the spectral density \mathbf{f}^N .

Corollary 5.4.1. Let $N = (N_1, N_2)$ be a bivariate noisy Hawkes process defined by the superposition of a stationary Hawkes process $H = (H_1, H_2)$ (with baseline intensities $\mu_1 > 0$ and $\mu_2 > 0$, and kernel functions $h_{ij} : \mathbb{R} \to \mathbb{R}_{\geq 0}, 1 \leq i, j \leq 2$) and an independent homogeneous Poisson process $P = (P_1, P_2)$ (with same constant intensity $\lambda_0 > 0$). Then the spectral density \mathbf{f}^N of N reads:

$$\forall \omega \in \mathbb{R}, \quad \mathbf{f}^{N}(\omega) = \begin{pmatrix} f_{11}^{H}(\omega) + \lambda_{0} & f_{12}^{H}(\omega) \\ f_{21}^{H}(\omega) & f_{22}^{H}(\omega) + \lambda_{0} \end{pmatrix},$$

where

$$\begin{cases} f_{11}^{H}(\omega) = \frac{m_{1}^{H} \left| 1 - \tilde{h}_{22}(\omega) \right|^{2} + m_{2}^{H} \left| \tilde{h}_{12}(\omega) \right|^{2}}{\left| \left(1 - \tilde{h}_{11}(\omega) \right) \left(1 - \tilde{h}_{22}(\omega) \right) - \tilde{h}_{12}(\omega) \tilde{h}_{21}(\omega) \right|^{2}} \\ f_{12}^{H}(\omega) = \frac{m_{1}^{H} \left(1 - \tilde{h}_{22}(\omega) \right) \tilde{h}_{21}(-\omega) + m_{2}^{H} \left(1 - \tilde{h}_{11}(-\omega) \right) \tilde{h}_{12}(\omega)}{\left| \left(1 - \tilde{h}_{11}(\omega) \right) \left(1 - \tilde{h}_{22}(\omega) \right) - \tilde{h}_{12}(\omega) \tilde{h}_{21}(\omega) \right|^{2}} \end{cases}$$

and

$$m_1^H = \frac{\mu_1 \left(1 - \|h_{22}\|_1 \right) + \mu_2 \|h_{12}\|_1}{\left(1 - \|h_{11}\|_1 \right) \left(1 - \|h_{22}\|_1 \right) - \|h_{12}\|_1 \|h_{21}\|_1},$$

and f_{22}^H , f_{21}^H and m_2^H are obtained by symmetry of all indices.

Proof. This is straightforward from Equations (5.9) and (5.10).

Then, the estimation procedure is exactly that described in Section 5.1.1, which is based on computing the cross periodogram \mathbf{I}^T and on maximising the spectral log-likelihood ℓ_T (see Equations (5.3) and (5.4)). Now, similarly to the univariate case detailed in Section 5.3.2, we consider exponential interaction functions, i.e. for $1 \leq i, j \leq 2$:

$$\forall t \in \mathbb{R}: \quad h_{ij}(t) = \alpha_{ij}\beta_i \mathrm{e}^{-\beta_i t} \mathbb{1}_{t \ge 0},$$

with $\alpha_{ij} \ge 0$ and $\beta_i > 0$. The kernel parameter is thus $\gamma = (\alpha, \beta)$, where $\alpha \in \mathbb{R}^{2 \times 2}_{\ge 0}$ and $\beta \in \mathbb{R}^2_{>0}$ and the statistical model for a bivariate noisy Hawkes process becomes:

$$\mathcal{Q}_{\Lambda} = \left\{ \mathbf{f}_{\theta}^{N} : \mathbb{R} \to \mathbb{C}^{2 \times 2}, \theta = (\mu, \alpha, \beta, \lambda_{0}) \in \mathbb{R}^{2}_{>0} \times \Lambda \times \mathbb{R}^{2}_{>0} \times \mathbb{R}_{>0}, \beta \in \Omega_{\alpha} \right\} ,$$

where $\Lambda \subset \{\alpha \in \mathbb{R}^{2 \times 2}_{\geq 0} : \rho(\alpha) < 1\}$ is subset of matrices α that will be specified later, and for all $\alpha \in \mathbb{R}^{2 \times 2}_{\geq 0}$,

$$\Omega_{\alpha} = \left\{ \beta \in \mathbb{R}^2_{>0}, \beta_1 = 1 \text{ if } \alpha_{11} = \alpha_{12} = 0, \beta_2 = 1 \text{ if } \alpha_{21} = \alpha_{22} = 0 \right\}$$

is a subset of admissible values for β . The definition of Ω_{α} takes into account that when a row, say the first one, of the interaction matrix α is null, then the corresponding kernels verify $h_{11} = 0$ and $h_{12} = 0$ independently of the value of β_1 . Thus, identifiability for the parameter β_1 is hopeless, which justifies that we get rid of it (by fixing it to an arbitrary value) from the model.

Remark. Different versions of the multivariate exponential model exist. A first convention assumes that there is a unique $\beta \in \mathbb{R}_{>0}$ shared by all kernel functions (Chevallier et al. 2019; Bacry et al. 2020). A second and less restrictive option, which is that we opt for in this work, assumes that the recovery rate $\beta_i \in \mathbb{R}_{>0}$ for each subprocess N_i $(1 \le i \le d)$ is shared among received interactions (Bonnet et al. 2023). These choices allow for simplified derivations of estimators in the time domain and in the frequency domain, as shown below.

The aim of this section is to study identifiability of model \mathcal{Q}_{Λ} . A broad analysis (i.e. for $\Lambda = \{\alpha \in \mathbb{R}_{\geq 0}^{2 \times 2} : \rho(\alpha) < 1\}$) being out of reach for complexity reasons, we exhibit some situations (i.e. subsets Λ) for which non-identifiability (Proposition 5.4.1) or identifiability (Proposition 5.4.2) can be proved.

Proposition 5.4.1 (Non-identifiability in the bivariate setting). The model Q_{Λ} is not identifiable in the three situations:

- 1. $\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0 \\ 0 & \alpha_{22} \end{pmatrix}, 0 \le \alpha_{11}, \alpha_{22} < 1 \right\}, \text{ that is for diagonal matrices } \alpha \text{ (with possibly null entries).}$
- 2. $\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & \alpha_{12} \\ 0 & 0 \end{pmatrix}, 0 < \alpha_{11} < 1, \alpha_{12} > 0 \right\}, \text{ that is for matrices with positive entries in the first row and null entries in the second.}$
- 3. $\Lambda = \left\{ \begin{pmatrix} 0 & 0 \\ \alpha_{21} & \alpha_{22} \end{pmatrix}, \alpha_{21} > 0, 0 < \alpha_{22} < 1 \right\}, \text{ that is for matrices with null entries in the first row and positive entries in the second.}$

Proof. The proof is given in Appendix 5.E.

Remark. The proof of Proposition 5.4.1, Situation 1 reveals that non-identifiability stands actually for each subprocess (considered as a univariate process), such that all the submodels with null cross-interactions built by fixing α_{11} or α_{22} to zero, or by keeping them away from zero are also not identifiable.

Proposition 5.4.2 (Identifiability in the bivariate setting). The model Q_{Λ} is identifiable in the four situations:

- 1. $\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0 \\ \alpha_{21} & 0 \end{pmatrix}, 0 \le \alpha_{11} < 1, \alpha_{21} > 0 \right\}, \text{ that is for matrices } \alpha \text{ with null entries in the second column and a positive entry on the antidiagonal.}$
- 2. $\Lambda = \left\{ \begin{pmatrix} 0 & \alpha_{12} \\ 0 & \alpha_{22} \end{pmatrix}, \alpha_{12} > 0, 0 \le \alpha_{22} < 1 \right\}, \text{ that is for matrices } \alpha \text{ with null entries in the first column and a positive entry on the antidiagonal.}$
- 3. $\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0 \\ \alpha_{21} & \alpha_{22} \end{pmatrix}, 0 < \alpha_{11} < 1, \alpha_{21} > 0, 0 \le \alpha_{22} < 1 \right\}, \text{ that is for matrices } \alpha \text{ with positive entries in the first column and null upper right entry.}$
- 4. $\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & \alpha_{12} \\ 0 & \alpha_{22} \end{pmatrix}, 0 \le \alpha_{11} < 1, \alpha_{12} > 0, 0 < \alpha_{22} < 1 \right\}, \text{ that is for matrices } \alpha \text{ with a null lower left entry and positive entries in the second column.}$

Proof. The proofs of Situations 1 and 3 are respectively in Appendices 5.F and 5.G. The other situations are obtained by symmetry of all indices. \Box

Several lessons can be learnt from Propositions 5.4.1 and 5.4.2 and Remark 5.4. First, the statistical model Q_{Λ} is not identifiable if H reduces to a bivariate homogenous Poisson process (Proposition 5.4.1, Situation 1 with $\alpha_{11} = \alpha_{22} = 0$) or to two independent univariate Hawkes processes (Proposition 5.4.1, Situation 1 with $\alpha_{11} > 0$ and $\alpha_{22} > 0$) even if the noise P shares the same intensity λ_0 for both subprocesses. This result actually still holds true for a dimension d > 2.
Second, Proposition 5.4.2 tells in a nutshell that there must exist cross-interactions in the Hawkes process H (i.e. $\alpha_{12} > 0$ or $\alpha_{21} > 0$) for the model Q_{Λ} to be identifiable. However, interactions must not come from a Poisson subprocess and reach a self-exciting Hawkes subprocess (Proposition 5.4.1, Situations 2 and 3), but rather they must reach a self-neutral (i.e. with null self-excitation) Hawkes subprocess (Proposition 5.4.2, Situations 1 and 2) or come from a self-exciting Hawkes subprocess (Proposition 5.4.2, Situations 3 and 4).

5.5 Numerical results

This section numerically illustrates the behaviour of the proposed estimator $\hat{\theta}$ (described in Section 5.1.1) for noisy exponential Hawkes processes. It investigates the effect of horizon T and hyperparameter M in the univariate setting (Section 5.5.1), and the impact of model sparsity and interaction strength in the bivariate setting (Section 5.5.2).

In the whole study, point processes are simulated thanks to Ogata's thinning method (Ogata 1981) and numerical optimisation of the spectral log-likelihood is performed via the L-BFGS-B method (Byrd et al. 1995), implemented in the scipy.optimize.minimize Python function. Both simulation and estimation algorithms are freely available as a Python package on GitHub¹

5.5.1 Univariate setting

Simulation and estimation scenarios

Data simulation We consider observations $(T_k^N)_{1 \le k \le N(T)}$ coming from a univariate exponential noisy Hawkes process N = H + P, where P is a Poisson process with intensity $\lambda_0 > 0$ and H is a Hawkes process with baseline intensity $\mu = 1$ and kernel given by Equation (5.12) with parameters $\alpha = 0.5$ and $\beta = 1$.

In order to get close to stationarity while coping with inability to generate a process on the whole line \mathbb{R} , N is simulated on the window [-100, T] with no points in $(-\infty, -100)$ but only observations falling in [0, T] are considered.

The forthcoming section will illustrate the convergence of θ , thanks to its behaviour with respect to varying horizon $T \in \{250, 500, 1000, \dots, 8000\}$, and the impact of the intensity λ_0 of the noise process P on estimation accuracy, via varying noise-to-signal ratio $\lambda_0/m^H \in \{0.2, 0.4, \dots, 2.0\}$ (given the average intensity $m^H = \mu/(1 - \alpha) = 2$ of H).

Statistical models According to Proposition 5.3.1, which states four collections of identifiable models for univariate exponential Hawkes processes, estimation is successively performed in models Q_{μ} , Q_{α} , Q_{β} , Q_{λ_0} , where the known parameter is fixed to the value of the generated process (see above).

In addition, the behaviour of $\hat{\theta}$ will be assessed thanks to its relative error $\|\hat{\theta} - \theta^*\|_2 / \|\theta^*\|_2$ (where $\theta^* = (\mu, \alpha, \beta, \lambda_0)$ is the vector of the parameters of the generated process) averaged over 50 different trials.

Convergence, computation time and influence of parameter M

Up to now, the hyperparameter M, appearing in the spectral log-likelihood (Equations (5.4) and (5.11)) and determining the number of frequencies tested with the spectral density, has been let

¹https://github.com/migmtz/noisy-hawkes-process

free. However, the theoretical literature suggests that its choice has a lot of influence on the convergence of the estimation procedure, and has to be guided by $M/T \xrightarrow[T \to \infty]{} \infty$ (Pham 1996).

Since the rate of convergence is not specified but has an effect on the computational efficiency of the spectral estimator, we propose to study the compatible case $M = N(T) \log N(T)$ and the economy case M = N(T).

Figure 5.1 displays the relative errors $\|\hat{\theta} - \theta^*\|_2 / \|\theta^*\|_2$ with respect to both the simulation horizon T (top panels) and the estimation time (bottom panels). As expected, the quality of the estimations improves as T increases, and this independently of which parameter is fixed. Estimations are slightly better when considering $M = N(T) \log N(T)$ (orange line) especially for smaller values of T but the trade-off is a ten times higher computation time. Therefore, the performance benefit of taking $M = N(T) \log N(T)$ rather than M = N(T) seems minor when compared to the computational cost. For this reason, the forthcoming numerical experiments will be performed with M = N(T).



Figure 5.1: Relative estimation error with confidence bands (±1.96 empirical standard deviation) respectively for μ , α , β and λ_0 fixed (columns from left to right) with respect to the time horizon T (top) and the computation time (bottom) in logarithmic scale. Level of noise $\lambda_0 = 1.6$.

Influence of the noise level

Figure 5.2 shows the relative error with respect to the ratio λ_0/m_H , obtained when increasing the value of λ_0 while keeping the other parameters fixed, for a given horizon T = 8000.

First, we can see that the value of λ_0 has a low impact on the quality of estimations. However, let us notice that when β is fixed, the average error is substantially larger than when any of the other parameters is fixed. This could be explained by a compensation phenomenon inside the triplet (μ, α, λ_0) which occurs as our method implicitly adjusts the estimation to the mean intensity of the noisy Hawkes process:

$$m^N = \lambda_0 + \frac{\mu}{1 - \alpha},$$



Figure 5.2: Relative estimation error with confidence bands (± 1.96 empirical standard deviation) respectively for μ , α , β and λ_0 fixed with respect to the noise-to-signal ratio for the maximal horizon T = 8000.

which is indeed a quantity independent of β .

This numerical compensation is illustrated in Figure 5.3, where we can see that overestimating μ is systematically balanced by underestimating α and λ_0 and vice versa, whereas the estimated mean intensities remain accurate. In this experiment, the level of noise has been arbitrarily fixed to $\lambda_0 = 1.2$ but the observed behaviour appears similarly for all possible values of λ_0 used in the previous section.

When performing estimation with α fixed (the case for which the average error is the second largest, as illustrated in Figure 5.2), the compensation appears only between $\hat{\mu}$ and $\hat{\lambda}_0$ whereas $\hat{\beta}$ does not seem impacted, as shown in Figure 5.4.

5.5.2 Bivariate setting

This section illustrates numerical estimation of bivariate exponential noisy Hawkes processes (see Section 5.4) when conditions of identifiability are met (Proposition 5.4.2). We carry out two different studies, exploring different scenarios: Section 5.5.2 studies the influence of the strength of the cross-interaction between the two subprocesses and Section 5.5.2 investigates the performance of the estimator with and without knowledge of the null components. Indeed, since identifiability conditions depend on knowing which components are non-null, an information that is unlikely to be available in practical applications, we compare the performance of the estimator for both the reduced model Q_{Λ} , where the null components are known, and the complete model,

$$\mathcal{Q} = \left\{ \mathbf{f}_{\theta}^{N} : \mathbb{R} \to \mathbb{C}^{2 \times 2}, \theta = (\mu, \alpha, \beta, \lambda_{0}) \in \mathbb{R}^{2}_{>0} \times \mathbb{R}^{2 \times 2}_{\geq 0} \times \mathbb{R}^{2}_{>0} \times \mathbb{R}_{>0}, \rho(\alpha) < 1 \right\},$$

with no prior information.

Throughout this section, we consider a Hawkes process with $\mu = \begin{pmatrix} 1.0 \\ 1.0 \end{pmatrix}$ and $\beta = \begin{pmatrix} 1.0 \\ 1.3 \end{pmatrix}$. In



Figure 5.3: Estimations $\hat{\mu}$, $\hat{\alpha}$ and $\hat{\lambda}_0$ (of μ , α , and λ_0) for $\lambda_0 = 1.2$ when β is fixed, sorted by the values of $\hat{\mu}$ (top). In all plots, each color corresponds to one of the 50 repetitions. Boxplot of estimated mean intensities (bottom).

addition, fortified by the analysis of the univariate setting, the Poisson intensity is chosen to be $\lambda_0 = 0.5$ (the level of noise does not appear to have a significant impact on the quality of estimations, Figure 5.2) and it is considered M = N(T) (which provides accurate estimations in a reasonable amount of time, Figure 5.1).

Influence of the cross-interaction

Let us consider one of the identifiable scenarios where the only non-null interaction in the Hawkes process is one of the two cross-interactions (see Proposition 5.4.2, Situation 1). More precisely, we consider the reduced model Q_{Λ} , where

$$\Lambda = \left\{ \begin{pmatrix} 0 & 0\\ \alpha_{21} & 0 \end{pmatrix} : \alpha_{21} > 0 \right\} \,.$$

The Hawkes process is then simulated with different levels of cross-interaction: $\alpha_{21} \in \{0.2, 0.4, 0.6, 0.8\}$, and estimations are obtained by optimising the spectral log-likelihood on the non-null parameters $\mu_1, \mu_2, \alpha_{21}, \beta_2$, and λ_0 .

Figure 5.5 illustrates the influence of the true parameter α_{21} on the quality of the estimations, through the relative error of the estimations for the different values of α_{21} and an increasing range of horizons T. As a complement to what has been observed in Figure 5.1, our estimator appears to behave particularly well for higher values of T, but also for higher values of α_{21} .



Figure 5.4: Estimations $\hat{\mu}, \hat{\beta}$ and $\hat{\lambda}_0$ (of μ, β , and λ_0) for $\lambda_0 = 1.2$ when α is fixed, sorted by the values of $\hat{\mu}$ (top). In all plots, each color corresponds to one of the 50 repetitions. Boxplot of estimated mean intensities (bottom).



Figure 5.5: Heatmap of relative errors for each estimation $\hat{\mu}_1$, $\hat{\mu}_2$, $\hat{\alpha}_{21}$, $\hat{\beta}_2$, and $\hat{\lambda}_0$ for different levels of interaction α_{21} (x-axis) and horizons T (y-axis).

This is not surprising since, for smaller values of α_{21} , the Hawkes process behaves closely to a homogeneous Poisson process, and as proven in Proposition 5.4.1, the superposition of two Poisson processes leads to a non-identifiable model. Lower interactions necessitate then higher values of T to obtain satisfactory results. Inversely, for average and high interaction magnitudes, we start to obtain small errors for horizon values around T = 3000.

Influence of null interactions

In this section we simulate 50 repetitions with a fixed horizon T = 3000 for two identifiable scenarios regarding the Hawkes process $H = (H_1, H_2)$.

Scenario 1 The matrix of interactions is:

$$\alpha = \begin{pmatrix} 0.5 & 0\\ 0.4 & 0 \end{pmatrix} \,,$$

corresponding to Proposition 5.4.2, Situation 1. In other terms, H_1 excites both subprocesses whereas H_2 has no influence on the dynamics (See Figure 5.6, left).

Scenario 2 The matrix of interactions is:

$$\alpha = \begin{pmatrix} 0.5 & 0\\ 0.4 & 0.4 \end{pmatrix} \,,$$

corresponding to Proposition 5.4.2, Situation 3. In other terms, H_1 excites both subprocesses and H_2 excites itself (See Figure 5.6, right).



Figure 5.6: Interactions in the two numerical scenarios considered.

Graphics in the left column of Figure 5.7 present the boxplots of each parameter estimation when considering their respective reduced models Q_{Λ} . These results show that our method provides unbiased estimates of all parameters and is particularly efficient at inferring the interaction matrix α (estimations have very low variance). The larger variances are observed for parameters μ_1 , μ_2 and λ_0 , which is probably due to compensation effects already mentioned in Section 5.5.1.

Estimating in the reduced model Q_{Λ} requires prior knowledge on the null parameters α_{ij} ($1 \leq i, j \leq 2$), which is unlikely in practical applications. Therefore, we compare the results obtained in the reduced model to those in the complete model Q (see the right column graphics in Figure 5.7). If the estimates of α and β seem still empirically unbiased, we observed several deteriorations compared to the previous results. First, we notice a bias in the estimates of μ_1 , μ_2 and λ_0 : more precisely, μ_1 and μ_2 are overestimated while λ_0 is underestimated in both scenarios. Moreover, we observe in Scenario 2 some outlier estimations for the α_{ij} ($1 \leq i, j \leq 2$) coefficients, which did not appear when considering the reduced model.

Fortunately, Figure 5.7 also suggests that our estimator is able to detect the null interactions in the full model, which allows to re-estimate the parameters in the reduced model. To do so, we propose to look at the 5%-empirical quantile of each term of the estimated interaction matrix $\hat{\alpha}$, which are summarised in Table 5.1.

We can notice that the empirical 5%-quantiles are set to zero for each real null parameters α_{ij} $(1 \le i, j \le 2)$ in both scenarios. This suggests that when enough repetitions are available,



Figure 5.7: Boxplots of parameter estimations in the reduced model Q_{Λ} (left) and full model Q (right) in Scenario 1 (top) and Scenario 2 (bottom) (50 trials). Average estimation (green triangle) is to be compared to true parameter (blue point).

	α_{11}	α_{12}	α_{21}	α_{22}
Scenario 1	0.40	0.00	0.32	0.00
Scenario 2	0.41	0.00	0.34	0.35

Table 5.1: 5%-quantile of each parameter of the estimated interaction matrix $\hat{\alpha}$. The true parameter α_{12} is null in both scenarios and α_{22} is null only in Scenario 1.

it is possible to use these empirical quantiles to estimate the null interactions. An estimation

procedure when no prior information is known about the support of the interaction graph would then consist in a three-step approach: first, estimating all parameters in the complete model Q; second, computing the 5%-quantiles for all estimated interactions parameters in matrix $\hat{\alpha}$, and defining the support of α to be entries corresponding to a positive empirical quantile; finally, re-estimating all parameters in the reduced model defined by the support of α . Let us remark that the proposed support estimation step boils down to correspond to a multiple test that, when the noisy Hawkes process has significantly more that 2 dimensions, can be corrected thanks to usual procedures such as Bonferroni and Benjamini-Hochberg methods.

5.6 Discussion

In this chapter, we propose a spectral approach to estimate the parameters of a noisy Hawkes process, the performance of which is illustrated on an extended numerical study. Although we highlight the great benefit of considering a spectral analysis when standard inference methods are not available, we also bring out identifiability issues that may arise, either from the model itself or from the spectral approach. While we exhibit several identifiable and non-identifiable scenarios in both univariate and bivariate contexts, a general result on identifiability is still to be established, in particular in higher dimensions. For this purpose, the number of non-null cross-interactions and the choice of the kernel functions appear to be key elements in order to obtain identifiability guarantees.

More generally, we believe that the spectral analysis can provide efficient estimators in many frameworks of inaccurately or partially observed data. A natural extension of this work is to consider alternative forms for the noise process, for instance an heterogeneous Poisson process. Another topic of interest is to investigate another mechanism for the noise when some points are randomly missing. This would be complementary to our work since it would allow to model both false positive and false negative occurrences. In practice, this could be of great interest for applications, especially for the tracking of epidemics. Finally, let us mention that this chapter focuses on the linear Hawkes model, which excludes notably inhibition phenomenons. Though the nonlinear framework is particularly interesting for many applications, for instance in neuroscience, all the spectral theory for point processes only exists in a linear context so that we believe that developing a spectral inference procedure for nonlinear processes would be very challenging and remains a widely open topic.

5.A Proof of Proposition 5.2.1

Lemma 5.A.1. Let X and Y be two independent and stationary multivariate point processes with same dimension d. If they admit second order moment measures, denoted respectively $(M_{ij}^X)_{1\leq i,j\leq d}$ and $(M_{ij}^Y)_{1\leq i,j\leq d}$, then the process N = X + Y also admits a second order moment measure, noted $(M_{ij}^N)_{1\leq i,j\leq d}$, and for any pair $(A, B) \in (\mathcal{B}^c_{\mathbb{R}})^2$ and all $1 \leq i, j \leq d$:

$$M_{ij}^{N}(A,B) = M_{ij}^{X}(A,B) + M_{ij}^{Y}(A,B) + (m_{i}^{X}m_{j}^{Y} + m_{j}^{X}m_{i}^{Y})\ell_{\mathbb{R}}(A)\ell_{\mathbb{R}}(B), \qquad (5.15)$$

where for all $i \in \{1, \ldots, d\}$ and $m_i^X = \mathbb{E}[X_i([0, 1))]$ and $m_i^Y = \mathbb{E}[Y_i([0, 1))]$. Furthermore, the reduced measure of N reads:

$$\breve{M}_{ij}^{N}(B) = \breve{M}_{ij}^{X}(B) + \breve{M}_{ij}^{Y}(B) + (m_{i}^{X}m_{j}^{Y} + m_{j}^{X}m_{i}^{Y})\ell_{\mathbb{R}}(B).$$
(5.16)

Proof. Let $(A, B) \in (\mathcal{B}^c_{\mathbb{R}})^2$. Then for any $1 \leq i, j \leq d$:

$$\begin{split} M_{ij}^{N}(A,B) &= \mathbb{E}[N_{i}(A)N_{j}(B)] \\ &= \mathbb{E}[(X_{i}(A) + Y_{i}(A))(X_{j}(B) + Y_{j}(B))] \\ &= \mathbb{E}[X_{i}(A)X_{j}(B)] + \mathbb{E}[X_{i}(A)Y_{j}(B)] + \mathbb{E}[X_{j}(A)Y_{i}(B)] + \mathbb{E}[Y_{j}(A)Y_{j}(B)] \\ &= M_{ij}^{X}(A,B) + \mathbb{E}[X_{i}(A)]\mathbb{E}[Y_{j}(B)] + \mathbb{E}[X_{j}(A)]\mathbb{E}[Y_{i}(B)] + M_{ij}^{Y}(A,B) \\ &= M_{ij}^{X}(A,B) + m_{i}^{X}m_{j}^{Y}\ell(A)\ell(B) + m_{j}^{X}m_{i}^{Y}\ell(A)\ell(B) + M_{ij}^{Y}(A,B) \,, \end{split}$$

where the last line comes from the stationarity, which implies that $\mathbb{E}[X_i(A)] = m_i^X \ell(A)$. By applying Equation (5.5) to the function $g(x, y) = \mathbb{1}_{x \in [0,1]} \mathbb{1}_{y-x \in B}$ for any $B \in \mathcal{B}^c_{\mathbb{R}}$, we can remark that:

$$\int_{\mathbb{R}^2} g(x,y) M_{ij}^N(\mathrm{d} x,\mathrm{d} y) = \int_{[0,1]} \ell_{\mathbb{R}}(\mathrm{d} x) \int_B \breve{M}_{ij}^N(\mathrm{d} u) = \breve{M}_{ij}^N(B) \, du$$

In particular this equality is also true if we replace N by X and Y. By leveraging Equation (5.15) on the left-side integral, we obtain:

$$\begin{split} \int_{\mathbb{R}^2} g(x,y) \, M_{ij}^N(\mathrm{d}x,\mathrm{d}y) &= \breve{M}_{ij}^X(B) + \breve{M}_{ij}^Y(B) + (m_i^X m_j^Y + m_j^X m_i^Y) \int_{x \in [0,1]} \int_{y \in B+x} \ell_{\mathbb{R}}(\mathrm{d}y) \, \ell_{\mathbb{R}}(\mathrm{d}x) \\ &= \breve{M}_{ij}^X(B) + \breve{M}_{ij}^Y(B) + (m_i^X m_j^Y + m_j^X m_i^Y) \ell_{\mathbb{R}}([0,1]) \ell_{\mathbb{R}}(B) \\ &= \breve{M}_{ij}^X(B) + \breve{M}_{ij}^Y(B) + (m_i^X m_j^Y + m_j^X m_i^Y) \ell_{\mathbb{R}}(B) \,, \end{split}$$

which achieves the proof.

Proof of Proposition 5.2.1

By definition of the spectral density, for all $\omega \in \mathbb{R}$:

$$\begin{split} f_{ij}^{N}(\omega) &= \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^{N}(dx) - m_{i}^{N} m_{j}^{N} \delta(\omega) \\ &= \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^{X}(dx) + \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^{Y}(dx) \\ &\quad + (m_{i}^{X} m_{j}^{Y} + m_{j}^{X} m_{i}^{Y}) \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \ell_{\mathbb{R}}(dx) - (m_{i}^{X} + m_{i}^{Y})(m_{j}^{X} + m_{j}^{Y}) \delta(\omega) \\ &= \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^{X}(dx) - m_{i}^{X} m_{j}^{X} \delta(\omega) + \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \breve{M}_{ij}^{Y}(dx) - m_{i}^{Y} m_{j}^{Y} \delta(\omega) \\ &\quad + (m_{i}^{X} m_{j}^{Y} + m_{j}^{X} m_{i}^{Y}) \int_{\mathbb{R}} e^{-2\pi i x \omega} \, \ell_{\mathbb{R}}(dx) - (m_{i}^{X} m_{j}^{Y} + m_{i}^{Y} m_{j}^{X}) \delta(\omega) \\ &= f_{ij}^{X}(\omega) + f_{ij}^{Y}(\omega) + (m_{i}^{X} m_{j}^{Y} + m_{j}^{X} m_{i}^{Y}) \int_{\mathbb{R}} e^{-2\pi i x \omega} \, dx - (m_{i}^{X} m_{j}^{Y} + m_{i}^{Y} m_{j}^{X}) \delta(\omega) \,. \end{tabular}$$

By properties of the Dirac measure, for all $\omega \in \mathbb{R}$:

$$\int_{\mathbb{R}} e^{-2\pi i x \omega} \delta(x) \, \mathrm{d}x = 1 \,,$$

and so by duality of the Fourier transform Pinsky (2008, Proposition 5.2.4.) it follows that:

$$\int_{\mathbb{R}} e^{-2\pi i x\omega} \, \mathrm{d}x = \delta(\omega) \,.$$

The last two terms of Equation (5.17) being equal, we obtain $f_{ij}^N = f_{ij}^X + f_{ij}^Y$.

5.B Proof of Proposition 5.3.1

We first show that equality of two spectral densities with different parameters is equivalent to a system of equations (Equations (5.18)). This result will be used then to prove both parts of Proposition 5.3.1.

Let $(\mu, \alpha, \beta, \lambda_0)$ and $(\mu', \alpha', \beta', \lambda'_0)$ be two admissible 4-tuples for the exponential noisy Hawkes model. Let us assume that for all $\omega \in \mathbb{R}$:

$$f^{N}_{(\mu,\alpha,\beta,\lambda_0)}(\omega) = f^{N}_{(\mu',\alpha',\beta',\lambda_0')}(\omega)$$

Thanks to Equation (5.14), this equality implies the following system of equations:

$$\begin{cases} \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} + \lambda'_0 & (\omega \to +\infty) \\ \frac{\mu}{1-\alpha} \frac{\beta^2 \alpha (2-\alpha)}{\beta^2 (1-\alpha)^2} + \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} \frac{\beta'^2 \alpha' (2-\alpha')}{\beta'^2 (1-\alpha')^2} + \frac{\mu'}{1-\alpha'} + \lambda'_0 & (\omega = 0) \\ \frac{\mu}{1-\alpha} \frac{\beta^2 \alpha (2-\alpha)}{\beta^2 (1-\alpha)^2 + 4\pi^2} + \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} \frac{\beta'^2 \alpha' (2-\alpha')}{\beta'^2 (1-\alpha')^2 + 4\pi^2} + \frac{\mu'}{1-\alpha'} + \lambda'_0 & (\omega = 1). \end{cases}$$

The first equality can be used to simplify the second and third equalities, leading to:

$$\begin{cases} \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} + \lambda'_0 \\ \frac{\mu}{1-\alpha} \frac{\alpha(2-\alpha)}{(1-\alpha)^2} = \frac{\mu'}{1-\alpha'} \frac{\alpha'(2-\alpha')}{(1-\alpha')^2} \\ \frac{\mu}{1-\alpha} \frac{\beta^2 \alpha(2-\alpha)}{\beta^2 (1-\alpha)^2 + 4\pi^2} = \frac{\mu'}{1-\alpha'} \frac{\beta'^2 \alpha'(2-\alpha')}{\beta'^2 (1-\alpha')^2 + 4\pi^2} \end{cases}$$

Now, given that $\alpha \in (0, 1)$ (same for α'), the last two equalities lead to:

$$\frac{\beta^2 (1-\alpha)^2}{\beta^2 (1-\alpha)^2 + 4\pi^2} = \frac{{\beta'}^2 (1-\alpha')^2}{{\beta'}^2 (1-\alpha')^2 + 4\pi^2} \,,$$

which in turn implies $\beta(1-\alpha) = \beta'(1-\alpha')$.

Thus, it results the following system of equations:

$$\int \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} + \lambda'_0 \tag{5.18a}$$

$$\int \frac{\mu\alpha(2-\alpha)}{(1-\alpha)^3} = \frac{\mu'\alpha'(2-\alpha')}{(1-\alpha')^3}$$
(5.18b)

$$\zeta \beta(1-\alpha) = \beta'(1-\alpha').$$
(5.18c)

Now, Equations (5.18) lead to

$$\begin{cases} \frac{\mu}{1-\alpha} + \lambda_0 = \frac{\mu'}{1-\alpha'} + \lambda'_0 \\ \frac{\mu}{1-\alpha} \beta^2 \alpha (2-\alpha) = \frac{\mu'}{1-\alpha'} {\beta'}^2 \alpha' (2-\alpha') \\ \beta (1-\alpha) = \beta' (1-\alpha') \,, \end{cases}$$

which, by Equation (5.14), implies straightforwardly $f^{N}_{(\mu,\alpha,\beta,\lambda_0)} = f^{N}_{(\mu',\alpha',\beta',\lambda'_0)}$. Consequently,

$$f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} = f^{N}_{(\mu',\alpha',\beta',\lambda'_{0})} \iff \text{Equations (5.18)}$$

The model Q is not identifiable.

Let $\tau > -\lambda_0$ and $\lambda'_0 = \lambda_0 + \tau > 0$. Then, by denoting $\kappa = \frac{\mu}{1-\alpha}\beta^2\alpha(2-\alpha)$, Equations (5.18) are equivalent to:

$$\begin{cases} \mu' = \frac{\beta(1-\alpha)(\frac{\mu}{1-\alpha}-\tau)}{\sqrt{\beta^2(1-\alpha)^2 + \frac{\kappa}{1-\alpha}-\tau}} \\ \alpha' = 1 - \frac{\beta(1-\alpha)}{\sqrt{\beta^2(1-\alpha)^2 + \frac{\kappa}{1-\alpha}-\tau}} \\ \beta' = \sqrt{\beta^2(1-\alpha)^2 + \frac{\kappa}{\frac{\mu}{1-\alpha}-\tau}} . \end{cases}$$
(5.19)

From (5.19), it is clear that for all $\tau \in \left(-\lambda_0, \frac{\mu}{1-\alpha}\right) \setminus \{0\}, (\mu', \alpha', \beta', \lambda_0 + \tau) \neq (\mu, \alpha, \beta, \lambda_0)$ is an admissible parameter for \mathcal{Q} and $f^N_{(\mu', \alpha', \beta', \lambda_0 + \tau)} = f^N_{(\mu, \alpha, \beta, \lambda_0)}$. Consequently, \mathcal{Q} is not identifiable.

The reduced model defined by a triplet of admissible parameters is identifiable.

It will be shown that, for admissible parameters, $f^N_{(\mu,\alpha,\beta,\lambda_0)} = f^N_{(\mu',\alpha',\beta',\lambda_0')}$ implies

$$\mu = \mu' \iff \alpha = \alpha' \iff \beta = \beta' \iff \lambda_0 = \lambda'_0 \,,$$

from which we can deduce identifiability of the four collections of models mentioned in Proposition 5.3.1. Indeed, let, for instance, $\alpha^{\circ} \in (0, 1)$ and consider

$$\mathcal{Q}_{\alpha} = \left\{ f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} \in \mathcal{Q}, \alpha = \alpha^{\circ} \right\}$$

Then for all $f^N_{(\mu,\alpha,\beta,\lambda_0)} \in \mathcal{Q}_{\alpha}$ and $f^N_{(\mu',\alpha',\beta',\lambda'_0)} \in \mathcal{Q}_{\alpha}$,

$$f^{N}_{(\mu,\alpha,\beta,\lambda_{0})} = f^{N}_{(\mu',\alpha',\beta',\lambda'_{0})} \implies \begin{cases} \mu = \mu' \iff \alpha = \alpha' \iff \beta = \beta' \iff \lambda_{0} = \lambda'_{0} \\ \alpha = \alpha^{\circ} = \alpha' \end{cases} \implies \begin{cases} \mu = \mu' \\ \beta = \beta' \\ \lambda_{0} = \lambda'_{0} \\ \alpha = \alpha' \end{cases}$$

So, let us assume that $f_{(\mu,\alpha,\beta,\lambda_0)} = f_{(\mu',\alpha',\beta',\lambda_0')}$, for some admissible parameters. As shown beforehand this implies Equations (5.18). From Equation (5.18c), we establish that

$$\alpha = \alpha' \iff \beta = \beta',$$

since, $\beta > 0$ and $\alpha < 1$.

From Equation (5.18b), since $\alpha > 0$, it is clear that $\alpha = \alpha' \implies \mu = \mu'$. Conversely, if $\mu = \mu' > 0$, Equation (5.18b) becomes:

$$g(\alpha) = \frac{\alpha(2-\alpha)}{(1-\alpha)^3} = \frac{\alpha'(2-\alpha')}{(1-\alpha')^3} = g(\alpha').$$

In particular, g is a strictly increasing function on (0, 1) and so it can be deduced that $\alpha = \alpha'$. So

$$\mu = \mu' \iff \alpha = \alpha'.$$

Finally, as $\mu = \mu' \iff \alpha = \alpha'$, from Equation (5.18a) we conclude that $\alpha = \alpha' \implies \lambda_0 = \lambda'_0$. Let us now assume that $\lambda_0 = \lambda'_0$, Equation (5.18a) then reads

$$\frac{\mu}{1-\alpha} = \frac{\mu'}{1-\alpha'}$$

As $\mu > 0$ and $\mu' > 0$, Equation (5.18b) can be then reduced to

$$s(\alpha) := \frac{\alpha(2-\alpha)}{(1-\alpha)^2} = \frac{\alpha(2-\alpha)}{(1-\alpha)^2} = s(\alpha'),$$

where s is a strictly increasing function on (0,1) and so $\alpha = \alpha'$. With this we have proved that $\alpha = \alpha' \iff \lambda_0 = \lambda'_0$ which achieves the proof.

5.C Proof of Proposition 5.3.2

Let $(\mathcal{H}_t^N)_{t\geq 0}$ (respectively $(\mathcal{H}_t^H)_{t\geq 0}$) be the natural filtration associated with $(\lambda^N(t))_{t\geq 0}$ (respectively $(\lambda^H(t))_{t\geq 0}$). Let us now consider the conditional survival function of the first non-negative jump τ_1 of N given the past (before 0): for all $t \geq 0$,

$$\mathbb{P}\left(\tau_{1} > t \mid \mathcal{H}_{0}^{N}\right) = \mathbb{P}\left(N(t) = 0 \mid \mathcal{H}_{0}^{N}\right)$$

$$= \mathbb{P}\left(P(t) = 0, H(t) = 0 \mid \mathcal{H}_{0}^{N}\right)$$

$$= \mathbb{P}\left(P(t) = 0 \mid \mathcal{H}_{0}^{N}\right) \mathbb{P}\left(H(t) = 0 \mid \mathcal{H}_{0}^{N}\right) \qquad \text{(by independence)}$$

$$= \mathbb{P}\left(P(t) = 0\right) \mathbb{P}\left(H(t) = 0 \mid \mathcal{H}_{0}^{H}\right) \qquad \text{(by definition)}$$

$$= e^{-\lambda_{0}t} e^{-\mu(t-u_{t})-u_{t}\lambda^{H}(0)}.$$

with $u_t = (1 - e^{-\beta t})\beta^{-1}$ and where we have used that, given that H(t) = 0, for all $u \in [0, t]$,

$$\lambda^{H}(u) = \mu + \left(\lambda^{H}(0) - \mu\right) e^{-\beta u}$$

Let us remark that the last equality can also be deduced from Dassios et al. (2011, Corollary 3.3) with the same notation u_t :

$$\mathbb{P}\left(H(t) = 0 \mid \mathcal{H}_0^H\right) = \exp\left(-\int_0^{u_t} \frac{\mu\beta v}{1-\beta v} \,\mathrm{d}v\right) \mathrm{e}^{-u_t\lambda^H(0)}$$

It appears that $\mathbb{P}(\tau_1 > t \mid \mathcal{H}_0^N)$ depends on the past only through $\lambda^H(0)$. But $\lambda^H(0) = \lambda^N(0) - \lambda_0$ and $\lambda^N(0)$ is distributed according to the stationary distribution of $(\lambda^N(t))_{t\geq 0}$. It results that $\lambda^H(0)$ is distributed according to the stationary distribution of $(\lambda^H(t))_{t\geq 0}$, the Laplace transform of which is given by Dassios et al. (2011, Corollary 3.1). Plugging in this result, we have:

$$\mathbb{P}(\tau_1 > t) = e^{-\lambda_0 t - \mu(t - u_t)} \mathbb{E}\left[e^{-u_t \lambda^H(0)}\right]$$
$$= e^{-\lambda_0 t - \mu(t - u_t)} \exp\left(-\int_0^{u_t} \frac{\mu \beta v}{\beta v + e^{-\alpha\beta v} - 1} dv\right).$$

For ease of derivation, let $G : t \in \mathbb{R}_{\geq 0} \mapsto -\log \mathbb{P}(\tau_1 > t)$. The function G is differentiable and has derivative D given by:

$$D(t) = \lambda_0 + \mu \left(1 - e^{-\beta t}\right) + \frac{\mu \left(1 - e^{-\beta t}\right) e^{-\beta t}}{\exp\left(-\alpha \left(1 - e^{-\beta t}\right)\right) - e^{-\beta t}}.$$

Now define in the same manner τ'_1 , G', and D' from the process N' with parameter $(\mu', \alpha', \beta', \lambda'_0)$, with $\lambda^{N'}(0)$ being distributed according to the stationary distribution of $(\lambda^{N'}(t))_{t\geq 0}$, and assume that N and N' have the same distribution. Then it follows that both τ_1 and τ'_1 have the same distribution, so that G(t) = G'(t) for all $t \geq 0$. Since G and G' are everywhere differentiable and have the same initial value G(0) = G'(0) = 0, it results that D(t) = D'(t) for all $t \geq 0$.

We want to establish a system of four equations satisfied by the parameters that leads to the equality of the 4-tuples. First, noting that $\lim_{t\to\infty} D(t) = \lambda_0 + \mu$, we get

$$\lambda_0 + \mu = \lambda'_0 + \mu'. \tag{5.20}$$

Then, since $\lim_{t\to 0} D(t) = \lambda_0 + \mu/(1-\alpha)$, we get, using Equation (5.20),

$$\frac{\mu\alpha}{1-\alpha} = \frac{\mu'\alpha'}{1-\alpha'}\,.\tag{5.21}$$

To highlight two other equations on the parameters, we establish the Taylor expansion of D(t) around t = 0 up to order 2. After some calculation, we find that

$$D(t) = \lambda_0 + \frac{\mu}{1-\alpha} + \frac{\mu\alpha}{1-\alpha} \left(\frac{1}{2}\beta t \frac{\alpha-2}{1-\alpha} + \frac{1}{12}\beta^2 t^2 \frac{\alpha^3 - \alpha^2 - 3\alpha + 6}{(1-\alpha)^2} + o(t^2) \right)$$

From the first-order term of the expansion, Equation (5.21) and the equality of D and D', we find that

$$\beta \frac{\alpha - 2}{1 - \alpha} = \beta' \frac{\alpha' - 2}{1 - \alpha'}.$$
(5.22)

Then the second-order term of the expansion can be rewritten

$$\frac{\mu\alpha}{1-\alpha}\frac{1}{12}\left(\frac{\beta(\alpha-2)}{1-\alpha}\right)^2\frac{\alpha^3-\alpha^2-3\alpha+6}{(\alpha-2)^2}\,,$$

so that from Equations (5.21) and (5.22), we find that

$$g(\alpha) = \frac{\alpha^3 - \alpha^2 - 3\alpha + 6}{(\alpha - 2)^2} = \frac{\alpha'^3 - \alpha'^2 - 3\alpha' + 6}{(\alpha' - 2)^2} = g(\alpha').$$
(5.23)

 $\alpha \mapsto g(\alpha)$ can be shown to be strictly increasing for $\alpha \in (0, 1)$, so that Equation (5.23) yields that $\alpha = \alpha'$. With this remark, one can easily show from the system composed by Equations (5.20)–(5.23) that the tuples $(\mu, \alpha, \beta, \lambda_0)$ and $(\mu', \alpha', \beta', \lambda'_0)$ are equal.

5.D Proof of Proposition 5.3.3

Lemma 5.D.1. Assume all moments of h are finite: $\forall n \geq 0, m_n = \int t^n h(t) dt < \infty$. Then the spectral density f^N of the noisy Hawkes process N has a Taylor expansion around 0 given by:

$$\forall t \in \mathbb{R}: \quad f^{N}(t) = \frac{\mu}{(1-\alpha)^{3}} + \frac{\mu}{(1-\alpha)^{3}} \sum_{q \ge 1} \left(\frac{\alpha}{1-\alpha} \sum_{n \ge 1} b_{n} t^{2n} \right)^{q} + \lambda_{0}, \quad (5.24)$$

with

$$b_n = \frac{(-1)^n (2\pi)^{2n} a_n}{(2n)!}, \quad and \quad a_n = 2m_{2n} - \frac{\alpha}{1-\alpha} \sum_{k=1}^{2n-1} \binom{2n}{k} (-1)^k m_k m_{2n-k}. \quad (5.25)$$

Proof. Introduce the Taylor expansion of the Fourier transform of h:

$$\forall t \in \mathbb{R}: \quad \tilde{h}(t) = \sum_{n \ge 0} (i\tau)^n \frac{m_n}{n!},$$

where $\tau = -2\pi t$. Then, given that $m_0 = 1$,

$$\begin{split} \left| 1 - \alpha \tilde{h}(t) \right|^2 &= \left(1 - \alpha \sum_{n \ge 0} (i\tau)^n \frac{m_n}{n!} \right) \left(1 - \alpha \sum_{n \ge 0} (-i\tau)^n \frac{m_n}{n!} \right) \\ &= 1 - 2\alpha \sum_{n \ge 0} (-1)^n \tau^{2n} \frac{m_{2n}}{(2n)!} + \alpha^2 \sum_{n \ge 0} \sum_{k=0}^n (-1)^k (i\tau)^n \frac{m_{n-k} m_k}{(n-k)! k!} \\ &= 1 - \alpha \sum_{n \ge 0} 2m_{2n} \frac{(-1)^n \tau^{2n}}{(2n)!} + \alpha^2 \sum_{n \ge 0} \sum_{k=0}^{2n} (-1)^k m_k m_{2n-k} \binom{2n}{k} \frac{(-1)^n \tau^{2n}}{(2n)!} \\ &= (1 - \alpha)^2 \left(1 - \frac{\alpha}{1 - \alpha} \sum_{n \ge 1} a_n \frac{(-1)^n \tau^{2n}}{(2n)!} \right). \end{split}$$

Inverting this expression and taking the Taylor expansion of $x \mapsto (1-x)^{-1}$ around 0 yields the desired result.

Remark. For $n = 1, 2, a_n$ is given by

$$\begin{cases} a_1 = 2m_2 + \frac{\alpha}{1-\alpha}2m_1^2 \\ a_2 = 2m_4 + \frac{\alpha}{1-\alpha}(8m_1m_3 - 6m_2^2), \end{cases}$$

so that the Taylor expansion of f^N up to order 5 is:

$$f^{N}(t) = a + c_{1}t^{2} + c_{2}t^{4} + o(t^{5}),$$

with

$$a = \frac{\mu}{(1-\alpha)^3} + \lambda_0$$

$$c_1 = 4\frac{\mu\alpha\pi^2}{(1-\alpha)^4} \left[-m_2 - \frac{\alpha}{1-\alpha}m_1^2 \right]$$

$$c_2 = 16\frac{\mu\alpha\pi^4}{(1-\alpha)^4} \left[\frac{m_4}{12} + \frac{\alpha}{1-\alpha} \left(\frac{m_1m_3}{3} + \frac{3m_2^2}{4} \right) + \left(\frac{\alpha}{1-\alpha} \right)^2 2m_2m_1^2 + \left(\frac{\alpha}{1-\alpha} \right)^3 m_1^4 \right]$$

Proof of Proposition 5.3.3

Let $(\mu, \alpha, \phi, \lambda_0)$ and $(\mu', \alpha', \phi', \lambda'_0)$ be two admissible 4-tuples for the uniform noisy Hawkes model \mathcal{R} , and assume that, for all $\omega \in \mathbb{R}$,

$$f^{N}_{(\mu,\alpha,\phi,\lambda_{0})}(\omega) = f^{N}_{(\mu',\alpha',\phi',\lambda_{0}')}(\omega)$$

First, noting that $\lim_{\omega\to\infty} f^N_{(\mu,\alpha,\phi,\lambda_0)}(\omega) = \lambda_0 + \mu/(1-\alpha)$, we get

$$\lambda_0 + \frac{\mu}{1 - \alpha} = \lambda'_0 + \frac{\mu'}{1 - \alpha'} \,. \tag{5.26}$$

Then, since $f^N_{(\mu,\alpha,\phi,\lambda_0)}(0) = \lambda_0 + \mu/(1-\alpha)^3$ and using Equation (5.26), we get:

$$\frac{\mu\alpha(2-\alpha)}{(1-\alpha)^3} = \frac{\mu'\alpha'(2-\alpha')}{(1-\alpha')^3} \,. \tag{5.27}$$

Now, since the Taylor expansions, given by Equation (5.24), of $f_{(\mu,\alpha,\phi,\lambda_0)}^N$ and $f_{(\mu',\alpha',\phi',\lambda'_0)}^N$ around 0 coincide, their respective Taylor coefficients (c_1, c_2, \ldots) and (c'_1, c'_2, \ldots) are equal. Plugging in the moments of the uniform distribution on $[0, \phi]$,

$$m_n = \frac{\phi^n}{n+1}$$
, for all $n \ge 0$,

the first order coefficient c_1 can be written

$$c_1 = -\frac{\pi}{3} \frac{\mu \alpha (2-\alpha)}{(1-\alpha)^3} \phi^2 \frac{4-\alpha}{(1-\alpha)^2 (2-\alpha)},$$

so that, with the use of Equation (5.27), we get

$$\phi^2 \frac{4-\alpha}{(1-\alpha)^2 (2-\alpha)} = \phi'^2 \frac{4-\alpha'}{(1-\alpha')^2 (2-\alpha')} \,. \tag{5.28}$$

Similarly, by plugging the moments in the second order coefficient c_2 , we get

$$c_2 = \frac{\pi^4}{15} \frac{\mu \alpha (2-\alpha)}{(1-\alpha)^3} \left(\phi^2 \frac{4-\alpha}{(1-\alpha)^2 (2-\alpha)} \right)^2 \frac{(2-\alpha)(\alpha^3 - 8\alpha^2 + 18\alpha + 4)}{(4-\alpha)^2} \,,$$

so that, by Equations (5.27) and (5.28),

$$g(\alpha) = \frac{(2-\alpha)(\alpha^3 - 8\alpha^2 + 18\alpha + 4)}{(4-\alpha)^2} = \frac{(2-\alpha')(\alpha'^3 - 8\alpha'^2 + 18\alpha' + 4)}{(4-\alpha')^2} = g(\alpha').$$
(5.29)

But $\alpha \mapsto g(\alpha)$ can be shown to be strictly increasing for $\alpha \in (0, 1)$, so that Equation (5.29) yields that $\alpha = \alpha'$. Finally, it is easily proven from the system composed by Equations (5.26)–(5.29) that the tuples $(\mu, \alpha, \beta, \lambda_0)$ and $(\mu', \alpha', \beta', \lambda'_0)$ are equal.

5.E Proof of Proposition 5.4.1

Let N be a bivariate noisy Hawkes process parametrised by the exponential model Q_{Λ} . We will prove that if either of the conditions in Proposition 5.4.1 is fulfilled then Q_{Λ} is not identifiable.

• Condition 1: Let

$$\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0\\ 0 & \alpha_{22} \end{pmatrix}, 0 \le \alpha_{11}, \alpha_{22} < 1 \right\}$$

For any admissible $\theta = (\mu, \alpha, \beta, \lambda_0)$, the spectral matrix \mathbf{f}_{θ}^N is diagonal such that, for any integer $i \in \{1, 2\}$:

$$f_{\theta \ ii}^{N}(\omega) = \frac{\mu_{i}}{1 - \alpha_{ii}} \left(\frac{\beta_{i}^{2} \alpha_{ii}(2 - \alpha_{ii})}{\beta_{i}^{2}(1 - \alpha_{ii})^{2} + 4\pi^{2}\omega^{2}} \right) + \left(\frac{\mu_{i}}{1 - \alpha_{ii}} + \lambda_{0} \right)$$

We will show that there exists an admissible $\theta' = (\mu', \alpha', \beta', \lambda'_0)$ such that $\theta' \neq \theta$ and $\mathbf{f}_{\theta}^N = \mathbf{f}_{\theta'}^N$. Let $\tau > -\lambda_0$ and $\lambda'_0 = \lambda_0 + \tau$.

For every $i \in \{1, 2\}$, on the one hand, if $\alpha_{ii} \neq 0$ then, as shown in the univariate case (Appendix 5.B), letting $\kappa_i = \frac{\mu_i}{1 - \alpha_{ii}} \beta_i^2 \alpha_{ii} (2 - \alpha_{ii})$ and defining the parameters:

$$\begin{cases} \mu_{i}' = \frac{\beta_{i}(1-\alpha_{ii})(\frac{\mu_{i}}{1-\alpha_{ii}}-\tau)}{\sqrt{\beta_{i}^{2}(1-\alpha_{ii})^{2} + \frac{\kappa_{i}}{1-\alpha_{ii}}-\tau}} \\ \alpha_{ii}' = 1 - \frac{\beta_{i}(1-\alpha_{ii})}{\sqrt{\beta_{i}^{2}(1-\alpha_{ii})^{2} + \frac{\kappa_{i}}{1-\alpha_{ii}}-\tau}} \\ \beta_{i}' = \sqrt{\beta_{i}^{2}(1-\alpha_{ii})^{2} + \frac{\kappa_{i}}{\frac{\mu_{i}}{1-\alpha_{ii}}-\tau}}, \end{cases}$$
(5.30)

leads to $f^N_{\theta \ ii} = f^N_{\theta' \ ii}$.

On the other hand, if $\alpha_{ii} = 0$ then, $f^N_{\theta_{ii}} = \mu_i + \lambda_0$ and it is enough to consider:

$$\begin{cases} \mu'_i = \mu_i - \tau \\ \alpha'_{ii} = 0 \\ \beta'_i = \beta_i \,, \end{cases}$$

to get $f^N_{\theta \ ii} = f^N_{\theta' \ ii}$.

In both cases, for $\tau \in \left(-\lambda_0, \min_{1 \leq i \leq 2} \left\{\frac{\mu_i}{1-\alpha_{ii}}\right\}\right) \setminus \{0\}$, we obtain an admissible parameter $\theta' \neq \theta$ and $\mathbf{f}_{\theta'}^N = \mathbf{f}_{\theta'}^N$. Thus, the model is not identifiable.

• Condition 2 and 3: Without loss of generality, let

$$\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & \alpha_{12} \\ 0 & 0 \end{pmatrix}, 0 < \alpha_{11} < 1, \alpha_{12} > 0 \right\} \,.$$

For any admissible parameter $\theta = (\mu, \alpha, \beta, \lambda_0)$ and for any $\omega \in \mathbb{R}$, the spectral matrix reads:

$$\mathbf{f}_{\theta}^{N}(\omega) = \begin{bmatrix} f_{\theta}^{N}{}_{11}(\omega) & f_{\theta}^{N}{}_{12}(\omega) \\ f_{\theta}^{N}{}_{21}(\omega) & f_{\theta}^{N}{}_{22}(\omega) \end{bmatrix} = \begin{bmatrix} f_{\theta}^{N}{}_{11}(\omega) & \frac{\mu_{2}\beta_{1}\alpha_{12}}{\beta_{1}(1-\alpha_{11})+2\pi\mathrm{i}\omega} \\ \frac{\mu_{2}\beta_{1}\alpha_{12}}{\beta_{1}(1-\alpha_{11})-2\pi\mathrm{i}\omega} & \mu_{2}+\lambda_{0} \end{bmatrix}$$

with

$$\begin{split} f^N_{\theta \ 11}(\omega) = & \left(\frac{\mu_1}{1-\alpha_{11}} + \frac{\mu_2\alpha_{12}}{1-\alpha_{11}}\right) \frac{\beta_1^2\alpha_{11}(2-\alpha_{11})}{\beta_1^2(1-\alpha_{11})^2 + 4\pi^2\omega^2} + \frac{\mu_2\beta_1^2\alpha_{12}^2}{\beta_1^2(1-\alpha_{11})^2 + 4\pi^2\omega^2} \\ & + \frac{\mu_1}{1-\alpha_{11}} + \frac{\mu_2\alpha_{12}}{1-\alpha_{11}} + \lambda_0 \,. \end{split}$$

Let us introduce the following constants:

$$\begin{cases}
A = \mu_2 + \lambda_0 \\
B = \mu_2 \beta_1 \alpha_{12} \\
C = \beta_1 (1 - \alpha_{11}) \\
D = \frac{\mu_1 + \mu_2 \alpha_{12}}{1 - \alpha_{11}} + \lambda_0 \\
E = (\frac{\mu_1 + \mu_2 \alpha_{12}}{1 - \alpha_{11}}) \beta_1^2 \alpha_{11} (2 - \alpha_{11}) + \mu_2 \beta_1^2 \alpha_{12}^2 ,
\end{cases}$$

which allow us to rewrite the spectral matrix as:

$$\mathbf{f}_{\theta}^{N}(\omega) = \begin{bmatrix} \frac{E}{C^{2} + 4\pi^{2}\omega^{2}} + D & \frac{B}{C + 2\pi \mathrm{i}\omega} \\ \frac{B}{C - 2\pi \mathrm{i}\omega} & A \end{bmatrix}.$$
(5.31)

Let $\tau \in \mathbb{R} \setminus \{\mu_2, (\mu_1 + \mu_2 \alpha_{12})/(1 - \alpha_{11})\}$ and:

$$\kappa_{\tau} = \frac{(\mu_1 + \mu_2 \alpha_{12})\alpha_{11}(2 - \alpha_{11})(\mu_2 - \tau) - \tau \mu_2^2 \alpha_{12}^2(1 - \alpha_{11})}{(\mu_2 - \tau)(\mu_1 + \mu_2 \alpha_{12} - \tau(1 - \alpha_{11}))} \,.$$

Now, consider the parameter $\theta' = (\mu', \alpha', \beta', \lambda'_0)$, defined as follows:

$$\begin{cases} \mu_1' = \frac{\mu_1 - \tau (1 - \alpha_{11})}{\sqrt{(1 - \alpha_{11})^2 + \kappa_\tau}} \\ \mu_2' = \mu_2 - \tau \\ \alpha_{11}' = 1 - \frac{1 - \alpha_{11}}{\sqrt{(1 - \alpha_{11})^2 + \kappa_\tau}} \\ \alpha_{12}' = \frac{\mu_2 \alpha_{12}}{(\mu_2 - \tau)\sqrt{(1 - \alpha_{11})^2 + \kappa_\tau}} \\ \beta_1' = \beta_1 \sqrt{(1 - \alpha_{11})^2 + \kappa_\tau} \\ \lambda_0' = \lambda_0 + \tau \,. \end{cases}$$
(5.32)

The goal will be to show that there exists τ such that θ' is well defined, admissible for \mathcal{Q}_{Λ} , such that $\theta \neq \theta'$ and $\mathbf{f}_{\theta}^{N} = \mathbf{f}_{\theta'}^{N}$.

First, in order for θ' to be an admissible parameter, let us remark that $\rho(S) = \alpha_{11}$ and so we obtain the following constrains:

$$\begin{pmatrix}
\mu_{1}' > 0 \\
\mu_{2}' > 0 \\
0 < \alpha_{11}' < 1 \\
\alpha_{12}' > 0 \\
\beta_{1}' > 0 \\
\lambda_{0}' > 0
\end{pmatrix}
\begin{pmatrix}
\mu_{1} - \tau(1 - \alpha_{11}) > 0 \\
\mu_{2} - \tau > 0 \\
\frac{1 - \alpha_{11}}{\sqrt{(1 - \alpha_{11})^{2} + \kappa_{\tau}}} < 1 \\
(1 - \alpha_{11})^{2} + \kappa_{\tau} > 0 \\
\lambda_{0} + \tau > 0
\end{pmatrix}
\begin{pmatrix}
\tau < \frac{\mu_{1}}{1 - \alpha_{11}} \\
\tau < \mu_{2} \\
\kappa_{\tau} > 0 \\
\tau > -\lambda_{0}.
\end{pmatrix}$$
(5.33)

Since $\tau < \mu_1/(1-\alpha_{11}) \implies \mu_1 + \mu_2\alpha_{12} - \tau(1-\alpha_{11}) > 0$, the third inequality becomes

$$\tau < \frac{(\mu_1 + \mu_2 \alpha_{12})\alpha_{11}(2 - \alpha_{11})\mu_2}{(\mu_1 + \mu_2 \alpha_{12})\alpha_{11}(2 - \alpha_{11}) + \mu_2^2(1 - \alpha_{11})\alpha_{12}^2}.$$

Then for

$$\tau \in \left(-\lambda_0, \min\left\{\frac{\mu_1}{1-\alpha_{11}}, \mu_2, \frac{(\mu_1+\mu_2\alpha_{12})\alpha_{11}(2-\alpha_{11})\mu_2}{(\mu_1+\mu_2\alpha_{12})\alpha_{11}(2-\alpha_{11})+\mu_2^2(1-\alpha_{11})\alpha_{12}^2}\right\}\right) \setminus \{0\},\$$

the right-hand side of Equations (5.33) is verified and so θ' defined by Equations (5.32) is well defined, admissible for Q_{Λ} and $\theta' \neq \theta$.

Then, we can show that:

$$\begin{cases} \mu_2' + \lambda_0' = \mu_2 + \lambda_0 = A \\ \mu_2' \beta_1' \alpha_{12}' = \mu_2 \beta_1 \alpha_{12} = B \\ \beta_1' (1 - \alpha_{11}') = \beta_1 (1 - \alpha_{11}) = C \\ \frac{\mu_1' + \mu_2' \alpha_{12}'}{1 - \alpha_{11}'} + \lambda_0' = \frac{\mu_1 + \mu_2 \alpha_{12}}{1 - \alpha_{11}} + \lambda_0 = D \\ \left(\frac{\mu_1' + \mu_2 \alpha_{12}'}{1 - \alpha_{11}'}\right) \beta_1'^2 \alpha_{11}' (2 - \alpha_{11}') + \mu_2' \beta_1'^2 \alpha_{12}'^2 = \left(\frac{\mu_1 + \mu_2 \alpha_{12}}{1 - \alpha_{11}}\right) \beta_1^2 \alpha_{11} (2 - \alpha_{11}) + \mu_2 \beta_1^2 \alpha_{12}^2 = E . \end{cases}$$

This assures that, for all $\omega \in \mathbb{R}$:

$$\mathbf{f}_{\theta'}^{N}(\omega) = \begin{bmatrix} \frac{E}{C^{2} + 4\pi^{2}\omega^{2}} + D & \frac{B}{C + 2\pi \mathrm{i}\omega} \\ \frac{B}{C - 2\pi \mathrm{i}\omega} & A \end{bmatrix} = \mathbf{f}_{\theta}^{N}(\omega) \,,$$

Thus, the model is not identifiable.

Proof of Proposition 5.4.2, Situations 1 and 2 5.F

Without loss of generality, let

$$\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0 \\ \alpha_{21} & 0 \end{pmatrix}, 0 \le \alpha_{11} < 1, \alpha_{21} > 0 \right\} \,.$$

Then, for any admissible parameter $\theta = (\mu, \alpha, \beta, \lambda_0)$ and all $\omega \in \mathbb{R}$:

$$\begin{cases} f^{N}_{\theta \ 11}(\omega) = \frac{\mu_{1}}{1 - \alpha_{11}} \frac{\beta_{1}^{2} + 4\pi^{2}\omega^{2}}{\beta_{1}^{2}(1 - \alpha_{11})^{2} + 4\pi^{2}\omega^{2}} + \lambda_{0} \\ \\ f^{N}_{\theta \ 12}(\omega) = \frac{\mu_{1}}{1 - \alpha_{11}} \frac{\beta_{1}^{2} + 4\pi^{2}\omega^{2}}{\beta_{1}^{2}(1 - \alpha_{11})^{2} + 4\pi^{2}\omega^{2}} \frac{\alpha_{21}\beta_{2}}{\beta_{2} - 2\pi i\omega} \\ \\ f^{N}_{\theta \ 22}(\omega) = \frac{\mu_{1}}{1 - \alpha_{11}} \frac{\beta_{1}^{2} + 4\pi^{2}\omega^{2}}{\beta_{1}^{2}(1 - \alpha_{11})^{2} + 4\pi^{2}\omega^{2}} \frac{\alpha_{21}^{2}\beta_{2}^{2}}{\beta_{2}^{2} + 4\pi^{2}\omega^{2}} + \mu_{2} + \frac{\mu_{1}\alpha_{21}}{1 - \alpha_{11}} + \lambda_{0} \,, \end{cases}$$

and $f_{\theta \ 21}^{N}(\omega) = \overline{f_{\theta \ 12}^{N}(\omega)}$. These expressions can be reformulated as follows:

$$\begin{cases} f^N_{\theta \ 11}(\omega) = m^H_1 \frac{1}{|1 - \tilde{h}_{\theta \ 11}(\omega)|^2} + \lambda_0 \\ f^N_{\theta \ 12}(\omega) = (f^N_{\theta \ 11}(\omega) - \lambda_0)\tilde{h}_{\theta \ 21}(-\omega) \\ f^N_{\theta \ 22}(\omega) = \frac{|f^N_{\theta \ 12}(\omega)|^2}{f^N_{\theta \ 11}(\omega) - \lambda_0} + \lim_{\omega' \to +\infty} f^N_{\theta \ 22}(\omega') \end{cases}$$

where $m_1^H = \mu_1/(1 - \alpha_{11}), m_2^H = \mu_2 + \mu_1 \alpha_{21}/(1 - \alpha_{11})$ and $\lim_{\omega' \to +\infty} f_{\theta \ 22}^N(\omega') = m_2^H + \lambda_0$. Let $\theta' = (\mu', \alpha', \beta', \lambda'_0)$ be an admissible parameter such that $\mathbf{f}_{\theta}^N = \mathbf{f}_{\theta'}^N$. For $\omega = 0$, it comes $f_{\theta \ 22}^N(0) = f_{\theta' \ 22}^N(0)$, which implies that

$$\frac{|f^N_{\theta \ 12}(0)|^2}{f^N_{\theta \ 11}(0) - \lambda_0} = \frac{|f^N_{\theta' \ 12}(0)|^2}{f^N_{\theta' \ 11}(0) - \lambda_0'}$$

Since $|f_{\theta 12}^N(0)|^2 = \mu_1 \alpha_{21} (1 - \alpha_{11})^{-3} \neq 0$ (as $\alpha_{11} < 1$ and $\alpha_{21} > 0$), $f_{\theta 12}^N(0) = f_{\theta' 12}^N(0)$ and $f_{\theta 11}^N(0) = f_{\theta' 11}^N(0)$, it results that:

$$\lambda_0 = \lambda'_0$$
.

Now, for all $\omega \in \mathbb{R}$, since $f_{\theta \ 12}^N(\omega) = f_{\theta' \ 12}^N(\omega)$, it comes $(f_{\theta \ 11}^N(\omega) - \lambda_0)\tilde{h}_{\theta \ 21}(-\omega) = (f_{\theta' \ 11}^N(\omega) - \lambda_0')\tilde{h}_{\theta',21}(-\omega)$, then $\tilde{h}_{\theta \ 21}(\omega) = \tilde{h}_{\theta',21}(\omega)$. By Equation (5.13), it results that:

$$\frac{\alpha_{21}\beta_2}{\beta_2 + 2\pi \mathrm{i}\omega} = \frac{\alpha'_{21}\beta'_2}{\beta'_2 + 2\pi \mathrm{i}\omega} \,.$$

For $\omega = 0$, this simplifies to

For $\omega = 1$, as $\alpha_{21} \neq 0$ it comes $\beta_2/(\beta_2 + 2\pi i) = \beta'_2/(\beta'_2 + 2\pi i)$, then

$$\beta_2 = \beta_2' \,.$$

 $\alpha_{21} = \alpha'_{21} \,.$

Now, by considering the limits as $\omega \to \infty$, we obtain the two following equations:

$$\begin{cases} \frac{\mu_1}{1-\alpha_{11}} + \lambda_0 = \frac{\mu_1'}{1-\alpha_{11}'} + \lambda_0 \\ \mu_2 + \frac{\mu_1\alpha_{21}}{1-\alpha_{11}} + \lambda_0 = \mu_2' + \frac{\mu_1'\alpha_{21}}{1-\alpha_{11}'} + \lambda_0 , \end{cases}$$

which, using that $\mu_1 > 0$, can be simplified to:

$$\begin{cases} \frac{\mu_1}{1 - \alpha_{11}} = \frac{\mu'_1}{1 - \alpha'_{11}} \\ \mu_2 = \mu'_2 \,. \end{cases}$$

Now, as $f_{\theta 11}^N(0) = \mu_1/(1 - \alpha_{11})^3$, we have the two following equations:

$$\begin{cases} \frac{\mu_1}{1-\alpha_{11}} = \frac{\mu_1'}{1-\alpha_{11}'} \\ \frac{\mu_1}{(1-\alpha_{11})^3} = \frac{\mu_1'}{(1-\alpha_{11}')^3} \end{cases}$$

which imply

$$\begin{cases} \mu_1 = \mu'_1 \\ \alpha_{11} = \alpha'_{11} \end{cases}$$

At last, since $f_{\theta \ 11}^N(1) = f_{\theta' \ 11}^N(1)$ and $\mu_1 > 0$,

$$\frac{\beta_1^2 + 4\pi^2}{\beta_1^2 (1 - \alpha_{11})^2 + 4\pi^2} = \frac{\beta_1'^2 + 4\pi^2}{\beta_1'^2 (1 - \alpha_{11})^2 + 4\pi^2} \,,$$

which implies

$$\alpha_{11}(2 - \alpha_{11})\beta_1^2 = \alpha_{11}(2 - \alpha_{11})\beta_1^{\prime 2}$$

Either $\alpha_{11} > 0$, so the previous equation leads to

$$\beta_1 = \beta_1' \,,$$

or $\alpha_{11} = 0$, so $\alpha'_{11} = \alpha_{11} = 0$ and $\beta_1 = \beta'_1 = 1$ (since θ and θ' are admissible for the model). This proves that $\mathbf{f}^N_{\theta} = \mathbf{f}^N_{\theta'} \implies \theta = \theta'$, which achieves the proof.

Proof of Proposition 5.4.2, Situations 3 and 4 **5.**G

Without loss of generality, let

$$\Lambda = \left\{ \begin{pmatrix} \alpha_{11} & 0 \\ \alpha_{21} & \alpha_{22} \end{pmatrix}, 0 < \alpha_{11} < 1, \alpha_{21} > 0, 0 \le \alpha_{22} < 1 \right\}.$$

Then, for any admissible parameter $\theta = (\mu, \alpha, \beta, \lambda_0)$ and all $\omega \in \mathbb{R}$:

$$\begin{cases} f^{N}_{\theta\ 11}(\omega) = \frac{\mu_{1}}{1 - \alpha_{11}} \frac{\beta_{1}^{2} \alpha_{11}(2 - \alpha_{11})}{\beta_{1}^{2}(1 - \alpha_{11})^{2} + 4\pi^{2}\omega^{2}} + \frac{\mu_{1}}{1 - \alpha_{11}} + \lambda_{0} \\ f^{N}_{\theta\ 12}(\omega) = \left(f^{N}_{\theta\ 11}(\omega) - \lambda_{0}\right) \frac{\alpha_{21}\beta_{2}}{\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}\omega^{2}} \left(\beta_{2}(1 - \alpha_{22}) + 2\pi i\omega\right) \\ f^{N}_{\theta\ 22}(\omega) = \left(\frac{\mu_{1}\alpha_{21}}{(1 - \alpha_{22})(1 - \alpha_{11})} + \frac{\mu_{2}}{1 - \alpha_{22}}\right) \frac{\beta_{2}^{2} + 4\pi^{2}\omega^{2}}{\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}\omega^{2}} + \frac{\mu_{1}\alpha_{21}^{2}\beta_{2}^{2}}{1 - \alpha_{11}} \frac{\beta_{1}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}\omega^{2}}{(\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}\omega^{2})(\beta_{1}^{2}(1 - \alpha_{11})^{2} + 4\pi^{2}\omega^{2})} + \lambda_{0} \,, \end{cases}$$

and $f_{\theta}^{N}{}_{21}(\omega) = \overline{f_{\theta}^{N}{}_{12}(\omega)}$. Let $\theta' = (\mu', \alpha', \beta', \lambda'_0)$ be an admissible parameter such that $\mathbf{f}_{\theta}^{N} = \mathbf{f}_{\theta'}^{N}$. We start by showing that $\lambda_0 = \lambda'_0$. From $\mathfrak{Re}\left(f_{\theta}^{N}{}_{12}(1)\right) = \mathfrak{Re}\left(f_{\theta'}{}_{12}(1)\right)$ and $\mathfrak{Im}\left(f_{\theta}^{N}{}_{12}(1)\right) = \mathfrak{Im}\left(f_{\theta'}{}_{12}(1)\right)$, the following system of equations is established:

$$\begin{cases}
\frac{(f_{\theta}^{N}{}_{11}(1) - \lambda_{0})\alpha_{21}\beta_{2}}{\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}}\beta_{2}(1 - \alpha_{22}) = \frac{(f_{\theta'}^{N}{}_{11}(1) - \lambda'_{0})\alpha'_{21}\beta'_{2}}{\beta'_{2}{}^{2}(1 - \alpha'_{22})^{2} + 4\pi^{2}}\beta'_{2}(1 - \alpha'_{22})\\ \frac{(f_{\theta}^{N}{}_{11}(1) - \lambda_{0})\alpha_{21}\beta_{2}}{\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}} = \frac{(f_{\theta'}^{N}{}_{11}(1) - \lambda'_{0})\alpha'_{21}\beta'_{2}}{\beta'_{2}{}^{2}(1 - \alpha'_{22})^{2} + 4\pi^{2}}.
\end{cases}$$
(5.34)

Since

$$f_{\theta_{11}}^N(1) - \lambda_0 = \frac{\mu_1}{1 - \alpha_{11}} \frac{\beta_1^2 + 4\pi^2}{\beta_1^2 (1 - \alpha_{11})^2 + 4\pi^2} \neq 0$$

and $\alpha_{21}\beta_2 > 0$, Equations (5.34) imply that

$$\beta_2(1 - \alpha_{22}) = \beta_2'(1 - \alpha_{22}').$$
(5.35)

Now, from $f_{\theta 12}^N(0) = (f_{\theta 11}^N(0) - \lambda_0) \alpha_{21}/(1 - \alpha_{22})$ and $f_{\theta 11}^N(0) - \lambda_0 = \mu_1/(1 - \alpha_{11})^3 \neq 0$, it comes $\alpha_{21} = f_{\theta 12}^N(0)(1 - \alpha_{22})/(f_{\theta 11}^N(0) - \lambda_0)$ and the second equality of Equations (5.34) implies that

$$\frac{f_{\theta \ 11}^{N}(1) - \lambda_{0}}{f_{\theta \ 11}^{N}(0) - \lambda_{0}} \frac{f_{\theta \ 12}^{N}(0)\beta_{2}(1 - \alpha_{22})}{\beta_{2}^{2}(1 - \alpha_{22})^{2} + 4\pi^{2}} = \frac{f_{\theta' \ 11}^{N}(1) - \lambda_{0}'}{f_{\theta' \ 11}^{N}(0) - \lambda_{0}'} \frac{f_{\theta' \ 12}^{N}(0)\beta_{2}'(1 - \alpha_{22}')}{\beta_{2}'^{2}(1 - \alpha_{22}')^{2} + 4\pi^{2}}.$$
(5.36)

By Equation (5.35)

$$\frac{f_{\theta \ 12}^N(0)\beta_2(1-\alpha_{22})}{\beta_2^2(1-\alpha_{22})^2+4\pi^2} = \frac{f_{\theta'\ 12}^N(0)\beta_2'(1-\alpha_{22}')}{\beta_2'^2(1-\alpha_{22}')^2+4\pi^2} \,,$$

and since $\alpha_{21} > 0$, $f^N_{\theta_{-12}}(0) \neq 0$ and Equation (5.36) leads to

$$\frac{f_{\theta \ 11}^N(1) - \lambda_0}{f_{\theta \ 11}^N(0) - \lambda_0} = \frac{f_{\theta' \ 11}^N(1) - \lambda'_0}{f_{\theta' \ 11}^N(0) - \lambda'_0}$$

Since $f_{\theta 11}^N$ is strictly decreasing, $f_{\theta 11}^N(1) - f_{\theta 11}^N(0) = f_{\theta 11}^N(1) - f_{\theta 11}^N(0) \neq 0$, and it comes $\lambda_0 = \lambda'_0$.

Now, the expression of $f_{\theta \ 11}^N$ is similar to that of the univariate spectral density f_{θ}^N in Proposition 5.3.1. Thus, following the proof in Appendix 5.B, from $f_{\theta \ 11}^N = f_{\theta' \ 11}^N$ and $\lambda_0 = \lambda'_0$ (since $\alpha_{11} > 0$ and $\alpha'_{11} > 0$) it comes:

$$\begin{cases} \mu_1 = \mu'_1 \\ \alpha_{11} = \alpha'_{11} \\ \beta_1 = \beta'_1 . \end{cases}$$

Next, from $f^N_{\theta \ 12}(0) = f^N_{\theta' \ 12}(0)$, it comes:

$$\left(f_{\theta 11}^{N}(0) - \lambda_{0}\right) \frac{\alpha_{21}}{1 - \alpha_{22}} = \left(f_{\theta' 11}^{N}(0) - \lambda_{0}'\right) \frac{\alpha_{21}'}{1 - \alpha_{22}'},$$

which implies, with $f_{\theta 11}^N(0) - \lambda_0 = f_{\theta 11}^N(0) - \lambda'_0 \neq 0$:

$$\frac{\alpha_{21}}{1 - \alpha_{22}} = \frac{\alpha'_{21}}{1 - \alpha'_{22}} \,. \tag{5.37}$$

Now, from $\lim_{\omega'\to\infty} f^N_{\theta~22}(\omega') = \lim_{\omega'\to\infty} f^N_{\theta'~22}(\omega'),$

$$\frac{\mu_1 \alpha_{21}}{(1-\alpha_{11})(1-\alpha_{22})} + \frac{\mu_2}{1-\alpha_{22}} = \frac{\mu_1 \alpha'_{21}}{(1-\alpha_{11})(1-\alpha'_{22})} + \frac{\mu'_2}{1-\alpha'_{22}}$$

and leveraging Equation (5.37), it comes:

$$\frac{\mu_2}{1 - \alpha_{22}} = \frac{\mu_2'}{1 - \alpha_{22}'} \,. \tag{5.38}$$

Moreover,

$$f_{\theta}^{N}{}_{22}(0) = \left(\frac{\mu_{1}\alpha_{21}}{(1-\alpha_{11})(1-\alpha_{22})} + \frac{\mu_{2}}{1-\alpha_{22}} + \frac{\mu_{1}\alpha_{21}^{2}}{(1-\alpha_{11})^{3}}\right)\frac{1}{(1-\alpha_{22})^{2}} + \lambda_{0}$$
$$= \left(\frac{\mu_{1}}{1-\alpha_{11}}\frac{\alpha_{21}}{1-\alpha_{22}} + \frac{\mu_{2}}{1-\alpha_{22}}\right)\frac{1}{(1-\alpha_{22})^{2}} + \frac{\mu_{1}}{(1-\alpha_{11})^{3}}\frac{\alpha_{21}^{2}}{(1-\alpha_{22})^{2}} + \lambda_{0}.$$

Thus, by Equations (5.37) and (5.38), we obtain from $f_{\theta}^{N}{}_{22}(0) = f_{\theta'}^{N}{}_{22}(0)$:

$$\frac{1}{(1-\alpha_{22})^2} = \frac{1}{(1-\alpha_{22}')^2} \,,$$

which implies

$$\alpha_{22} = \alpha'_{22} \,.$$

To conclude, from Equations (5.35), (5.38) and (5.37), $\beta_2 = \beta'_2$, $\mu_2 = \mu'_2$ and $\alpha_{21} = \alpha'_{21}$. This proves that $\mathbf{f}^N_{\theta} = \mathbf{f}^N_{\theta'} \implies \theta = \theta'$, which achieves the proof.

A numerical exploration of thinned Hawkes processes through spectral theory

Standard estimation methods for Hawkes processes with missing information are either intractable or computationally expensive. Nonetheless this is a common scenario when studying real-world data where detection procedures may miss event times, which can be modelled through the thinning alteration of point processes. We assume that we observe a thinned version of a Hawkes process and we propose a parametric estimation procedure by maximising a version of the log-likelihood from spectral analysis theory. We propose identifiability conditions for the ensuing statistical model for the classical exponential kernel functions and the performance of our estimator is studied on simulated data. Our inference method gives access to a subsampling method for our point processes via the thinning operation. We propose to use this paradigm to enhance the performance of a penalised version of the spectral estimator. Numerically, we show that, in a small-sized sample context with a short observation windows, this estimator performs better than alternative approaches.

Outline of the current chapter

6.1	Introduction	112
6.2	Mathematical setting	113
	6.2.1 The <i>p</i> -thinned point processes	113
6.3	Parametric estimation of a thinned process	114 115
	6.3.1 The spectrum of a thinned point process	$\begin{array}{c} 115 \\ 116 \end{array}$
6.4	Numerical illustrations6.4.1 Spectral estimator for missing data6.4.2 p-thinning as a subsampling method	118 118 119
6.5	Discussion	122
6. A	Proof of Proposition 6.3.1	122
6.B	Proof of Proposition 6.3.2	124

6.1 Introduction

The Hawkes process model, introduced for the first time in Hawkes (1971), is a past-dependent point process used to characterise self-exciting interactions among event times. Its applications are numerous among various fields with examples including sociology (Linderman et al. 2014), biology (Gupta et al. 2018; Lambert et al. 2018; Rizoiu et al. 2018), seismology (Ogata 1988; Ogata 1998) and finance (Bacry et al. 2013; Bacry et al. 2015; Hawkes 2018), to mention a few.

As a result, there are many efforts focused around proposing estimation methods. These include estimators obtained through likelihood maximisation (Ogata 1978; Ozaki 1979), least-squares error minimisation (Reynaud-Bouret et al. 2014; Bacry et al. 2020) and method of moments (Da Fonseca et al. 2013), in both parametric and non-parametric settings.

Most of these methods tend to study the model under the assumption that there is no missing data in the samples used for estimation. However, if this assumption does not hold, it is common for estimators to be biased. Some works in the literature consider different missing or imperfect data contexts such as binned observations in Cheysson et al. (2022), jittering (or random displacement) as exemplified in Antoniadis et al. (2006) and Bonnet et al. (2022b), superposition as in Bonnet et al. (2024).

Jittering and superposition are two of the three most common alterations on point processes, along with thinning. Thinning is more often used in simulation scenarios for point processes, with the main example being Ogata's thinning procedure (Ogata 1981) or to accelerate estimations as a method of downsampling (Li et al. 2019). Nonetheless, the use of thinning to model missing data is scarcer in the literature, which could be useful in order to account for sporadic detection errors.

A main work focusing on noised data for both point processes and proposing numerical applications to Hawkes processes can be found in Lund et al. (2000). They propose a conditional log-likelihood model for point process that have been noised through the three classic operations in the field: jittering, superposition and thinning. Their method consists in establishing an expression of such a log-likelihood when the base point process (or an estimation of it) is available. They obtain an approximation of a maximiser through some simplifications and an iterative optimisation on a small space of parameters. The main obstacle that arises in this context is that it is essential to have some information on the ground process in order to estimate the parameters associated with each noise.

In this chapter, we take inspiration from the spectral estimation methods as presented in Cheysson et al. (2022) and Bonnet et al. (2024). Spectral theory for point processes was properly introduced for the first time in Bartlett (1963), focusing around the Bartlett spectrum. Works concerning the spectral study and estimation of point processes subsequently appeared in the literature with numerous theoretical approaches in temporal and spatial contexts (Daley 1971; Tuan 1981; Mugglestone et al. 2001; Rajala et al. 2023). Practical applications are scarcer, with an application to multiple clustering models, including Hawkes processes, in Adamopoulos (1976), in Karavasilis et al. (2007) to stationary bivariate processes and in Roueff et al. (2019) to a locally stationary version of the Hawkes process.

In our work, we explicit the form of the Bartlett spectrum for a point process thinned with independent and identically distributed probabilities. In particular, we follow a similar scheme as the one presented for the superposition of point processes in Bonnet et al. (2024) to establish an estimation method by maximising the spectral log-likelihod, with a particular interest in the univariate Hawkes process with exponential kernel.

A second contribution of our work focuses on the thinning operation as a tool to provide a subsampling scheme for point processes. This approach has been explored mainly in spatial contexts (Møller et al. 2003; Moradi et al. 2019; Cronie et al. 2024) showing the advantages of considering thinning for point processes, either for improving estimations, or for proposing crossvalidation procedures in spatio-temporal contexts, as in Coeurjolly et al. (2024). Our effort in this chapter consists in illustrating its use for spectral estimators when combined with a penalised version of the log-likelihood.

After a general presentation of the mathematical setting in Section 6.2 for both thinning and spectral theory of point processes, we introduce the results concerning the spectral density function for thinned point processes in Section 6.3.1. We leverage this result in the case of a Hawkes process with exponential kernel in Section 6.3.2 along with an identifiability condition for the statistical model. Finally, we illustrate the performance of our estimation procedure in the context of missing data in Section 6.4.1, and for enhancing estimation through ℓ_2 penalisation with subsampling in Section 6.4.2

6.2 Mathematical setting

6.2.1 The *p*-thinned point processes

In this chapter, we will focus on the stationary univariate Hawkes process on the real line \mathbb{R} , noted H. The Hawkes process can be characterised by its conditional intensity function defined, for all $t \in \mathbb{R}$, by:

$$\lambda(t) = \mu + \int_{-\infty}^{t} h(t-s) H(\mathrm{d}s) = \mu + \sum_{T_k \le t} h(t-T_k) , \qquad (6.1)$$

where $\mu > 0$ is the baseline intensity, and $h: \mathbb{R} \to \mathbb{R}_{\geq 0}$ is the kernel function modelling the self-exciting behaviour of past events $(T_k)_{k \in \mathbb{Z}}$.

In order for H to be a point process (i.e. an a.s. finite measure in any bounded set B), a sufficient and necessary condition (Hawkes 1971) is that

$$||h||_1 = \int_{\mathbb{R}} h(t) \, \mathrm{d}t < 1.$$

Let $\mathcal{B}^c_{\mathbb{R}}$ denote the set of bounded Borel sets on \mathbb{R} , and let, for any $B \in \mathcal{B}^c_{\mathbb{R}}$:

$$H(B) = \sum_{k \in \mathbb{Z}} \mathbb{1}_{T_k \in B}$$

be the number of event times in B.

In this chapter, we will study a thinned version of process H defined as follows:

Definition 6.2.1. For any $p \in (0, 1]$, we denote H_p a *p*-thinning of a point process H, defined for any $B \in \mathcal{B}^c_{\mathbb{R}}$ as:

$$H_p(B) = \sum_{k \in \mathbb{Z}} \mathbb{1}_{T_k \in B} Z_k$$

where $(Z_k)_{k\in\mathbb{Z}}$ is an i.i.d. collection of Bernoulli random variables of parameter p. We refer to H_p as a p-thinned point process.

In practice, the event times of process H_p correspond to a subset of $(T_k)_{k\in\mathbb{Z}}$ where each point is erased with a random probability 1 - p. Setting p = 1 keeps all original points from process H which is not a proper thinning per se, but we include this value in order to lighten notations in the incoming sections.

6.2.2 Spectral theory for point processes

For a point process H, we define the first and second-order measures M_1 and M_2 , for any $A, B \in \mathcal{B}^c_{\mathbb{R}}$, by:

$$M_1(A) = \mathbb{E}[H(A)], \qquad M_2(A, B) = \mathbb{E}[H(A)H(B)].$$

Under stationarity conditions, it follows that (Daley et al. 2003, Proposition 8.1.I), for any $A, B \in \mathcal{B}_{\mathbb{R}}^{c}$:

$$M_1(A) = m_1 \ell_{\mathbb{R}}(A)$$

where $m_1 = \mathbb{E}[H([0,1])]$ is usually known as the average intensity of process H.

The spectral analysis approach of point processes is based on the Bartlett spectrum Γ (Bartlett 1963), a complex-valued measure on \mathbb{R} associated with the second-order moment measure M_2 . To introduce it properly, let us consider the Schwartz space S defined as:

$$\mathcal{S} = \left\{ f \in C^{\infty}, \forall k \in \{1, 2, \dots\}, \forall r \in \{1, 2, \dots\}, \sup_{x \in \mathbb{R}} \left| x^r f^{(k)}(x) \right| < \infty \right\},\$$

where C^{∞} denotes the set of infinitely differentiable functions f from \mathbb{R} to \mathbb{R} , and $f^{(k)}$ denotes the k^{th} -order derivative of f. For any $f \in S$, we define its Fourier transform \tilde{f} , for any $\omega \in \mathbb{R}$, by:

$$\tilde{f}(\omega) = \int_{\mathbb{R}} f(x) \mathrm{e}^{-2\pi \mathrm{i} x \omega} \,\mathrm{d} x \,.$$

The Bartlett spectrum of a point process H is the measure Γ such that, for any $\varphi \in S$ (Brémaud et al. 2005, Definition 2):

$$\mathbb{V}\left[\int_{\mathbb{R}}\varphi(x)\,H(\mathrm{d}x)\right] = \int_{\mathbb{R}}|\tilde{\varphi}(\omega)|^2\,\Gamma(\mathrm{d}\omega)\,. \tag{6.2}$$

Existence of such a measure is established for any stationary point process (Daley et al. 2003, Proposition 8.2.I.(a)), and by polarisation of Equation (6.2), we obtain for any $\varphi, \psi \in S$:

$$\operatorname{cov}\left(\int_{\mathbb{R}}\varphi(x)\,H(\mathrm{d}x),\int_{\mathbb{R}}\psi(x)\,H(\mathrm{d}x)\right) = \int_{\mathbb{R}}\tilde{\varphi}(\omega)\tilde{\psi}(-\omega)\,\Gamma(\mathrm{d}\omega)\,.$$
(6.3)

Whenever Γ is an absolutely continous measure, we denote $f : \mathbb{R} \to \mathbb{C}$ its Radon-Nikodym derivative, known as the spectral density of process H.

Another important quantity in the spectral theory of point processes is the periodogram $I^T \colon \mathbb{R} \to \mathbb{C}$. For a realisation $(T_k)_{k=1:N(T)}$ of a process H in a time window [0,T], the periodogram is defined, for all $\omega \in \mathbb{R}$, as:

$$I^{T}(\omega) = \frac{1}{T} \sum_{k=1}^{N(T)} \sum_{l=1}^{N(t)} e^{-2\pi i \omega (T_{k} - T_{l})}.$$

Furthermore, for any sequence $(\omega_k)_{k=1:M}$, such that $\omega_k \neq \omega_l$, for all integers $k \neq l$, the random variables $(I^T(\omega_k))_{k=1:M}$ are asymptotically independent and exponentially distributed (Tuan 1981) with respective parameter $(1/f(\omega_k))_{k=1:M}$.

In this chapter, we will work on a parametric setting, so let a statistical model \mathcal{P} for the spectral density be defined as:

$$\mathcal{P} = \{ f_{\theta} \colon \mathbb{R} \to \mathbb{C}, \theta \in \Theta \}$$
.

We can then define the spectral log-likelihood $\ell_T(\theta)$ for an observation $(T_k)_{k=1:N(T)}$ of H as:

$$\ell_T(\theta) = -\frac{1}{T} \sum_{k=1}^M \log(f_\theta(\omega_k)) + \frac{I^T(\omega_k)}{f_\theta(\omega_k)}.$$
(6.4)

We can then introduce the Whittle estimator $\hat{\theta}$ (Whittle 1952) as:

$$\hat{\theta} = \arg \max_{\theta \in \Theta} \ \ell_T(\theta)$$

6.3 Parametric estimation of a thinned process

6.3.1 The spectrum of a thinned point process

The goal of this chapter is to establish a parametric estimation method for the observation of a thinned Hawkes process. We will leverage the study of spectral quantities on marked point processes, which can be found in Brémaud et al. (2002) and Brémaud et al. (2005).

For this purpose, let us introduce an alternative way of viewing the *p*-thinning of a point process through marked point process theory (as also done in Cronie et al. (2024)). In this section, H will denote a stationary point process on \mathbb{R} with Bartlett spectrum Γ^{H} and spectral density function f^{H} .

We define the marked point process \overline{H} associated with H, with marks Z_k on a metric space \mathcal{K} as the collection of points $(T_k, Z_k)_{k \in \mathbb{Z}} \in (\mathbb{R} \times \mathcal{K})^{\mathbb{Z}}$ (see Daley et al. (2003, Chapter 6.4) for a more thorough presentation of marked point processes). The random marks Z_k are usually used to represent underlying information on the event times of a point process H, which is often referred to as the ground process. In our work, we will restrict ourselves to the case where the random variables $(Z_k)_{k \in \mathbb{Z}}$ are independent and identically distributed. In this setting, process \overline{H} is well-defined (Daley et al. 2003, 6.4.IV(a)).

We can then view a *p*-thinning of H as a marked version \overline{H} where $\mathcal{K} = \{0, 1\}$ and the $(Z_k)_{k \in \mathbb{Z}}$ are a collection of Bernoulli random variables of parameter p. This way we may define the thinned process H_p , for any $B \in \mathcal{B}^c$, as:

$$H_p(B) = \bar{H}(B \times \{1\}).$$

Under this scope, we will apply the results of Brémaud et al. (2005), that we adapt to our notations.

Theorem 6.3.1 (Brémaud et al. (2005, Theorem 2)). Let H be a stationary point process with Bartlett spectrum measure Γ and let \overline{H} be a marked version of H with i.i.d. marks Z_k with shared distribution Z on a metric space \mathcal{K} . Let φ^*, ψ^* be measurable functions from $\mathbb{R} \times \mathcal{K} \to \mathbb{R}$, such that:

•
$$\int_{\mathbb{R}} \mathbb{E}\left[|\varphi^{\star}(x,Z)|\right] dx < +\infty, \qquad \int_{\mathbb{R}} \mathbb{E}\left[|\psi^{\star}(x,Z)|\right] dx < +\infty.$$

•
$$\int_{\mathbb{R}} \mathbb{E}\left[\varphi^{\star}(x,Z)^{2}\right] dx < +\infty, \qquad \int_{\mathbb{R}} \mathbb{E}\left[\psi^{\star}(x,Z)^{2}\right] dx < +\infty.$$

• By denoting $\bar{\varphi}: x \to \mathbb{E}[\varphi^*(x, Z)]$ and $\bar{\psi}: x \to \mathbb{E}[\psi^*(x, Z)]$,

 $\bar{\varphi}, \bar{\psi} \in \mathcal{S}$.

Then, it follows that:

$$\operatorname{cov}\left(\sum_{k\in\mathbb{Z}}\varphi^{\star}(T_{k},Z_{k}),\sum_{k\in\mathbb{Z}}\psi^{\star}(T_{k},Z_{k})\right) = \int_{\mathbb{R}}\tilde{\varphi}(\omega)\tilde{\psi}(-\omega)\,\Gamma^{H}(\mathrm{d}\omega) + \int_{\mathbb{R}}\operatorname{cov}\left(\tilde{\varphi}^{\star}(\omega,Z),\tilde{\psi}^{\star}(-\omega,Z)\right)M_{1}(\mathrm{d}\omega),$$

$$(6.5)$$

where, for any $\omega \in \mathbb{R}$, $\tilde{\varphi}^{\star}(\omega, Z)$ (resp. $\tilde{\psi}^{\star}(\omega, Z)$) denotes the Fourier transform of the function $x \to \varphi^{\star}(x, Z)$ (resp. $x \to \psi^{\star}(x, Z)$).

The utility of this result lies on the link that it establishes between the covariance of a marked point process \overline{H} and the Bartlett spectrum of its ground process H. This allows to establish an expression of the spectral density of the *p*-thinning of a process H, as presented in the following proposition:

Proposition 6.3.1. Let H be a stationary point process admitting a Bartlett spectrum Γ^H defined as in Equation (6.2). We assume that Γ^H is absolutely continuous and we note f^H the spectral density function. Let m_1 be the average intensity of H.

For any $p \in (0,1]$, let H_p be a p-thinning of H as explicited in Definition 6.2.1. Then, H_p admits a spectral density function, denoted f^{H_p} , such that for any $\omega \in \mathbb{R}$:

$$f^{H_p}(\omega) = p^2 f^H(\omega) + p(1-p)m_1.$$
(6.6)

Proof. The proof is given in Appendix 6.A

Let us remark that Equation (6.6) can be found in Daley et al. (2003, Equation 8.3.5), where they obtain this result by term identification of a bivariate point process. The proof presented in this chapter is an alternative way of establishing this expression, illustrating the usefulness of leveraging the spectral theory of point processes.

6.3.2 The *p*-thinned Hawkes process estimation and the exponential kernel

For the rest of this chapter, H will denote a stationary Hawkes process defined by the intensity λ as in Equation (6.1).

As shown in Hawkes (1971), the spectral density of a Hawkes process reads, for all $\omega \in \mathbb{R}$:

$$f^{H}(\omega) = \frac{\mu}{(1 - \|h\|_{1})|1 - \tilde{h}(\omega)|^{2}}$$

Corollary 6.3.1, presents the spectral density of the *p*-thinned univariate Hawkes process:

Corollary 6.3.1. Let H be a stationary univariate Hawkes process with baseline intensity $\mu > 0$, and kernel function $h: \mathbb{R} \to \mathbb{R}_{>0}$, as defined by Equation (6.1). Let $p \in (0, 1]$ and H_p a p-thinning of H. Then the spectral density f^{H_p} of H_p is:

$$\forall \omega \in \mathbb{R}, \quad f^{H_p}(\omega) = p^2 \frac{\mu}{(1 - \|h\|_1)|1 - \tilde{h}(\omega)|^2} + p(1 - p) \frac{\mu}{1 - \|h\|_1}.$$
(6.7)

Proof. The proof is direct by applying Proposition 6.3.1 and with the expression of the Hawkes process spectral density and m_1 the mean number of points of a Hawkes process given by $m_1 = \mu/(1 - \|h\|_1)$.

We can define the statistical model:

$$\mathcal{P} = \{ f_{\theta} \colon \mathbb{R} \to \mathbb{C}, \theta = (\mu, \gamma, p) \in \Theta \}$$

where γ is a parameter of the interaction function h. From now on, we will drop the superscript on the spectral density function as specifying θ removes any ambiguity concerning the value of pand whether the point process is thinned or not. It is then possible to estimate θ by maximising the spectral log-likelihood (Equation (6.4)).

Let us now focus on the exponential interaction function, defined as:

$$\forall t \in \mathbb{R}, \quad h(t) = \alpha \beta e^{-\beta t} \mathbb{1}_{t>0},$$

where $\alpha \in (0, 1)$ and $\beta > 0$.

The Fourier transform of h is explicit and reads:

$$\tilde{h}(\omega) = \frac{\alpha\beta}{\beta + 2\pi \mathrm{i}\omega} \,,$$

and so Equation (6.7) reduces to:

$$\forall \omega \in \mathbb{R}, \quad f_{\theta}(\omega) = \frac{\mu p}{1 - \alpha} \left(1 + p \frac{\beta^2 \alpha (2 - \alpha)}{\beta^2 (1 - \alpha)^2 + 4\pi^2 \omega^2} \right).$$
(6.8)

In this context, the statistical model of f_{θ} is:

$$\mathcal{Q} = \{ f_{\theta} \colon \mathbb{R} \to \mathbb{C}, \theta = (\mu, \alpha, \beta, p) \in \mathbb{R}_{>0} \times (0, 1) \times \mathbb{R}_{>0} \times (0, 1] \} .$$

Similarly to the superposition case shown in Bonnet et al. (2024, Proposition 3.2), model Q is not identifiable in the general setting but identifiability can be retrieved, as shown in Proposition 6.3.2, as long as one parameter of the model is fixed.

Proposition 6.3.2. The model Q is identifiable if and only if one of the parameters in the 4-uplet (μ, α, β, p) is fixed.

In particular, for any admissible parameter $\theta = (\mu, \alpha, \beta, p)$, for any $\kappa \in (0, 1/p]$, let:

$$\begin{cases} \mu' = \frac{\mu}{\kappa(1-\alpha)\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2}-1\right)}} \\ \alpha' = 1 - \frac{1}{\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2}-1\right)}} \\ \beta' = \beta(1-\alpha)\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2}-1\right)} \\ p' = \kappa p \,. \end{cases}$$
(6.9)

Then, $\theta' = (\mu', \alpha', \beta', p')$ is an admissible parameter such that $f_{\theta} = f_{\theta'}$.

Proof. The proof is given in Appendix 6.B.

The non-identifiability of the full model (four unknown parameters) limits the implementation of an estimation method for $\theta \in \Theta$, but this problem is avoided as long as one of the parameters is known beforehand. In this work we will focus on the scenario where the value of p is known in advance, and so the previous result ensures that, for any $p^* \in (0, 1]$, the reduced model:

$$\mathcal{Q}_p = \{ f_\theta \colon \mathbb{R} \to \mathbb{C}, \theta = (\mu, \alpha, \beta, p) \in \Theta = \mathbb{R}_{>0} \times (0, 1) \times \mathbb{R}_{>0} \times (0, 1], p = p^* \} ,$$

is identifiable. We can then define an estimator $\hat{\theta} = (\hat{\mu}, \hat{\alpha}, \hat{\beta}, \hat{p})$ of θ with $\hat{p} = p^*$, as presented at the end of Section 6.2.2, as:

$$\hat{\theta} = \arg\max_{\theta\in\Theta} \ell_T(\theta) = \arg\max_{\theta\in\Theta} \left(-\frac{1}{T} \sum_{k=1}^M \log(f_\theta(\omega_k)) + \frac{I^T(\omega_k)}{f_\theta(\omega_k)} \right) \,.$$

6.4 Numerical illustrations

In this section, we will study the estimation of p-thinned Hawkes processes obtained by maximising the spectral lof-likelihood defined in Equation (6.4) under the mathematical model \mathcal{Q}_p , and unless said otherwise, p will be fixed to the true value p^* used for thinning.

Simulation is done by leveraging the cluster and branching representation of self-exciting Hawkes processes (Hawkes et al. 1974), which is explicited in Møller et al. (2005, Algorithm 2). In order to approximate the stationarity condition of the Hawkes process, for any simulation window [0, T] considered in our study, we will initially simulate the process in [-100, T] to avoid edge effects.

Our numerical procedure is implemented in Python and optimisation of all considered functions in this section will be done via the minimize method with the L-BFGS-B solver (Byrd et al. 1995).

To evaluate the performance of our estimator $\hat{\theta}$ against any set of parameters θ , we will consider its relative ℓ_2 error defined as:

$$\frac{\|\hat{\theta} - \theta\|_2}{\|\theta\|_2}$$

6.4.1 Spectral estimator for missing data

In this subsection, we will study the performance of estimator $\hat{\theta}$ for different levels of p. We simulate 100 different Hawkes processes (H^1, \ldots, H^{100}) with exponential kernel with parameters:

$$\mu = 1.25, \quad \alpha = 0.5, \quad \beta = 1.5,$$

and the following grid of values for thinning:

$$p \in \{0.1, 0.2, \dots, 0.9\}.$$

We perform then a *p*-thinning of each simulation obtaining the thinned sample $(H_p^1, \ldots, H_p^{100})$. For each level *p*, we will consider an observation window $[0, T_p]$ with T_p chosen so that the average number of points after thinning is constant equal to 5000. By remarking that:

$$\mathbb{E}[H_p([0,T_p])] = p\mathbb{E}[H([0,T_p])] = \frac{p\mu}{1-\alpha}T_p,$$

it follows that $T_p = 5000(1 - \alpha)/(p\mu)$. This ensures that for each scenario, all estimations take into account the same amount of information on average.

We will consider two different estimators in order to show the advantage of taking thinning into account. For each H_p^i , the first estimator $\hat{\theta}_p$ is obtained by estimating in the correct model

 Q_p and we will compare it to the estimator $\hat{\theta}_1$ obtained by maximising Equation (6.4) for p = 1. Estimator $\hat{\theta}_1$ corresponds to making the assumption that observations $(H_p^1, \ldots, H_p^{100})$ are univariate Hawkes processes that have not been thinned.

Figure 6.1 represents the boxplots of relative square errors for each parameter estimation in $\hat{\theta}_p$ and $\hat{\theta}_1$ in logarithmic scale. As expected, $\hat{\theta}_p$ performs significantly better than $\hat{\theta}_1$, in particular for higher levels of thinning (lower values of p). We can see that estimations are especially improved when estimating μ and α , which are the two parameters controlling the average number of points of H, that we recall here:

$$\mathbb{E}[H([0,T])] = m_1 T_p = \frac{\mu T_p}{1-\alpha}$$

Let us remark that the performance of $\hat{\theta}$ is fairly consistent accross all considered values of p. This is particularly encouraging as the more points are erased, interaction between points are attenuated and so the optimisation task is more difficult.



Figure 6.1: Boxplots of ℓ_2 relative error in log-scale with respect to level of thinning p. Estimations for each parameter on the correct thinned model (red boxes) and on the model without accounting for thinning (blue boxes).

6.4.2 *p*-thinning as a subsampling method

In this section, we assume that we are provided with a single and short observation H of a Hawkes process for the same parameters:

$$\mu = 1.25, \quad \alpha = 0.5, \quad \beta = 1.5,$$

in an observation window of [0, 50]. The lack of multiple repetitions and the small window size tend to negatively affect estimation procedures for point processes.

A common practice to improve estimations in such contexts is to consider a penalised version of the to-be-optimised function, in our case, the spectral log-likelihood (Equation (6.4)). We define then the ℓ_2 -penalised spectral log-likelihood, for any penalisation parameter $L \ge 0$, as:

$$\ell_T(\theta) - L \|\theta\|_2 \,. \tag{6.10}$$

A penalised estimator can then by obtained by maximising this quantity. In this penalised setting, we will illustrate how combining the ℓ_2 penalisation with a subsampling procedure by thinning provides better results than other approaches.

Benchmark estimators. As our work in this chapter concerns mainly the study of spectral approaches, we will solely take into consideration estimators obtained through such means. We will compare then four estimators:

- $\hat{\theta}$: the estimator obtained by maximising the non-penalised spectral log-likelihood (Equation (6.4)) on the single observation H.
- $\hat{\theta}^L$: the estimator obtained by maximizing the penalised spectral log-likelihood (Equation (6.10)) on the single observation H.
- $\hat{\theta}_{partition}^{L}$: we partition the observation window [0,T] in I equally sized intervals:

$$\left[\frac{iT}{I}, \frac{(i+1)T}{I}\right] \,,$$

for $i \in \{0, \ldots, I-1\}$. For each window, we obtain a penalised estimator by maximising Equation (6.10), and finally we obtain $\hat{\theta}_{partition}^L$ by averaging all estimations. This is similar to subsampling by means of covering regions (Possolo 1991; Politis et al. 1999; Guan et al. 2007).

• $\hat{\theta}_{thinning}^L$: we perform S different thinning procedures for a fixed $p \in (0,1)$ providing the p-thinned observations H_1, \ldots, H_S of H. For each H_s , we obtain a penalised estimation by maximising Equation (6.10) on the statistical model \mathcal{Q}_p and we obtain $\hat{\theta}_{thinning}^L$ by averaging all estimations.

 $\hat{\theta}_{partition}^{L}$ corresponds to a common practice in point processes theory used to split an observation into multiple samples, which is particularly useful for large values of T.

Our proposed estimator $\hat{\theta}_{thinning}^L$ can be seen as an estimator obtained through a subsampling procedure, similar to the Bernoulli subsampling scheme (see for example Särndal et al. (2003, Chapter 3.2)).

Choice of hyperparameters. We focus on studying the efficiency of all four methods and so we consider the following grids for the hyper-parameters I, p and L:

$$I \in \{2, 3, 4, 5\}, p \in \{0.1, \dots, 0.9\}, L \in \{10^{-6}, 10^{-5}, \dots, 10^{2}\}.$$

Concerning the subsampling parameter of $\hat{\theta}_{thinning}^L$, we fix it at S = 3 but we remark that all presented results are consistent for other values of S. Proposing a model selection procedure to choose the optimal hyper-parameters I, p, L is out of the scope of this work, so we solely present the best estimators obtained through each method in the sense of the relative ℓ_2 error with respect to the true parameters of the model.

Results. Figure 6.2 represents the ℓ_2 relative error for each estimator over 1000 different simulations. We ordered all curves so that the error of our proposed estimator $\hat{\theta}_{thinning}^L$ is shown in ascending order. We can initially observe that the non-penalised estimations $\hat{\theta}$ often attain high ℓ_2 relative errors with high variability, which are greatly improved by the penalised methods. Overall, our estimations $\hat{\theta}_{thinning}^L$ present lower errors than the other estimations (dashed lines). This can be confirmed in Table 6.1 displaying the proportion of times that each estimator

This can be confirmed in Table 6.1 displaying the proportion of times that each estimator outperforms the others. The non-penalised method provides by far the weakest estimator with a notably high MSRE when estimating β , which is consistently known as the hardest parameter



Figure 6.2: ℓ_2 relative error of all four estimators without penalisation (blue), with penalisation (orange), with penalisation and subsampling by partition (green) and our proposed method with penalisation and subsampling by thinning (red). 1000 independent samples are shown and samples are sorted so to display the error of $\hat{\theta}_{thinning}^L$ in ascending order.

to estimate in other parametric inference settings (see in Lemonnier et al. (2014) the discussion on the non-concavity of the log-likelihood). The three penalised methods greatly improve the estimations, in particular for β , significantly reducing MSRE all around. We notice that the lowest MSRE are obtained by our thinning method, with it being in most scenarios the best method all around. This is a testimony of the advantages of the subsampling and penalisation procedure that we propose in this work.

Estimator	$\hat{\mu}$	\hat{lpha}	$\stackrel{\text{MSRE}}{\hat{\beta}}$	$\hat{ heta}$	% best
$\hat{ heta}$	0.18	0.13	5.14×10^2	2.85×10^2	1%
$\hat{ heta}^L$	0.12	0.09	0.13	0.12	6.7%
$\hat{\theta}_{partition}^{L}$	0.08	0.07	0.04	0.05	22.3%
$\hat{ heta}^L_{thinning}$	0.03	0.04	0.02	0.02	70%

Table 6.1: Mean square relative error (MSRE) of (μ, α, β) along with the MSRE of θ . Last column shows the proportion of times that each estimator achieves the lower relative ℓ_2 error accross all 1000 simulations.

6.5 Discussion

In this chapter, we leverage the spectral theory of point processes to present a parametric estimation procedure for thinned versions of univariate Hawkes processes. A first motivation for studying this model concerns the study of imperfect data and we showcase the efficiency of the spectral analysis to tackle such problems. This approach does not come without some issues, as we show that the underlying spectral density model is not identifiable without additional information on the parameter to be estimated. However, we highlight the efficiency of our estimation method once that identifiability is recovered, which is a step towards improving estimation in situations with potential missing event times. Our second contribution is to make use of thinning as a tool to perform subsampling in scenarios with short observations of processes. By introducing a penalised version of the spectral log-likelihood and combining this with subsampling, we demonstrate that we are able to significantly improve estimations. This is encouraging for real-world applications, such as in health or biology, where acquiring sizeable quantities of data can be difficult.

The main extension of our work would be to adapt our methods to higher dimensional data in order to consider multiple inter-connected phenomena, for instance in epidemiology, social interaction or neurobiology. In particular, our idea of introducing a subsampling approach for penalised optimisation methods is highly motivated by the high interest in Lasso approaches for multivate point processes (Reynaud-Bouret et al. 2013; Hansen et al. 2015; Bacry et al. 2020). This kind of regularisation is used in order to determine a sparse matrix of interaction for highdimensional networks where only few connections between individuals exist. As these approaches tend to require a high amount of information or samples in order to perform well, adapting our subsampling scheme would make their use more accessible. A last avenue of exploration would be to propose a model selection procedure adapted to spectral approaches in order to choose the hyper-parameters of the penalised and subsampling methods in practical contexts where information on the real parameters is unavailable.

6.A Proof of Proposition 6.3.1

Let H_p be a *p*-thinning of a stationary point process H admitting a Bartlett spectrum Γ^H and a spectral density function f^H . The stationarity of H_p is given by the fact that the variables $(Z_k)_{k\in\mathbb{Z}}$ are i.i.d. and so for any integer r, for any $B_1 \ldots, B_r \in \mathcal{B}^c$, the random vector $(N_p(B_k)_{k=1:r})$ has the same distribution as $(N_p(B_k+t)_{k=1:r})$, for any $t \in \mathbb{R}$. We will then denote Γ^{H_p} and f^{H_p} respectively the Bartlett spectrum and spectral density function of H_p

In order to establish Equation (6.6), we will leverage Theorem 6.3.1. We consider then the marked version \overline{H} of H where the marks Z_k are i.i.d. Bernoulli distributions with common probability p.

Let $\varphi, \psi \in \mathcal{S}$, we define, for all $x, z \in \mathbb{R} \times \{0, 1\}$, the functions:

$$\varphi^{\star}(x,z) = \varphi(x)z, \qquad \psi^{\star}(x,z) = \psi(x)z.$$

Let us verify that these functions satisfy the conditions of Theorem 6.3.1. Without loss of generality, we will work uniquely with φ^* , as the arguments are exactly the same for ψ^* .

For any $x \in \mathbb{R}$, $\varphi^*(x, Z) = \varphi(x)Z$ for Z a Bernoulli distribution of parameter p. It follows that $\varphi^*(x, Z)$ admits a first and second order moment, and as $\varphi \in S$, $\varphi(x)Z$ is integrable and

twice integrable, which shows that:

$$\int_{\mathbb{R}} \mathbb{E}\left[|\varphi^{\star}(x, Z)| \right] \mathrm{d}x < +\infty, \qquad \int_{\mathbb{R}} \mathbb{E}\left[\varphi^{\star}(x, Z)^{2} \right] \mathrm{d}x < +\infty$$

Furthermore, for any $x \in \mathbb{R}$, $\bar{\varphi}(x) = \varphi(x)\mathbb{E}[Z] = p\varphi(x)$, and so, as the Schwartz space is closed under scalar multiplication, it follows that $\bar{\varphi} \in S$.

We can then apply Equation (6.5) to our marked process \overline{H} . For this, let us notice that:

$$\sum_{k\in\mathbb{Z}}\varphi^{\star}(T_k,Z_k)=\sum_{k\in\mathbb{Z}}\varphi(T_k)Z_k=\int_{\mathbb{R}}\varphi(t)H_p(\mathrm{d}t)\,,$$

with the same expression holding for ψ^* and ψ . So, the left-hand side of Equation (6.5) reads:

$$\operatorname{cov}\left(\sum_{k\in\mathbb{Z}}\varphi^{\star}(T_{k},Z_{k}),\sum_{k\in\mathbb{Z}}\psi^{\star}(T_{k},Z_{k})\right) = \operatorname{cov}\left(\int_{\mathbb{R}}\varphi(t)H_{p}(\mathrm{d}t),\int_{\mathbb{R}}\psi(t)H_{p}(\mathrm{d}t)\right)$$
$$=\int_{\mathbb{R}}\tilde{\varphi}(\omega)\tilde{\psi}(-\omega)\,\Gamma^{H_{p}}(\mathrm{d}\omega)\,,\tag{6.11}$$

where the last equality comes from Equation (6.3).

For the left-hand side, let us remark that $\bar{\varphi}(x) = p\varphi(x)$ for any $x \in \mathbb{R}$ and so, for any $\omega \in \mathbb{R}$,

$$\tilde{\bar{\varphi}}(\omega) = p\tilde{\varphi}(\omega) \,,$$

and,

$$\tilde{\phi}^{\star}(\omega, Z) = \tilde{\phi}(\omega)Z$$

The right-hand side of Equation (6.5) becomes:

$$\int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \Gamma^{H}(\mathrm{d}\omega) + \int_{\mathbb{R}} \operatorname{cov} \left(\tilde{\varphi}^{\star}(\omega, Z), \tilde{\psi}^{\star}(-\omega, Z) M_{1}(\mathrm{d}\omega) \right) \\
= \int_{\mathbb{R}} p^{2} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \Gamma^{H}(\mathrm{d}\omega) + \int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(\omega) \operatorname{cov} (Z, Z) M_{1}(\mathrm{d}\omega) \\
= \int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \left(p^{2} \Gamma^{H}(\mathrm{d}\omega) + p(1-p) M_{1}(\mathrm{d}\omega) \right).$$
(6.12)

By combining both sides (Equations (6.11) and (6.12)) it follows that, for any $\varphi, \psi \in S$:

$$\int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \Gamma^{H_p}(\mathrm{d}\omega) = \int_{\mathbb{R}} \tilde{\varphi}(\omega) \tilde{\psi}(-\omega) \left(p^2 \Gamma^H(\mathrm{d}\omega) + p(1-p) M_1(\mathrm{d}\omega) \right).$$

As this equality holds for any functions in the Schwartz space, by duality of the Fourier transform (Pinsky 2008), then,

$$\Gamma_p = p^2 \Gamma + p(1-p)M_1 \,,$$

and as $M_1 = m_1 \ell_{\mathbb{R}}$,

$$f_p(\omega) = p^2 f(\omega) + p(1-p)m_1 \,$$

which achieves the proof.

6.B Proof of Proposition 6.3.2

Let $\theta = (\mu, \alpha, \beta, p)$ and $\theta' = (\mu', \alpha', \beta', p')$ be two admissible parameters for model Q.

In order to prove that model Q is identifiable if and only if one of four parameters is fixed, we will begin by retrieving the system of Equations (6.9) and verify that it defines an admissible parameter for Q. Subsequently we will verify that by fixing one parameter, we retrieve the identifiability of the model.

We begin by assuming that $f_{\theta} = f_{\theta'}$. By Equation (6.8), this equality reads:

$$\forall \omega \in \mathbb{R}, \quad \frac{\mu p}{1 - \alpha} \left(1 + p \frac{\beta^2 \alpha (2 - \alpha)}{\beta^2 (1 - \alpha)^2 + 4\pi^2 \omega^2} \right) = \frac{\mu' p'}{1 - \alpha'} \left(1 + p' \frac{\beta'^2 \alpha' (2 - \alpha')}{\beta'^2 (1 - \alpha')^2 + 4\pi^2 \omega^2} \right)$$

Let us remark that a sufficient condition for both sides to be equal, for all $\omega \in \mathbb{R}$, is that the following system of equations is verified:

$$\begin{cases} \frac{\mu p}{1-\alpha} = \frac{\mu' p'}{1-\alpha'}\\ p\beta^2 \alpha (2-\alpha) = p'\beta'^2 \alpha' (2-\alpha')\\ \beta (1-\alpha) = \beta' (1-\alpha'), \end{cases}$$
(6.13)

as this would mean that the three constants (w.r.t. ω) in both sides are equal. In fact this is also a necessary condition as the three equalities can be retrieved for f(0), $\lim_{\omega \to +\infty} f(\omega)$ and f(1) and combining the equations.

Let $\kappa \in (0, 1/p)$ and $p' = \kappa p$, which reduces the system of Equations (6.13) to:

$$\int \frac{\mu}{1-\alpha} = \frac{\mu'\kappa}{1-\alpha'} \tag{6.14a}$$

$$\beta^2 \alpha (2 - \alpha) = \kappa {\beta'}^2 \alpha' (2 - \alpha') \tag{6.14b}$$

$$\int \beta(1-\alpha) = \beta'(1-\alpha'), \qquad (6.14c)$$

As $\alpha < 1$, $\alpha' < 1$, $\beta > 0$ and $\beta' > 0$, Equation (6.14c) can be expressed as:

$${\beta'}^2 = \frac{\beta^2 (1-\alpha)^2}{(1-\alpha')^2} \,,$$

and by replacing ${\beta'}^2$ in Equation (6.14b), we obtain:

$$\frac{\alpha(2-\alpha)}{(1-\alpha)^2} = \kappa \frac{\alpha'(2-\alpha')}{(1-\alpha')^2}$$

$$\iff \frac{1-(1-\alpha)^2}{(1-\alpha)^2} = \kappa \frac{1-(1-\alpha')^2}{(1-\alpha')^2}$$

$$\iff 1 + \frac{1}{\kappa} \left(\frac{1}{(1-\alpha)^2} - 1\right) = \frac{1}{(1-\alpha')^2} ,$$
(6.15)

and as $\alpha \in (0, 1)$, the left-side term is positive and so:

$$\alpha' = 1 - \frac{1}{\sqrt{1 + \frac{1}{\kappa} \left(\frac{1}{(1-\alpha)^2} - 1\right)}}$$
We can then obtain the explicit expressions of μ' and β' with Equations (6.14a) and (6.14c) providing the following system:

$$\begin{cases} \mu' = \frac{\mu(1-\alpha')}{\kappa(1-\alpha)} \\ \alpha' = 1 - \frac{1}{\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2} - 1\right)}} \\ \beta' = \beta \frac{1-\alpha}{1-\alpha'} \\ p' = \kappa p \end{cases} \iff \begin{cases} \mu' = \frac{\mu}{\kappa(1-\alpha)\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2} - 1\right)}} \\ \alpha' = 1 - \frac{1}{\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2} - 1\right)}} \\ \beta' = \beta(1-\alpha)\sqrt{1+\frac{1}{\kappa}\left(\frac{1}{(1-\alpha)^2} - 1\right)} \\ \beta' = \kappa p . \end{cases}$$
(6.16)

To verify that, for any $\kappa \in (0, 1/p)$, parameter $\theta' = (\mu', \alpha', \beta', p')$ is an admissible parameter, we have to make sure that $\theta' \in \Theta = \mathbb{R}_{>0} \times (0, 1) \times \mathbb{R}_{>0} \times (0, 1)$. Let $\theta = (\mu, \alpha, \beta, p) \in \Theta$, $\kappa \in (0, 1/p)$ and θ' defined by Equations (6.16). For such a κ , $p' \in (0, 1)$ is immediately verified. The part of the conditions are verified if and only if.

The rest of the conditions are verified if and only if:

$$\sqrt{1 + \frac{1}{\kappa} \left(\frac{1}{(1 - \alpha)^2} - 1\right)} > 1 \iff \frac{1}{(1 - \alpha)^2} - 1 > 0,$$

This is immediate as $\alpha \in (0, 1)$ and so $\theta' \in \Theta$. From Equations (6.16) we can see that $\theta' = \theta$ and we have proven that $f_{\theta} = f_{\theta'}$, so model Q is not identifiable.

Lastly, let us show that if any of the four parameters is known then the model defined by the remaining triplet is identifiable. By considering the previously established equations (see Equations (6.15) and (6.16)), we know that for $\theta = (\mu, \alpha, \beta, p) \in \Theta$ and $\theta' = (\mu', \alpha', \beta', p') \in \Theta$:

$$f_{\theta} = f_{\theta'} \quad \iff \quad \begin{cases} \mu' = \frac{\mu(1-\alpha')}{\kappa(1-\alpha)} \\ \frac{\alpha(2-\alpha)}{(1-\alpha)^2} = \kappa \frac{\alpha'(2-\alpha')}{(1-\alpha')^2} \\ \beta' = \beta \frac{1-\alpha}{1-\alpha'} \\ p' = \kappa p \end{cases}$$
(6.17)

• If $\mu = \mu'$, the system of Equations (6.17) reduces to:

$$\begin{cases} \kappa = \frac{(1-\alpha')}{(1-\alpha)} \\ \kappa \frac{\alpha'(2-\alpha')}{(1-\alpha')^2} = \frac{\alpha(2-\alpha)}{(1-\alpha)^2} \\ \beta' = \frac{\beta}{\kappa} \\ p' = \kappa p \end{cases} \longleftrightarrow \begin{cases} \kappa = \frac{(1-\alpha')}{(1-\alpha)} \\ \frac{\alpha'(2-\alpha')}{(1-\alpha')} = \frac{\alpha(2-\alpha)}{(1-\alpha)} \\ \beta' = \frac{\beta}{\kappa} \\ p' = \kappa p \end{cases}$$

As $\alpha \in (0,1)$ and $\alpha' \in (0,1)$, the second equation implies that $\alpha = \alpha'$ and so $\kappa = 1$ and $\beta' = \beta$.

- If $\alpha = \alpha'$, the second equation in (6.17) implies directly that $\kappa = 1$ and so all other equalities follow.
- If $\beta = \beta'$, the third equation in (6.17) implies that $\alpha = \alpha'$ and by the previous point all other equalities hold.

• If p = p', $\kappa = 1$ and the rest of equalities are verified. This shows that whenever one of the parameters is fixed, $f_{\theta} = f_{\theta'}$ implies $\theta = \theta'$. This achieves the proof.

Discussion

The Hawkes process model is used accross a vast multitude of application fields, and so establishing inference procedures to model more complex phenomena is undeniably important. Our work provided novel statistical tools for the estimation of Hawkes processes within the frequentist parametric framework. Our effort was centered around establishing methodological procedures for two different submodels, with code implementations of our estimators and numerical illustrations of their efficiency.

The first part of our work focused on the study of inhibiting interactions for Hawkes processes by establishing the maximum likelihood estimation procedure. In Chapter 3, we implemented this method under the assumption of monotony of the kernel function, which allowed us to derive a closed-form expression of the log-likelihood. In Chapter 4, we extended our previous work to the multivariate setting for exponentially-shaped interactions which revealed the challenges of establishing the identifiability of our model. This issue arised primarily from our use of the positive part function in the non-linear Hawkes process model as, by taking null values, it hinders the accurate identification of interaction parameters. On the other hand, the choice of this function helps modelling a closer behaviour to the classical linear Hawkes model and simplifies the computation of the intensity function. Prioritising positive non-linear functions is a way to circumvent this problem, as shown in the Bayesian context in Sulem et al. (2024), for a shifted version of the positive part function. By adapting our procedure to an approximation of the positive part function, for example through the softplus function, the resulting method could be a satisfying compromise between model fidelity and theoretical guarantees.

Numerically, we illustrated the improvements in our estimations by incorporating a support inference step for the interaction matrix, as evidenced by our goodness-of-fit hypothesis testing. Our methods relied on data-based paradigms which are highly dependant on sample size. In practical applications, such as medical data analysis with limited observations, it is important to propose more robust methods. Future research could explore Lasso penalisation procedures, in the same vein as the work of Bacry et al. (2020), where both an ℓ_1 and trace-norm penalisations are tuned by the means of data-driven weights and that can benefit from cross-validation procedures. We believe this is a promising research field with novel contributions as shown in the work of Lotz (2024) on sparsity tests for Hawkes processes.

In our application to the study of neuronal activity, the results exhibited a clear self-interaction behaviour for each studied neuron. This is consistent with the known behaviour of neurons as they present a post-activation refractory period, reflecting an incapacity to reactivate after a synaptic impulse is produced. The fact that our estimations are not consistently self-inhibitory is most likely due to the choice of a monotone kernel in the form of the exponential function. In reality, when membrane potential is measured, the refractory effect is delayed which is better represented by a non-monotone function suggesting that extending our work for more complex interaction functions is a clear path to obtaining more faithful models.

The second part of our work examined noisy observations of linear Hawkes processes. In Chapter 5, we modelled the presence of exogeneous points by superposing a homogeneous Poisson process to the Hawkes process. Chapter 6 adresses the absence of points through a thinning representation of the original point process. Our spectral approach, utilising the Bartlett spectrum and the periodogram, enabled us to establish an estimation procedure by optimising a spectral version of the log-likelihood. The main difficulty encountered was about identifiability of the statistical models, which derived directly from the definition of the spectral density function. To partially solve this problem, we provided conditions on the kernel functions that allowed us to recover identifiability of our model. Our method will greatly benefit from more general results, particularly for higher dimensional processes with more complex interactions. Another possible solution could be found in the recent works by Roueff et al. (2016) and Roueff et al. (2019), that establish an equivalent expression of the Bartlett spectrum by relaxing the stationarity condition to a local stationarity. This could help to incorporate more general shapes of noise which in turn could help to better identify the noise dynamics.

Our numerical illustrations on synthetic data showed the efficiency of our approach to perform this difficult estimation task and the next step is to apply our results to real-world data. This can help to provide a better insight in contexts where measurement errors are more likely to appear. In particular, it can be interesting to combine both models studied in Chapters 5 and 6in order to study the spread of illnesses through screening. In this context, the presence of false positives and false negatives can be modelled by the superposition and thinning operations and so our model can clearly help to obtain better interpretations of disease transmission dynamics.

Finally, in the context of subsampling through thinning, a significant issue concerns the lack of evaluation methods for spectral estimators in unsupervised settings. As suggested in the recent works by Cronie et al. (2024) and Coeurjolly et al. (2024), subsampling paradigms can greatly improve the estimation procedures and allow for more complex methods as cross-validation paradigms to be available for the study of point processes. Establishing asymptotic results for our estimators could partially address this issue by providing a way of establishing asymptotic confidence intervals. Encouragingly, there has been recent advances concerning these guarantees for the periodogram in the works by Rajala et al. (2023) and Yang et al. (2024), which pave the way for the study of the periodogram.

Bibliography

- Achab, M., E. Bacry, S. Gaïffas, I. Mastromatteo, and J-F. Muzy (2016). "Uncovering Causality from Multivariate Hawkes Integrated Cumulants". In: *Journal of Machine Learning Research* 18, 192:1–192:28.
- Adamopoulos, L. (1976). "Cluster models for earthquakes: Regional comparisons". In: Journal of the International Association for Mathematical Geology 8.4, pp. 463–475.
- Antoniadis, A. and J. Bigot (2006). "Poisson inverse problems". In: Annals of Statistics 34.5, pp. 2132–2158.
- Baccelli, F., B. Błaszczyszyn, and M. Karray (Jan. 2020). Random Measures, Point Processes, and Stochastic Geometry.
- Bacry, E, S. Delattre, M. Hoffmann, and J.F. Muzy (2013). "Scaling limits for Hawkes processes and application to financial statistics". In: *Stochastic Processes and Applications* 123, pp. 2475–2499.
- Bacry, E., M. Bompaire, P. Deegan, S. Gaïffas, and S.V. Poulsen (2018). "tick: a Python Library for Statistical Learning, with an emphasis on Hawkes Processes and Time-Dependent Models". In: Journal of Machine Learning Research 18.1, pp. 7937–7941.
- Bacry, E., M. Bompaire, S. Gaïffas, and J.F. Muzy (2020). "Sparse and low-rank multivariate Hawkes processes". In: *Journal of Machine Learning Research* 21.1, pp. 1–32.
- Bacry, E., I. Mastromatteo, and J-F. Muzy (2015). "Hawkes Processes in Finance". In: Market Microstructure and Liquidity 01.01, p. 1550005.
- Bacry, E. and J.F. Muzy (2016). "First- and Second-Order Statistics Characterization of Hawkes Processes and Non-Parametric Estimation". In: *IEEE Transactions on Information Theory* 62.4, pp. 2184–2202.
- Baddeley, A. (Jan. 2006). "Spatial Point Processes and their Applications". In: Stochastic Geometry: Lectures given at the C.I.M.E. 2004, Lecture Notes in Mathematics 1892.
- Baouan, A., S Coustou, M. Lacome, S. Pulido, and M. Rosenbaum (2023). "Crediting football players for creating dangerous actions in an unbiased way: the generation of threat (GoT) indices". In: Preprint at https://arxiv.org/abs/2304.05242.
- Bartlett, M. S. (1963). "The Spectral Analysis of Point Processes". In: Journal of the Royal Statistical Society. Series B (Methodological) 25.2, pp. 264–296.
- (1964). "The Spectral Analysis of Two-Dimensional Point Processes". In: *Biometrika* 51.3/4, pp. 299–311.
- Blundell, C., J. Beck, and K.A. Heller (2012). "Modelling Reciprocating Relationships with Hawkes Processes". In: Advances in Neural Information Processing Systems. Ed. by F. Pereira, C.J. Burges, L. Bottou, and K.Q. Weinberger. Vol. 25. Curran Associates, Inc.

- Bompaire, M., E. Bacry, and S. Gaïffas (2018). "Dual optimization for convex constrained objectives without the gradient-Lipschitz assumption". In: Preprint at https://arxiv.org/abs/1807.03545.
- Bonnet, A, F. Cheysson, M. Martinez Herrera, and M. Sangnier (2024). "Spectral analysis for noisy Hawkes processes inference". In: Preprint at https://arxiv.org/abs/2405.12581.
- Bonnet, A, M. Martinez Herrera, and M. Sangnier (2023). "Inference of multivariate exponential Hawkes processes with inhibition and application to neuronal activity". In: *Statistics and Computing* 33.91.
- Bonnet, A., C. Dion-Blanc, F. Gindraud, and S. Lemler (2022a). "Neuronal network inference and membrane potential model using multivariate Hawkes processes". In: *Journal of Neuroscience Methods* 372, p. 109550.
- Bonnet, A., C. Lacour, F. Picard, and V. Rivoirard (2022b). "Uniform deconvolution for Poisson point processes". In: *Journal of Machine Learning Research* 23.1.
- Bonnet, A., M. Martinez Herrera, and M. Sangnier (2021). "Maximum likelihood estimation for Hawkes processes with self-excitation or inhibition". In: *Statistics & Probability Letters* 179, p. 109214.
- Brémaud, P and L. Massoulié (2002). "Power Spectra of General Shot Noises and Hawkes Point Processes with a Random Excitation". In: Advances in Applied Probability 34.1, pp. 205–222.
 — (1996). "Stability of nonlinear Hawkes processes". In: The Annals of Probability 24.3, pp. 1563– 1588.
- Brémaud, P., L. Massoulié, and A. Ridolfi (Dec. 2005). "Power spectra of random spike fields and related processes". In: *Advances in Applied Probability* 37.
- Brillinger, D. R. (2012). Statistical Inference for Stationary Point Processes. New York, NY: Springer New York, pp. 499–543.
- Byrd, R.H., P. Lu, J. Nocedal, and C. Zhu (1995). "A Limited Memory Algorithm for Bound Constrained Optimization". In: SIAM Journal on Scientific and Statistical Computing 16.5, pp. 1190–1208.
- Carstensen, L., A. Sandelin, O. Winther, and N.R. Hansen (2010). "Multivariate Hawkes process models of the occurrence of regulatory elements". In: *BMC Bioinformatics* 11.1, p. 456.
- Chen, J.M., A.G. Hawkes, E. Scalas, and M. Trinh (2018). "Performance of information criteria for selection of Hawkes process models of financial data". In: *Quantitative Finance* 18.2, pp. 225–235.
- Chen, S., A. Shojaie, E. Shea-Brown, and D. Witten (2017). "The Multivariate Hawkes Process in High Dimensions: Beyond Mutual Excitation". In: Preprint at https://arxiv.org/abs/ 1707.04928.
- Chevallier, J., A. Duarte, E. Löcherbach, and G. Ost (2019). "Mean field limits for nonlinear spatially extended Hawkes processes with exponential memory kernels". In: *Stochastic Processes* and their Applications 129.1, pp. 1–27.
- Cheysson, F. and G. Lang (2022). "Spectral estimation of Hawkes processes from count data". In: The Annals of Statistics 50.3, pp. 1722–1746.
- Chiang, W-H., X. Liu, and G. Mohler (2022). "Hawkes process modeling of COVID-19 with mobility leading indicators and spatial covariates." In: *International Journal of Forecasting* 38.2, pp. 505–520.
- Chornoboy, E. S., L. P. Schramm, and A. F. Karr (1988). "Maximum likelihood identification of neural point process systems". In: *Biological Cybernetics* 59.4, pp. 265–275.
- Coeurjolly, J-F., A-L. Fougères, T. Espinasse, and M. Ribatet (2024). "Spatio-temporal point process intensity estimation using zero-deflated subsampling applied to a lightning strikes dataset in France". In: Preprint at https://arxiv.org/abs/2403.11564.

- Costa, M., C. Graham, L. Marsalle, and V. Tran (2020). "Renewal in Hawkes processes with self-excitation and inhibition". In: *Advances in Applied Probability* 52.3, pp. 879–915.
- Cox, D.R. and V. Isham (1980). Point Processes. Chapman & Hall/CRC Monographs on Statistics & Applied Probability. Taylor & Francis.
- Cronie, O., M. Moradi, and C.A.N. Biscio (June 2024). "A cross-validation-based statistical theory for point processes". In: *Biometrika* 111.2, pp. 625–641.
- Da Fonseca, J. and R. Zaatour (2013). "Hawkes Process: Fast Calibration, Application to Trade Clustering, and Diffusive Limit". In: Journal of Futures Markets 34.6, pp. 548–579.
- Daley, D. J. (1971). "Weakly Stationary Point Processes and Random Measures". In: Journal of the Royal Statistical Society. Series B (Methodological) 33.3, pp. 406–428.
- Daley, D.J. and D. Vere-Jones (2003). An introduction to the theory of point processes. Vol. I. Second. New York: Springer-Verlag.
- (2008). An introduction to the theory of point processes. Vol. II. Second. New York: Springer-Verlag.
- Dassios, A. and H. Zhao (2011). "A dynamic contagion process". In: Advances in Applied Probability 43.3, pp. 814–846.
- Davis, M. H. A. (1984). "Piecewise-Deterministic Markov Processes : A General Class of Non-Diffusion Stochastic Models". In: Journal of the Royal Statistical Society Series B (Methodological) 46.3, pp. 353–388.
- Denis, C., C. Dion-Blanc, R.E. Lacoste, L. Sansonnet, and Y. Bas (June 2024). "Bats monitoring: a classification procedure of bats behaviours based on Hawkes processes". In: *Journal of the Royal Statistical Society Series C: Applied Statistics*, qlae024.
- Deutsch, I. and G.J. Ross (2020). "ABC Learning of Hawkes Processes with Missing or Noisy Event Times". In: Preprint at https://arxiv.org/abs/2006.09015.
- (2024). "Estimating Product Cannibalisation in Wholesale using Multivariate Hawkes Processes with Inhibition". In: To appear in *Annals of Applied Statistics*.
- Donnet, S., V. Rivoirard, and J. Rousseau (2020). "Nonparametric Bayesian estimation for multivariate Hawkes processes". In: Annals of Statistics 48.5, pp. 2698–2727.
- Duarte, A., E. Löcherbach, and G. Ost (2019). "Stability, convergence to equilibrium and simulation of non-linear Hawkes processes with memory kernels given by the sum of Erlang kernels". In: ESAIM: Probability and Statistics 23, pp. 770–796. eprint: 1610.03300.
- Düker, M-C. and V. Pipiras (2019). "Asymptotic results for multivariate local Whittle estimation with applications". In: 2019 IEEE 8th International Workshop on Computational Advances in Multi-Sensor Adaptive Processing (CAMSAP), pp. 584–588.
- Duval, C., E. Luçon, and C. Pouzat (2022). "Interacting Hawkes processes with multiplicative inhibition". In: Stochastic Processes and their Applications 148, pp. 180–226.
- Eichler, M., R. Dahlhaus, and J. Dueck (2016). "Graphical Modeling for Multivariate Hawkes Processes with Nonparametric Link Functions". In: *Journal of Time Series Analysis* 38, pp. 225– 242.
- Embrechts, P., T. Liniger, and L. Lin (2011). "Multivariate Hawkes processes: an application to financial data". In: Journal of Applied Probability 48.A, pp. 367–378.
- Gerhard, F., M. Deger, and W. Truccolo (2017). "On the stability and dynamics of stochastic spiking neuron models: Nonlinear Hawkes process and point process GLMs." In: *PLoS* computational biology 13.2.
- Guan, Y. and D. R. Afshartous (2007). "Test for independence between marks and points of marked point processes: a subsampling approach". In: *Environmental and Ecological Statistics* 14.2, pp. 101–111.
- Guo, X., A. Hu, R. Xu, and J. Zhang (2018). "Consistency and Computation of Regularized MLEs for Multivariate Hawkes Processes". In: Preprint at https://arxiv.org/abs/1810.02955.

- Gupta, A., M. Farajtabar, B. Dilkina, and H. Zha (July 2018). "Discrete Interventions in Hawkes Processes with Applications in Invasive Species Management". In: Proceedings of the Twenty-Seventh International Joint Conference on Artificial Intelligence, IJCAI-18. International Joint Conferences on Artificial Intelligence Organization, pp. 3385–3392.
- Hansen, N.R., P. Reynaud-Bouret, and V. Rivoirard (2015). "Lasso and probabilistic inequalities for multivariate point processes". In: *Bernoulli* 21.1, pp. 83–143.
- Harris, T.E. (2002). The Theory of Branching Processes. Dover phoenix editions. Dover Publications.
- Hawkes, A.G. (1971). "Spectra of Some Self-Exciting and Mutually Exciting Point Processes". In: *Biometrika* 58.1, pp. 83–90.
- (2018). "Hawkes processes and their applications to finance: a review". In: Quantitative Finance 18.2, pp. 193–198.
- Hawkes, A.G. and D. Oakes (1974). "A cluster process representation of a self-exciting process". In: Journal of Applied Probability 11.3, pp. 493–503.
- Hohage, Thorsten and Frank Werner (2016). "Inverse problems with Poisson data: statistical regularization theory, applications and algorithms". In: *Inverse Problems* 32.9, p. 093001.

Joliffe, I.T. (2002). Principal Component Analysis. Second. New York: Springer-Verlag.

Karavasilis, G. J. and A. G. Rigas (2007). "Spectral analysis techniques of stationary point processes used for the estimation of cross-correlation: Application to the study of a neurophysiological system". In: 2007 15th European Signal Processing Conference, pp. 2479–2483.

- Kirchner, M. (2017). "An estimation procedure for the Hawkes process". In: Quantitative Finance 17.4, pp. 571–595.
- Kwon, J., Y. Zheng, and M. Jun (2023). "Flexible spatio-temporal Hawkes process models for earthquake occurrences". In: Spatial Statistics 54, p. 100728.
- Lambert, R.C., C. Tuleau-Malot, T. Bessaih, V. Rivoirard, Y. Bouret, N. Leresche, and P. Reynaud-Bouret (2018). "Reconstructing the functional connectivity of multiple spike trains using Hawkes models". In: Journal of Neuroscience Methods 297, pp. 9–21.
- Last, G. and M. Penrose (2017). *Lectures on the Poisson Process*. Institute of Mathematical Statistics Textbooks. Cambridge University Press.
- Laub, P. (2014). "Hawkes Processes: Simulation, Estimation, and Validation". In: Bachelor's Thesis, University of Queensland.
- Lemonnier, R. and N. Vayatis (2014). "Nonparametric Markovian Learning of Triggering Kernels for Mutually Exciting and Mutually Inhibiting Multivariate Hawkes Processes". In: *Machine Learning and Knowledge Discovery in Databases*. Ed. by T. Calders, F. Esposito, E. Hüllermeier, and R. Meo. Vol. 8725. Berlin: Springer Berlin Heidelberg, pp. 161–176.
- Lesage, L., M. Deaconu, A. Lejay, J.A. Meira, G. Nichil, and R. State (2022). "Hawkes Processes Framework With a Gamma Density As Excitation Function: Application to Natural Disasters for Insurance". In: *Methodology and Computing in Applied Probability* 24.4, pp. 2509–2537.
- Lewis, E. and G. Mohler (2011). "A Nonparametric EM Algorithm for Multiscale Hawkes Processes". In: Journal of Nonparametric Statistics 1, pp. 1–20.
- Lewis, Peter A. W. (1964). "A Branching Poisson Process Model for the Analysis of Computer Failure Patterns". In: Journal of the Royal Statistical Society: Series B (Methodological) 26.3, pp. 398–441.
- Li, T. and Y. Ke (2019). "Thinning for Accelerating the Learning of Point Processes". In: Advances in Neural Information Processing Systems. Ed. by H. Wallach, H. Larochelle, A. Beygelzimer, F. d'Alché-Buc, E. Fox, and R. Garnett. Vol. 32. Curran Associates, Inc.
- Linderman, S. and R Adams (2014). "Discovering Latent Network Structure in Point Process Data". In: Proceedings of the 31st International Conference on Machine Learning. Ed. by

E. P. Xing and T. Jebara. Vol. 32. Beijing, China: Proceedings of Machine Learning Research, pp. 1413–1421.

- Lotz, A. (2024). "A sparsity test for multivariate Hawkes processes". In: Preprint at https://arxiv.org/abs/2405.08640.
- Lu, X and F. Abergel (2018). "High-dimensional Hawkes processes for limit order books: modelling, empirical analysis and numerical calibration". In: *Quantitative Finance* 18.2, pp. 249– 264.
- Lund, J. and M. Rudemo (2000). "Models for Point Processes Observed with Noise". In: Biometrika 87.2, pp. 235–249.
- Mei, H. and J. Eisner (2017). "The Neural Hawkes Process: A Neurally Self-Modulating Multivariate Point Process". In: Proceedings of the 31st International Conference on Neural Information Processing Systems. California: Curran Associates Inc., pp. 6757–6767.
- Mei, H., G. Qin, and J. Eisner (2019). "Imputing Missing Events in Continuous-Time Event Streams". In: Proceedings of the 36th International Conference on Machine Learning. Ed. by Kamalika Chaudhuri and Ruslan Salakhutdinov. Vol. 97. Proceedings of Machine Learning Research. PMLR, pp. 4475–4485.
- Menon, A. and Y. Lee (Apr. 2018). "Proper Loss Functions for Nonlinear Hawkes Processes". In: Proceedings of the AAAI Conference on Artificial Intelligence 32.
- Mishra, S., M.A. Rizoiu, and L. Xie (2016). "Feature Driven and Point Process Approaches for Popularity Prediction". In: Proceedings of the 25th ACM International Conference on Information and Knowledge Management, CIKM 2016, Indianapolis, IN, USA, October 24-28, 2016. New York: ACM, pp. 1069–1078.
- Mohler, G., M. Short, P. Brantingham, F. Schoenberg, and G. Tita (Mar. 2011). "Self-Exciting Point Process Modeling of Crime". In: *Journal of the American Statistical Association* 106, pp. 100–108.
- Møller, J. and J.G. Rasmussen (2005). "Perfect Simulation of Hawkes Processes". In: Advances in Applied Probability 37.3, pp. 629–646.
- Møller, J. and R.P. Waagepetersen (2003). Statistical Inference and Simulation for Spatial Point Processes. First. New York, NY: Chapman and Hall/CRC.
- Moradi, M. M., O. Cronie, E. Rubak, R. Lachieze-Rey, J. Mateu, and A. Baddeley (2019). "Resample-smoothing of Voronoi intensity estimators". In: *Statistics and Computing* 29.5, pp. 995–1010.
- Mugglestone, M.A. and E. Renshaw (1996). "A practical guide to the spectral analysis of spatial point processes". In: *Computational Statistics & Data Analysis* 21.1, pp. 43–65.
- (2001). "Spectral tests of randomness for spatial point patterns". In: Environmental and Ecological Statistics 8.3, pp. 237–251.
- Narayanan, S., I. Kosmidis, and P. Dellaportas (2023). "Flexible marked spatio-temporal point processes with applications to event sequences from association football". In: Journal of the Royal Statistical Society Series C: Applied Statistics 72.5, pp. 1095–1126.
- Nicvert, L., S. Donnet, M. Keith, M. Peel, M.J. Somers, K.H. Swanepoel, J. Venter, H. Fritz, and S. Dray (2024). "Using the multivariate Hawkes process to study interactions between multiple species from camera trap data". In: *Ecology* 105.4, e4237.
- Ogata, Y. (1978). "The asymptotic behaviour of maximum likelihood estimators for stationary point processes". In: Annals of the Institute of Statistical Mathematics 30, pp. 243–261.
- (1981). "On Lewis' simulation method for point processes". In: *IEEE Transactions on Infor*mation Theory 27, pp. 23–30.
- (1988). "Statistical Models for Earthquake Occurrences and Residual Analysis for Point Processes". In: Journal of the American Statistical Association 83.1, pp. 9–27.

- Ogata, Y. (1998). "Space-Time Point-Process Models for Earthquake Occurrences". In: Annals of the Institute of Statistical Mathematics 50.2, pp. 379–402.
- Olinde, J. and M.B. Short (2020). "A Self-limiting Hawkes Process: Interpretation, Estimation, and Use in Crime Modeling". In: 2020 IEEE International Conference on Big Data (Big Data), pp. 3212–3219.
- Ozaki, T. (1979). "Maximum likelihood estimation of Hawkes' self-exciting point processes". In: Annals of the Institute of Statistical Mathematics 31.1, pp. 145–155.
- Petersen, P.C. and R.W. Berg (2016). "Lognormal firing rate distribution reveals prominent fluctuation-driven regime in spinal motor networks". In: *eLife* 5, e18805.
- Pham, D.T. (1996). "Blind separation of instantaneous mixture of sources via an independent component analysis". In: *IEEE Transactions on Signal Processing* 44.11, pp. 2768–2779.
- Pinsky, M.A. (2008). Introduction to Fourier Analysis and Wavelets. Graduate studies in mathematics. American Mathematical Society.
- Pinto, J.C.L., T. Chahed, and E. Altman (2015). "Trend detection in social networks using Hawkes processes". In: 2015 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining (ASONAM), pp. 1441–1448.
- Politis, D. N., J. P. Romano, and M. Wolf (1999). Subsampling Marked Point Processes. New York, NY: Springer New York, pp. 138–158.
- Possolo, A. (1991). "Subsampling a Random Field". In: Lecture Notes-Monograph Series 20, pp. 286–294.
- Radosevic, M., A. Willumsen, P.C. Petersen, H. Lindén, M. Vestergaard, and R.W. Berg (2019). "Decoupling of timescales reveals sparse convergent CPG network in the adult spinal cord". In: *Nature Communications* 10.2937, pp. 1–14.
- Rajala, T.A., S.C. Olhede, J.P. Grainger, and D.J. Murrell (2023). "What is the Fourier Transform of a Spatial Point Process?" In: *IEEE Transactions on Information Theory*.
- Rasmussen, J.G. (2013). "Bayesian Inference for Hawkes Processes". In: Methodology and Computing in Applied Probability 15.3, pp. 623–642.
- Reynaud-Bouret, P., A. Muzy, and I. Bethus (2021). "Towards a mathematical definition of functional connectivity". In: Comptes rendus. Mathématique 359, pp. 481–492.
- Reynaud-Bouret, P., V. Rivoirard, F. Grammont, and C. Tuleau-Malot (2014). "Goodness-of-Fit Tests and Nonparametric Adaptive Estimation for Spike Train Analysis". In: *The Journal of Mathematical Neuroscience* 4.3, p. 3.
- Reynaud-Bouret, P., V. Rivoirard, and C. Tuleau-Malot (2013). "Inference of functional connectivity in Neurosciences via Hawkes processes". In: pp. 317–320.
- Reynaud-Bouret, P. and S. Schbath (2009). "Adaptive estimation for Hawkes processes; application to genome analysis". In: The Annals of Statistics 38.
- Rizoiu, M., S. Mishra, Q. Kong, M. Carman, and L. Xie (2018). "SIR-Hawkes: linking epidemic models and Hawkes processes to model diffusions in finite populations". In: *Proceedings of* the 2018 World Wide Web Conference. International World Wide Web Conferences Steering Committee, pp. 419–428.
- Rizoiu, M.A., Y. Lee, S. Mishra, and L. Xie (2017). "A Tutorial on Hawkes Processes for Events in Social Media". In: Preprint at https://arxiv.org/abs/1708.06401.
- Roueff, F. and R. von Sachs (2019). "Time-frequency analysis of locally stationary Hawkes processes". In: *Bernoulli* 25.2, pp. 1355–1385.
- Roueff, F., R. von Sachs, and L. Sansonnet (2016). "Locally stationary Hawkes processes". In: Stochastic Processes and their Applications 126.6, pp. 1710–1743.
- Särndal, C.E., B. Swensson, and J. Wretman (2003). Model Assisted Survey Sampling. Springer Series in Statistics. New York, NY: Springer New York.

- Simon, C. and Y. Nakahiro (2017). "Statistical inference for ergodic point processes and application to Limit Order Book". In: Stochastic Processes and their Applications 127.6, pp. 1800– 1839.
- Sulem, D., V. Rivoirard, and J. Rousseau (2023). "Scalable and adaptive variational Bayes methods for Hawkes processes". In: Preprint at https://arxiv.org/abs/2212.00293.
- (2024). "Bayesian estimation of nonlinear Hawkes processes". In: *Bernoulli* 30.2, pp. 1257–1286.
- Tibshirani, R. (1996). "Regression Shrinkage and Selection via the Lasso". In: Journal of the Royal Statistical Society. Series B (Methodological) 58.1, pp. 267–288.
- Trouleau, W., J. Etesami, M. Grossglauser, N. Kiyavash, and P. Thiran (2019). "Learning Hawkes Processes Under Synchronization Noise". In: *Proceedings of the 36th International Conference* on Machine Learning. Ed. by Kamalika Chaudhuri and Ruslan Salakhutdinov. Vol. 97. Proceedings of Machine Learning Research. PMLR, pp. 6325–6334.
- Tuan, P.D. (1981). "Estimation of the Spectral Parameters of a Stationary Point Process". In: *The Annals of Statistics* 9.3, pp. 615–627.
- Veen, A. and F.P. Schoenberg (2008). "Estimation of Space-Time Branching Process Models in Seismology Using an EM-Type Algorithm". In: Journal of the American Statistical Association 103.482, pp. 614–624.
- Villani, M., M. Quiroz, R. Kohn, and R. Salomone (2022). "Spectral Subsampling MCMC for Stationary Multivariate Time Series with Applications to Vector ARTFIMA Processes". In: *Econometrics and Statistics*.
- Watson, H. W. and F. Galton (1875). "On the Probability of the Extinction of Families". In: The Journal of the Anthropological Institute of Great Britain and Ireland 4, pp. 138–144.
- Whittle, P (1952). "Some results in time series analysis". In: Scandinavian Actuarial Journal 1952.1-2, pp. 48–60.
- Xu, H., M. Farajtabar, and H. Zha (2016). "Learning Granger Causality for Hawkes Processes". In: Proceedings of The 33rd International Conference on Machine Learning. Ed. by Maria Florina Balcan and Kilian Q. Weinberger. Vol. 48. Proceedings of Machine Learning Research. New York, New York, USA: PMLR, pp. 1717–1726.
- Yang, J. and Y. Guan (2024). "Fourier analysis of spatial point processes". In: Preprint at https://arxiv.org/abs/2401.06403.
- Yang, Y., J. Etesami, N. He, and N. Kiyavash (2017). "Online Learning for Multivariate Hawkes Processes". In: Advances in Neural Information Processing Systems. Ed. by I. Guyon, U. Von Luxburg, S. Bengio, H. Wallach, R. Fergus, S. Vishwanathan, and R. Garnett. Vol. 30. California: Curran Associates, Inc.
- Zhang, C. (2016). "Modeling High Frequency Data Using Hawkes Processes with Power-law Kernels1". In: *Procedia Computer Science* 80. International Conference on Computational Science 2016, ICCS 2016, 6-8 June 2016, San Diego, California, USA, pp. 762–771.
- Zuo, S., H. Jiang, Z. Li, T. Zhao, and H. Zha (2020). "Transformer Hawkes Process". In: Proceedings of the 37th International Conference on Machine Learning. California: PMLR, pp. 11692– 11702.

INFERENCE OF NON-LINEAR OR IMPERFECTLY OBSERVED HAWKES PROCESSES Abstract

The Hawkes point process is a popular statistical tool to analyse temporal patterns. Modern applications propose extensions of this model to account for specificities in each field of study, which in turn complexifies the task of inference. In this thesis, we advance different approaches for the parametric estimation of two submodels of the Hawkes process in univariate and multivariate settings. Motivated by the modelling of complex neuronal interactions observed from spike train data, our first study focuses on accounting for both inhibition and excitation effects between neurons, modelled by the non-linear Hawkes process. We derive a closed-form expression of the log-likelihood in order to implement a maximum likelihood procedure. As a consequence of our approach, we gain access to a goodness-of-fit scheme allowing us to establish ad hoc model selection methods to estimate the interaction network in the multivariate setting. The second part of this thesis focuses on studying Hawkes process data noised by two different alterations: adding or removing points. The absence of knowledge on the noise dynamics makes classical inference procedures intractable or computationally expensive. Our solution is to leverage the spectral analysis of point processes to establish an estimator obtained by maximising the spectral log-likelihood. By deriving the spectral densities of the noised processes and by establishing identifiability conditions on our model, we show that the spectral inference method does not necessitate any information on the structure of the noise, effectively circumventing this issue. An additional result of the study of Hawkes processes with missing points is that it gives access to a subsampling paradigm to enhance the estimation methods by introducing a penalisation parameter. We illustrate the efficiency of all of our methods through reproducible numerical implementations.

Keywords: hawkes processes, parametric inference, identifiability, inhibition, spectral theory, neuronal data

Résumé

Le processus ponctuel de Hawkes est un outil statistique très répandu pour analyser des dynamiques temporelles. Les applications modernes des processus de Hawkes proposent des extensions du modèle initial pour prendre en compte certaines caractéristiques spécifiques à chaque domaine d'étude, ce qui complexifie les tâches d'inférence. Dans cette thèse, nous proposons différentes contributions à l'estimation paramétrique de deux variantes du processus de Hawkes dans les cadres univarié et multivarié. Motivée par la modélisation d'interactions complexes au sein d'une population de neurones, notre première étude porte sur la prise en compte conjointe d'effets excitateurs et inhibiteurs entre les signaux émis par les neurones au cours du temps, modélisés par un processus de Hawkes non-linéaire. Dans ce modèle, nous obtenons une expression explicite de la log-vraisemblance qui nous permet d'implémenter une procédure de maximum de vraisemblance. Nous établissons également une méthode de sélection de modèle qui fournit notamment une estimation du réseau d'interactions dans le cadre multivarié. La deuxième partie de cette thèse est consacrée à l'étude des processus de Hawkes bruités par deux types d'altérations : l'ajout ou la suppression de certains points. Le manque d'information lié à ces mécanismes de bruit rend les méthodes classiques d'inférence non-applicables ou numériquement coûteuses. Notre solution consiste à s'appuyer sur l'analyse spectrale des processus ponctuels afin d'établir un estimateur obtenu en maximisant la log-vraisemblance spectrale. Nous obtenons l'expression des densités spectrales des processus bruités et, après avoir établi des conditions d'identifiabilité pour nos différents modèles, nous montrons que cette méthode d'inférence ne nécessite pas de connaître la structure du bruit, contournant ainsi le problème d'estimation. Notre étude sur les processus bruités donne accès à une méthode de sous-échantillonnage qui nous permet d'améliorer les approches d'estimation en introduisant un paramètre de pénalisation. Nous illustrons la performance des différentes méthodes proposées à travers des implémentations numériques reproductibles.

Mots clés : processus de hawkes, inférence paramétrique, identifiabilité, inhibition, théorie spectrale, données neuronales

